ISOLATION AND CHARACTERISATION OF CELLULOLYTIC MICROORGANISMS FOR OIL PALM (*ELAEIS GUINEENSIS*) WASTE MANAGEMENT: EFFECTIVE COMPOSTING OF EMPTY FRUIT BUNCHES FOR SUSTAINABLE CROP CULTIVATION

By

MERTENSIA KHO KIA CHEN

A thesis

presented in fulfilment of the requirement for the degree of

Master of Science by Research

at Swinburne University of Technology

2018

Abstract

Oil palm, *Elaeis guineensis*, has been increasingly cultivated in the recent decades due to the growing requirement of palm oil, palm kernel oil, biodiesel and other products. Due to the ever increasing demands, wastes produced from the palm oil mill have also been steadily mounted up, becoming an environmental hazard. An estimate of 80 million tonnes of agro-industrial waste is generated by the oil palm industry each year. Moreover, the usage of chemical fertiliser to cultivate these oil palms could be an environmental impact and may pollute the environment. Besides, chemical fertilisers are costly and may consist up to 60% of the cost for the production of fresh fruit bunch in the oil palm plantation. Solutions must be found to overcome the problems and composting has often been proposed. Composting is a biological based process which converts organic waste to organic matter that is known as compost. Composting recycles the waste products produced from the oil palm industry into sustainable fertiliser which can be implemented back into the plantation. However, oil palm empty fruit bunch (OPEFB), one of the major wastes produced, is lignocellulosic thus requires a longer period of time to degrade. Studies had claimed that co-composting OPEFB with cellulolytic microbes that produce cellulase and hemicellulase enzymes degrades the lignocellulose at a faster rate. In this study, (a) soil samples collected from regions in Sarawak were isolated for cellulolytic microorganisms using selective media; (b) the microorganisms isolated were assessed and selected for their cellulose and hemicellulose degrading abilities using 2cyanoacetamide method; (c) the selected microorganisms were used to form a consortium to be inoculated into composting materials to produce compost, and (d) the compost produced was assessed for its efficiency in enhancing the growth of oil palm seedlings via 6-month pot trial. The indigenous microbial consortium consisting of selected cellulolytic strain Bacillus sp. S2, Bacillus sp. S8, Penicillium citrinum, Aspergillus nomius, and Penicillium rolfsii were identified through molecular methods. This indigenous microbial consortium had shown to decompose OPEFB and produce compost as efficient as the commercial microbial agents (containing Bacillus cereus, Bacillus subtilis, and Bacillus thuringiensis), as the indigenous microbial consortium actually produced compost with higher nutrient contents (35% N, 57% P, and 9% Mg) when compared to the commercial microbial agents. The indigenous microbial consortium had

also completed the composting of OPEFB in 30 days, shorter than a similar study by Baharuddin the composting process of OPEFB with partially treated POME was completed in 60 days. The results from the pot trial also revealed that the compost produced by the indigenous microbial consortium was able to enhance the growth of oil palm seedlings significantly better than no-fertiliser control, and comparable with chemical fertiliser. The compost could also be mixed with half amount of normal chemical fertiliser requirement and able to sustain the growth of seedlings equally well. On the basis of the results obtained in this research, it can be concluded that the indigenous microbial consortium containing the isolated cellulolytic microorganisms isolated from the soil have the potential to decompose OPEFB to produce compost for the use of oil palm cultivation.

Key words: Cellulolytic, empty fruit bunch, palm oil mill effluent, composting, oil palm

Acknowledgement

First and foremost, I would like to thank my research supervisor, Dr Daniel Tan Lee Tung for his unceasing guidance and support throughout my study.

Associate Professor Dr Wallace Wong for his help in securing the sponsorship and also employing me as a research assistant so that I would be financially stable throughout my study.

My co-supervisors, Dr Irine Runnie Henry Ginjom and Associate Professor Dr Moritz Müller for their guidance and support throughout my study.

Daitoku Sdn. Bhd. for the financial support in my masters' project (Agreement No. G01L212621-Q040).

My beloved family, especially my parents, for their trust, continuous support, and neverending encouragement to me that I can do my best in anything I want to do when I put my mind to it.

A special word of gratitude to my precious, Jasper, for all his love and support.

My lab mates, Cindy, Alan, Wen Loong, Yeli, Fiona, Ang, Johnny, Phyllis, and Wanie for their support and help throughout my study.

Laboratory staff in the Science department of Faculty of Engineering, Computing, and Science in Swinburne University of Technology Sarawak Campus for their assistance.

Sarawak Biodiversity Centre (SBC) for granting the permit approval which allowed me to collect soil samples from various location in Sarawak for my experiment. (Research Agreement No. SBC-RA-112-TLT)

Declaration

I, Mertensia Kho Kia Chen hereby declare that this research study entitled "Isolation and characterisation of cellulolytic microorganisms for oil palm (*Elaeis Guineensis*) waste management: effective composting of empty fruit bunches for sustainable crop cultivation" is original and does not contain any material which has been accepted for the award to the candidate of any other degree or diploma, except where due reference is made in the text of the examinable outcome. To the best of my knowledge, this thesis does not contain any material previously published or written by another person except where due reference is made in the text of the examinable outcome, I have disclosed relative contributions of the respective workers or authors.



Mertensia Kho Kia Chen

As the principal coordinating supervisor, I hereby acknowledge and verify that the abovementioned statements are legitimate to the best of my knowledge.

ynite une

Conference Presentations

Poster presenter, Isolation and characterisation of microorganisms for the formulation of biofertiliser for oil palm (*Elaeis guineensis*) cultivation, *33rd Symposium of the Malaysian Society for Microbiology*, 14-17 December 2016, Ramada Plaza, Melaka, Malaysia.

Award

3rd Best Poster Presenter

Awarded 3rd best poster presenter for the technical session of the general category at the 33rd Symposium of the Malaysian Society for Microbiology organised by Malaysian Society for Microbiology, 14-17 December 2016, Ramada Plaza, Melaka, Malaysia.

Table of Contents

Content		Page
Abstract		I
Acknowled	gement	III
Declaration	-	IV
Conforma	Prosontations	V
		V
Table of Co	ntents	VI
List of Abb	reviations	IX
List of Figu	res	X
List of Tabl	es	XIII
Chapter 1 I	ntroduction and Literature Review	1
1.1 Int	oduction	2
1.2 Lit	erature Review	4
1.2.1	Malaysia's Oil Palm Industry	4
1.2.2	Waste Production from the Oil Palm Industry in Malaysia	7
1.2.3	Composition of Oil Palm Empty Fruit Bunch	9
1.2.4	Composting of Oil Palm Empty Fruit Bunch	11
1.2.5	Cellulolytic Microorganisms	15
1.2.6	Mechanism of Cellulolytic Microorganisms	17
1.2.7	Screening for Cellulolytic Microorganisms	
1.2.8	Plant Growth Promoting Microorganisms	19
1.3 Res	search Aims and Objectives	22
1.4 The	esis Outline	23
Chapter 2 S	creening and Characterisation of Cellulolytic Microorganis	ns from
Sarawak So	il	25
2.1 Int	oduction	
2.2 Ma	terials and Methods	27
2.2.1	Sampling Locations and Collections	27
2.2.2	Screening for Potential Microorganisms from Soil Samples	27

2.2.3	Isolation of Pure Culture	
2.2.4	Storage and Preservation	
2.2.5	Molecular Identification	
2.3 Re	esults and Discussion	
2.3.1	Isolation of Celluloytic Microorganisms	
2.3.2	Screening for Plant Growth-Promoting Microorganisms	
2.3.3	Molecular Identification of the Isolated Bacteria and Fungus	40
2.4 Su	ımmary	
Chapter 3	Selection of Cellulolytic Microorganisms Based On Enzyme Act	tivity
and Optim	isation of Selected Microorganisms	
3.1 In	troduction	
3.2 M	aterials and Methods	51
3.2.1	Measurement of Enzyme Activity of Isolated Cellulolytic Microo 51	organisms
3.2.2	Screening for Antagonistic Activity of Selected Strains	53
3.2.3	Optimisation of Bacterial and Fungal Strains	53
3.2.4	Microbial Growth Rate of Selected Microorganisms	
3.2.5	Data Analysis	
3.3 Re	esults and Discussion	
3.3.1 Microo	Measurement of Enzyme Activity by Selected Cellulolytic organisms	57
3.3.2	Screening for Antagonism Activity Among Selected Strains	60
3.3.3	Optimisation of Bacterial and Fungal Strains	
3.3.4	Microbial Growth Rate of Selected Microorganisms	69
3.3.5 Consor	Selected Final Strains for Development of Indigenous Microbial rtium.	73
3.4 Su	immary	74
Chapter 4	Production of Compost Via In-Vessel Composting of Oil Palm	Empty
Fruit Bund	ch Using Cellulolytic Microorganisms and Its Enhancement in T	The T
Growth of	Oil Palm Seedlings	75
4.1 In	troduction	
4.2 M	aterials and Methods	
4.2.1	In-Vessel Composting	
	1 0	

4.2	2.2	Pot Trial	
4.3	Res	ults and Discussion	
4.3	3.1	In-Vessel Composting	
4.3	3.2	Pot Trial	95
4.4	Sun	nmary	
Chapt	er 5 G	General Conclusions, Recommendations, and Future Works	107
5.1	Ger	neral Conclusion	
5.	1.1	Aims of Thesis	
5.	1.2	Screening and Isolation of Cellulolytic Microorganism	
5.1 M Pr	1.3 icrobi oduce	Evaluating Efficiency of Cellulolytic Microorganisms as an India al Consortium in Degrading OPEFB Via In-Vessel Composting M Compost	genous 1ethod to 109
5.1 Co	1.4 ompos	Small Scale Pot Trial to Evaluate the Growth Enhancement Pote t Group in Oil Palm Seedlings	ncy of 110
5.2	Fut	ure Works	111
5.3	Cor	ncluding Statements	112
Refere	ences.		113
Appen	dix		127

List of Abbreviations

°C	Degree Celsius
BLAST	Basic Local Alignment Search Tool
Cm	Centimetre
СМС	Carboxymethylcellulose
DNA	Deoxyribonucleic acid
EFB	Empty fruit bunch
FFB	Fresh fruit bunch
Hr	Hour
NA	Nutrient agar
NB	Nutrient broth
NCBI	National Center for Biotechnology Information
PCR	Polymerase chain reaction
PDA	Potato dextrose agar
PDB	Potato dextrose broth
POME	Palm oil mill effluent
RBBR	Remazol brilliant blue-R
SBC	Sarawak Biodiversity Centre
UV	Ultraviolet

List of Figures

- Figure 1.1Malaysia's oil palm planted area by state at December 20155(Malaysian Palm Oil Board 2015)
- Figure 1.2Ten largest producers of palm oil in the world for the year62011 (Global Palm Oil Conference IUF 2015).
- Figure 1.3Average yield of oil from different oil-bearing crop in the 7
world (Malaysian Palm Oil Board 2011).
- Figure 1.4Products/wastes from each bunch of fresh fruit bunch 8
(Abdullah & Sulaiman 2013)
- Figure 1.5 Arrangement of fibrils, microfibrils, and cellulose in plant 10 cell walls
- Figure 2.1Clear zones indicating the presence of cellulase in CMC agar35(A: before staining) and (B: after staining).
- Figure 2.2 Pure isolates showing xylan-degrading activities on dyed 36 xylan agar: (A) control; (B) clear zone formed by bacterial isolates; and (C) fungal isolates when xylanase could be produced.
- Figure 2.3 Cellulolytic microorganisms grown on Jensen's medium 38 with bromothymol blue dye (A) neutral state green; (B) blue alkaline, and (C) yellow acidic.
- Figure 2.4 Cellulolytic microorganisms grown on Pikovskaya agar (A) 39 and Aleksandrow agar (B) showed clear zone indicates solubilising property.
- Figure 2.5 DNA bands of the PCR product extracted from the 41 cellulolytic bacterial isolates.
- Figure 2.6 DNA bands of the PCR product extracted from the 41 cellulolytic fungal isolates.
- Figure 2.7The molecular phylogenetic tree showing the position of the46cellulolytic bacterial isolates.

Figure 2.8The molecular phylogenetic tree showing the position of the47cellulolytic fungal isolates.

- Figure 3.1Glucose levels produced by cellulase and hemicellulase 58
producing microorganisms as detected using
2-cyanoacetamide method.
- Figure 3.2 Antagonistic activity evaluations were performed for the 5 61 selected cellulolytic strains: *Bacillus* sp. S2, and *Bacillus* sp. S8, *P. citrinum* F6, A. *nomius* F8, and *P. rolfsii* F10.
- Figure 3.3 The selected bacterial strains, *Bacillus* sp. S2 (A) and 63 *Bacillus* sp. S8 (B) exhibited higher growth rates when placed in growth condition of pH6.0.
- Figure 3.4 Effect of different pH on the growth for selected bacterial 63 strains *Bacillus* sp. S2 and *Bacillus* sp. S8.
- Figure 3.5 Selected fungal strain *P. citrinum* F6 (A), *A. nomius* F8 (B), 64 and *P. rolfsii* F10 (C) showed to favour the growth condition of pH6.5 where more mycelia were produced.
- Figure 3.6Effect of different pH on the growth of selected fungal strain64P. citrinum F6, A. nomius F8, and P. rolfsii F10.
- Figure 3.7 The selected bacterial strains *Bacillus* sp. S2 (A) and 65 *Bacillus* sp. S8 (B) exhibited higher growth rates when placed in a 45°C condition based on higher absorbance readings.
- Figure 3.8 Effect of different temperature on the growth for selected 66 bacterial strains *Bacillus* sp. S2 and *Bacillus* sp. S8.
- Figure 3.9 Selected fungal strain *P. citrinum* F6 (A), *A. nomius*, F8 (B), 67 and *P. rolfsii* F10 (C) showed to favour an incubation temperature of 35°C where more mycelia were produced.
- Figure 3.10 Effect of different temperature on the growth of selected 67 fungal strain *P. citrinum* F6, *A. nomius* F8, and *P. rolfsii* F10.
- Figure 3.11 Viable count or plate count was used for direct growth 70 measurement technique of the selected bacterial strain *Bacillus* sp. S2 (A) and *Bacillus* sp. S8 (B).

- Figure 3.12The dry weight of the mycelium obtained from the 5-day old72culture broth was used for the indirect growth measurement
technique to measure the fungal growth.
- Figure 3.13 The selected 5 strains namely *Bacillus* sp. S2 (A), *Bacillus* 73 sp. S8 (B), *P. citrinum* F6 (C), *A. nomius* F8 (D), and *P. rolfsii* F10 (E) were chosen for the formulation of the degrading microorganisms consortium for the in-vessel composting discussed in Chapter 4.
- Figure 4.1Composting vessels were fabricated with door latches (A)78and placed on stand stands with rollers (B) to allow easy
rotation.rotation.
- Figure 4.2 Technique used to measure the height (A) and stem girth 84 (B&C) of the oil palm.
- **Figure 4.3** Physical changes of oil palm empty fruit bunch at Day 0 (A), 85 Day 20 (B), and Day 60 (C).
- Figure 4.4 Compost temperature profile for 60 days of in-vessel 89 composting.
- Figure 4.5 Compost C: N ratio profile throughout 60 days of in-vessel 91 composting.
- Figure 4.6 Compost pH profile for 60 days of in-vessel composting. 92
- Figure 4.7 Nutrient increments of the composts after 60 days of in- 93 vessel composting.
- Figure 4.8 Physical appearances of oil palms treated with different 102 treatments which were chemical fertiliser (A), compost (B), and compost-chemical mixed fertiliser (C) after 6 months of the experiment.

List of Tables

Table 2.1	DNA extraction - Freeze and thaw cycle.		
Table 2.2	Primer sequences.		
Table 2.3	Summary of isolated cellulolytic microorganisms from soil	37	
	samples.		
Table 2.4	Indole acetic acid (IAA) produced by cellulolytic microorganisms		
Table 2.5	Molecular identification of the cellulolytic bacterial isolates.		
Table 2.6	Molecular identification of the cellulolytic fungal isolates.		
Table 2.7	The closest match of strains identity obtained for the isolated 4		
	isolates after comparison with GenBank database.		
Table 3.1	Summary of increasing growth rate of selected bacterial strains after	68	
	optimisation.		
Table 3.2	Summary of increasing growth rate of selected fungal strains after	69	
	optimisation.		
Table 3.3	Growth kinetics characteristics of the selected bacterial strains	70	
	Bacillus sp. S2 and Bacillus sp. S8 grown in NB at 35°C, 120 rpm.		
Table 3.4	Growth kinetics characteristics of the selected fungal strains, P.	72	
	citrinum F6, A. nomius F8, and P. rolfsii F10 grown in PDA at		
	35°C, 120 rpm.		
Table 4.1	Experimental groups for the in-vessel composting study.	81	
Table 4.2	Treatment groups for pot trial study.	83	
Table 4.3	Properties of shredded OPEFB, partially treated POME and final	87	
	compost for the three experimental test group at Day 60.		
Table 4.4	Numbers of new fully opened leaves of oil palms over a	96	
	6-month period.		
Table 4.5	Heights of oil palms over a 6-month period.	97	
Table 4.6	Stem girths of oil palms over a 6-month period.	98	
Table 4.7	Nutrient levels in oil palm leaves over a 6-month period.	99	

Chapter

1

INTRODUCTION AND LITERATURE REVIEW

1.1 Introduction

Malaysia is currently the second largest palm oil producer in the world. Malaysia produces about 39% of palm oil after Indonesia, who is the world largest palm oil producer, producing approximately 44% of the world's palm oil (Global Palm Oil Conference 2015). In the year 2015, it is recorded that 5.63 million hectares of Malaysian land are covered with oil palms. Sarawak being the largest state in Malaysia, produces 25% of the palm oil industry (Malaysian Palm Oil Board 2015). Sarawak and Sabah are located on the island of Borneo experience humid tropical climate with sunshine throughout the year which makes it an ideal place to plant oil palms.

Despite the great contribution of oil palm has given to the economy in Malaysia, waste produced from the oil palm industry has become a major environmental hazard. According to Hasamudin and Soom (2002), approximately 55 tonnes of dry fibrous biomass matter is produced from each hectare of oil palm plantation. These dry waste matters were usually incinerated by mills to obtain energy (Hasamudin & Soom 2002). However, burning the dry waste contributed greatly to the air quality problem in Malaysia (Mahjoub *et al.* 2013). Hence, sustainable waste management must be sought after to tackle the waste produced from the palm oil industry.

Oil palm empty fruit bunch (OPEFB) represents one of the major dry waste matters produced by the oil palm industry (Abdullah & Sulaiman 2013). OPEFB is a lignocellulosic material as it constitutes three major constituents namely cellulose, hemicellulose and lignin (Lai *et al.* 2015). Cellulose is the most abundant constituent contributing approximately 53% in OPEFB (Baharuddin *et al.* 2009). Cellulose occurs naturally in plants as a polymer which comprises of glucose units joined together by β -1,4-glycosidic bonds. Cellulose chains in linear form are bundled together to form microbrils. The microbils are composed of amorphous and crystalline regions. The crystalline regions in the microbils exhibit strong internal bonding while amorphous regions have a weaker internal bonding (Ramli *et al.* 2015).

Composting of lignocellulosic OPEFB often requires a long period, making it a stumbling block to this otherwise eco-friendly disposal procedure (Esse *et al.* 2001). Thus, solutions in accelerating the composting process must be pursued. For this study, a speedy composting method was developed to recycle the OPEFB waste by converting it into compost. It is one of the most promising ways as it recycles large volume of waste

generated from oil palm mills into compost (Ravikumar *et al.* 2008). According to Fang *et al.* (1999), composting is an acceptable alternative for converting waste into useful eco-friendly fertiliser as it improves soil fertility. Compost contains a full spectrum of essential plant nutrients that can be released slowly over months and years unlike chemical fertilisers (Singh *et al.* 2010). Nutrients in the chemical fertilisers on the other hand, are usually easily washed away by rain falls shortly after being applied to plants causing it to pollute water bodies such as ponds and streams (Sharma & Chetani 2017).

Zeng *et al.* (2009) mention that microbial activity plays an important role in the transformation of organic material into stabilised composting product. Cellulolytic microorganisms that were able to produce cellulases and hemicellulases can be targeted and screened from soil samples using the selective media. Cellulolytic microorganisms which showed proficiencies in producing these enzymes were used to create an indigenous microbial consortium. The indigenous microbial consortium accelerated the composting process of OPEFB in our composting experiment. Based on a previous study, indigenous cellulolytic microorganisms were co-composed with lignocellulosic OPEFB could speed up the composting process to 30 days (Zainudin *et al.* 2013). Without such indigenous cellulolytic microorganisms, the normal composting process may take 60-90 days (Baharuddin *et al.* 2009).

Enzymes produced by microorganism are selected as they have functional significance which suggests much novel application especially for environmentally-friendly industrial purposes (Binod *et al.* 2013). Enzymes that are produced by microorganisms are an essential source of catalysts used in the industry (Ibrahim 2008). Microbial enzymes are favoured more than enzymes derived from plant and animals as they are more stable and have diverse properties (Alves *et al.* 2014). Enzymes have the ability to produce quantitative production and can be typically extracted at a low cost using downstream processes (Pandey *et al.* 2010).

1.2 Literature Review

1.2.1 Malaysia's Oil Palm Industry

Oil palm (*Elaeis guineensis*) is a variant of tropical tree crop which originated from West Africa. It was first introduced to Malaysia as an ornamental plant by the British in early 1870's. It was then commercially planted at the start of the year 1917 at the Tennamaran Estate, Selangor as a means to replace an unsuccessful planting of coffee bushes (WWF 2002). During the 1960's, an agricultural diversification programme was introduced by the government to reduce Malaysia's dependency on rubber and tin which caused a rapid large scale expansion of oil palm cultivation.

Oil palm has since then become the crop of choice for vegetable oil production in the world. For the first 4 years in the market, the young oil palms production is often small and of poor quality, sometimes it may even be not economic to be harvested (Verheye 2010). The production begins to reach a peak from the 6th year onwards until it reaches the age of 20-25 years where it declines in production (Rupani et al. 2010). In the industrial plantations, the oil palms will be felled and replanted after 25-30 years. Each fresh fruit bunch (FFB) harvested can weigh up to 50 kg. Verheye (2010) found that the oil is mostly concentrated in the fresh fruit pulp and kernels. The oil content in the fruit pulp is about 50-60% or 20-22% of the bunch weight while the oil content in the fruit kernels is 48-52% or 2-3% of the bunch weight. The freshly harvested FFB must be processed at the oil mill within 48hr after harvesting to avoid reduction of quality of oil produced (Global Palm Oil Conference IUF 2015). A bunch of FFB can have up to 2000 fruits which consist of hard kernel within a shell surrounded by fleshy mesocarp. The oil palm requires a high and year-round rainfall with little or no dry season and stable high temperatures. The vegetative growth of the oil palms is not affected by dry spells or temperatures below 18°C, however, it does reduce the yield obtained from the oil palm (Verheye 2010).

According to Sime Darby Plantation, 62% of Malaysia's land area is covered with one of the world's oldest rainforest (Sime Darby Plantation 2014). As of December 2015, Malaysia's palm oil plantation accounted for a total planted area of 5.64 million hectares. Figure 1.1 shows the anatomy of the area planted with oil palms in the different states in Malaysia. Sabah and Sarawak showed to have the highest oil palm planted area accommodating a percentage of 53% out of the total planted area in Malaysia.



Figure 1.1 Malaysia's oil palm planted area by state at December 2015 (Malaysian Palm Oil Board 2015)

The main producers of palm oil in the world are Malaysia and Indonesia (Lai *et al.* 2015). Indonesia and Malaysia cover about 83% of the total for the palm oil production together in the world as shown in Figure 1.2. The need to expand the production of palm oil was triggered by the high demand for oil palm to use as biodiesel production apart from its traditional use of food preparation. Palm oil is mostly consumed as food items or used to produce soaps, detergents and cosmetics (Lai *et al.* 2015). In recent years, palm oil is also used as feedstock for biodiesel where is it expected to be the leading crop of choice (Global Palm Oil Conference IUF 2015). In the year 2006, Malaysia launched Envodiesel, where 5% of palm oil olein with 95% petrodiesel is mixed together to create a blend. Palm oil can be used as fuel as it is sustainable unlike coal, oil and natural gasses which will run out eventually in term of technical difficulties or high expenses (Halsall 2011). Palm oil produces 1.64 exajoules (1 exa = 10^{18}) of energy when combusted directly

(Halsall 2011). Palm oil was not known to be used as biofuel in the early 1990s. It only began in late 1990 where only 40000 tonnes of palm oil was converted into biofuel. The amount was predicted to increase by 139% by the year 2020 (Lai *et al.* 2015). Therefore, palm oil will be a dominant vegetable oil throughout the world (Sabrina *et al.* 2012).



Figure 1.2 Ten largest producers of palm oil in the world for the year 2011 (Global Palm Oil Conference IUF 2015).

About 1 tonne of crude palm oil (CPO) is produced from 5.8 tonnes of FFB (Singh *et al.* 2010). A unique feature of oil palm is where two types of oil can be obtained from it. Palm oil from the flesh of the fruit and also palm kernel oil from the seed or kernel. For every 10 tonnes of palm oil obtained from the process, approximately 1 tonne of palm kernel oil can be obtained too (Malaysian Palm Oil Council n.d.). The two oils have very different composition. Palm oil that is obtained from the mesocarp mainly contains palmitic acid (C16:0), stearic acid (C18:0), linoleic acid (C18:2), and oleic acid (C18:1). Palm oil kernel, on the other hand, contains mainly of lauric acid (C12:0) (Malaysian Palm Oil Board 2011).

The food and agriculture organisations (FAO) define oil crops as those where oil and fats are extracted for producing food products for consumption and industrial purposes (Global Palm Oil Conference IUF 2015). Sime Darby plantations mention that oil palm is the most efficient oil crop in the world (2014). Oil palms bear an average oil yield of

4.08 tonne per hectare per year when compared to other oil-bearing crop as shown in Figure 1.3. Oil palm shows a big advantage as it produces more yield per hectare. Oil palm also uses less land usage when compared to other oil seed crop as shown in Figure 1.3. From the data obtained, palm oil has five more times than efficiency of its next closest competitor: rapeseed and ten times better than soybean. Due to this fact, palm oil has become the most valuable oilseed crop in the world. Not to mention the multiple uses of its oil both in food and other industries has helped boost its value higher.





1.2.2 Waste Production from the Oil Palm Industry in Malaysia

Oil palm has helped change the scenario of Malaysia in terms of agriculture and economy. Despite the many benefits of oil palms, it has contributes significantly to the environmental degradation from both the input and the output side of its activities (Abdullah & Sulaiman 2013). One of the critical environmental impacts of the input side from the oil palm industry is leaching of chemical fertilisers used into the ground water sources (Rahman *et al.* 2013). On the other hand, the output from the oil palm industry such as the dry biomass and liquid waste has another major environmental impact. The waste produced from processing of palm oil is creating a major environmental danger and entail high disposal cost every year. An oil palm mill can handle an average of 100 metric tonnes of fresh fruit bunches daily generating solid residues and liquid waste (Singh *et al.* 2010). Empty fruit bunches (EFB) and palm oil mill effluents (POME) are

the main waste produced as shown in Figure 1.4. EFB and POME together comprise more than 50% of the waste produced from the processing of oil palm. According to Hassan and Soom (2002), approximately 55 tonnes of dry fibrous biomass matter is produced from each hectare of oil palm plantation. Other than that, an average of $0.9-1.5m^3$ POME is generated from each tonne of crude palm oil (CPO) produced (Abdullah & Sulaiman 2013). Sustainable waste management must be sought after to tackle these waste.



Figure 1.4 Products/wastes from each bunch of fresh fruit bunch (Abdullah & Sulaiman 2013).

Palm oil mill effluents (POME) is the colloidal slurry which consist of water, oil and suspended solid produced during the sterilisation, crude oil clarification and cracked mixture separation processes in the oil palm mill (Yeoh *et al.* 2011; Madaki & Lau 2013). POME contains about 96% water, 0.6% oil, and 4% total solid (Hassan *et al.* 2005). It is considered to be the most expensive and difficult waste to manage as large volumes are generated at a time. POME which has a high content in organic matter is considered non-toxic since no chemicals were added during oil extraction process, however it is identified as an aquatic pollutant when discharged into water bodies (Khalid & Mustafa 1992). In Malaysia, integrating raw POME into anaerobic and aerobic ponds are common methods used in treatment before releasing treated POME into rivers. Unfortunately, the use of anaerobic and aerobic ponds requires vast area. Furthermore, methane which is a greenhouse gas that is 25 times more potent than carbon dioxide that is released during the anaerobic digestion of POME, making it the chief contributor of methane generator to the world global warming (Madaki & Lau 2013).

Oil palm empty fruit bunches (OPEFB) waste product is produced during the milling process when oil seeds are removed from the brunch for oil extraction (Ramli *et al.* 2015). OPEFB are usually either left to rot under oil palm trees to provide organic as mulch or incinerated by mills to obtain energy (Hasamudin & Soom 2002). However, burning the biomass contributed greatly to the air quality problem in Malaysia (Mahjoub *et al.* 2013). The burning of substance releases enormous amounts of carbon dioxide into the atmosphere which is believed to cause greenhouse effect and drastic changes to the environment (Khalil *et al.* 2007). The Malaysian Environmental Air Quality Regulation was introduced in 1978 to overcome the air quality problem caused by incineration of OPEFB. With the new regulation introduced, it prompted mills to find alternative methods of the disposal of the OPEFB waste produced. The environmental awareness also caused an interest among the researchers throughout the world to use the natural fibres to replace the man-made fibres in composite materials (Khalil *et al.* 2007).

1.2.3 Composition of Oil Palm Empty Fruit Bunch

In this study, the focus will be on the oil palm empty fruit bunches (OPEFB) waste generated by the process of manufacturing palm oil as it one of the major wastes produced. OPEFB is a lignocellulosic material which constitutes of three major parts of plants, namely cellulose, hemicellulose and lignin (Sreekala *et al.* 1997; Lai *et al.* 2015). The composition of lignocelluloses depends on the species, the growth condition, the different part of plants and their age. OPEFB of palm oil collectively contains roughly 53% cellulose, 30% of hemicellulose and 17% lignin (Baharuddin *et al.* 2009). Ramli *et al.* (2015) confirmed through testing that OPEFB had similar cellulose, hemicellulose and lignin content across most of its variants. Sreekala *et al.* (1997) investigated the chemical composition of the OPEFB fibre, it was found that the major constituents of the OPEFB were found to be cellulose with the lesser constituent being the lignin. Cellulose content is usually the highest content in all parts of the oil palm tree (Lai *et al.* 2015).

1.2.3.1 Cellulose

Cellulose is the most abundant biological compound in the ecosystem and can be found to be the main component of plants biomass. According to Schmit (2006), the typical fibre wall contains cellulose as its major chemical component. The linear structure of cellulose tends to form intra- or intermolecular hydrogen bonds which form the cellulose microfibrils. These microfibrils promote the aggregation into crystalline. As shown in Figure 1.5, the arrangement of the microfibrils is in an orderly manner. However, there are also regions within these microfibrils which are termed as amorphous due to the less orderly fashion in which they are arranged. When both the amorphous and crystalline cellulose form an arrangement, the resulting formation allows for a stronger and more rigid structure yet still very much flexible. The hydrogen bond also allows for cellulose to be insoluble in most solvents and play a partial role in giving it resistance against microbial degradation (Jørgensen et al. 2005). Generally, cellulose degradation is hydrolysed by exoglucanase and endoglucanase. Endoglucanase which is commonly called as cellulase cleaves the glycosidic bonds in the cellulose chain while exoglucanase hydrolyses the cellulose on the free ends of the cellulose chain (Evans et al. 2000). Cellulases are conventionally sought from fungi but there is a recent trend in sourcing such enzymes from bacteria which are both good cellulose producers as well as showing rapid growth rate.



Figure 1.5: Arrangement of fibrils, microfibrils, and cellulose in plant cell walls (Löbmann & Svagan 2017)

1.2.3.2 Hemicellulose

Hemicelluloses is a common name for a combination of different heterogeneous polysaccharides where xylan and glucomannans are the main components (Bastawde 1992). Hemicellulose has a lower degree of polymerization due to having side chains that can be acetylated and are generally amorphous when compared to cellulose (Jeffries 1994). Biologically, hemicellulose plays a vital role in the strengthening of the cell wall by interacting with cellulose as well as lignin in some cases. These structures in

correlation to the widely accepted models of the primary wall have been discussed and due to the nature of hemicellulose being more heterogeneous when compared to cellulose, a much more complex combination of enzymes are needed to degrade it (Schmit 2006). Some examples are endoxylanases, β -xylosidases, endomannanases, β -mannosidases, α -L-arabinofuranosidases and α -galactosidases (Jørgensen *et al.* 2005). The presence of these enzymes has been seen to more common in wood fungi when compared to bacteria (Schmit 2006).

1.2.3.3 Lignin

Known as the third main type of organic polymer which is present in the cell wall, lignin is a made up of dehydrogenative polymerization of three phenyl propane units namely pcoumaryl alcohol, coniferyl alcohol and sinapyl alcohol joined through ether bonds (Jeffries 1994). It functions to provide strength to withstand against compression where the cellulose provides the plant with flexible strength in the cell wall. Lignin also doubles as the bonding component which aids in connecting cells to enable the cell wall of the xylem to harden. This is an important aspect for the xylem to smooth out to transport water from the roots to the leaves. In terms of degradability, lignin is resistant to most microorganisms due to its phenylpropane units within the structure along with the recalcitrant linkages which form between them (Schmit 2006). However, lignin can be effectively broken down through the implementation of white-rot fungi which as a group of filamentous fungi that are capable of producing numerous oxidoreducatases which are enzymes capable of attacking the phenolic structures inside the lignin (Hatakka 2005).

1.2.4 Composting of Oil Palm Empty Fruit Bunch

The boost in palm oil production to the economic growth and rapid development had also contributed to environmental pollution due to the increased waste produced by the milling process (Rupani *et al.* 2010; Singh *et al.* 2010). Ways to treat and dispose of the waste must be sought after to reduce the harmful effect that it has inflicted onto the environment. Researches in Malaysian palm oil board (MPOB) has studied the conversion of oil palm empty fruit bunch (OPEFB) waste into paper-making pulp as OPEFB are categorised as fibrous crops which are lignocellulosic. MPOB has also experimented road making using fibre from OPEFB to produce stone mastic asphalt (SMA) (Hasamudin & Soom 2002). Another way to reduce municipal waste into alternative energy was to convert lignocellulose biomass into ethanol by hydrolysing the biomass into reducing sugar and

fermenting the sugars to ethanol (Sun & Cheng 2002). However, the cost for ethanol production through this method is relatively high.

Composting has often been proposed to reduce agricultural waste products from the oil palm process whereby fertilisers are created from it (Viel *et al.* 1987; Sabrina *et al.* 2012). Composting waste generated from the palm oil mills can be one of the sustainable solutions as it helps recycle the plant nutrients creating a fertiliser (Singh *et al.* 2010). As stated in the Malaysian Biomass Industry Action Plan 2020, the development of bio-fertiliser produced from composting OPEFB would be able to reduce the dependency of chemical fertiliser especially in the palm oil industry (Rahman *et al.* 2013).

Composting involves a biological process where heterogeneous organic wastes are converted into stable and humified organic matter known as compost by the mixed microbial population under controlled environment where optimum conditions such as moisture and aeration are provided (Ahmad *et al.* 2008). The organic matter produced can be stored and applied to the plantations without any environmental effect. Singh (2010) stated compost can be used to stabilise the organic matter for soil amendment and protect the environment from the effect of the oil palm waste products. Additionally, compost has the ability to improve the anion and cation exchange capability, organic matter and microorganism activity (Sharma & Chetani 2017). Composting will recycle the waste products produced from the oil palm industry into sustainable fertiliser which can be implemented back into the plantation. Compost contains a full spectrum of essential plant nutrients. They contain macro and micro nutrients which are released slowly over the months and years unlike synthetic fertilisers (Singh *et al.* 2010).

There are several types of composting methods such as aerated windrow composting, aerated static pile composting, and in- vessel composting. In aerated windrow composting, the composting materials are arranged into rows of long piles called 'windrow' and aerated by turning the piles manually or mechanically (Shi *et al.* 1999). This method is suited for composting large volumes of waste. Aerated pile composting can also be used to convert large amounts of waste into compost. Composting materials are mixed together into large piles instead of rows. The pile composting method is aerated by adding layers of bulking agent such as wood chips or newspaper into composting material so that air can have a passage to pass from the bottom to the top of the pile. Another method for pile composting is where the composting materials are placed right on top of piping systems

where air are drawn out from the pipes and delivered directly into the composting materials (Leton & Stentiford 1990). On the other hand, in-vessel composting involves the usage of a drum, silo or similar equipment where composting materials are fed into these vessels. In-vessel composting usually has a mechanism where it turns or agitates the materials placed inside it to provide proper aeration.

In-vessel composting method was chosen to be used to test the indigenous microbial consortium containing selected cellulolytic microorganism in degrading OPEFB during the composting experiment in this thesis study. In-vessel composting method was chosen as it requires less land space than other composting methods. It also enables to provide a controlled environment such as temperature, moisture, and airflow to each composting vessel experimented on during the composting study (Viel *et al.* 1987). In-vessel composting method is self-contained allowing minimal odour and leachate from the composting process.

According to Bernal et al. (1998), there are four important stages in composting. The first stage is when the raw materials are freshly laid out and have not gone into the thermophilic process yet. In this initial stage, an explosive growth of the mesophilic microorganisms rapidly uptakes soluble sugars and starches present in the composting material. The active microbial decomposition of the readily available sugar and starch generates heat which then causes the composting process to move into the thermophilic phase. This second stage is usually known as the thermophilic stage, where the thermophilic microorganisms work together to break down the proteins, fats, hemicellulose, and cellulose of the OPEFB rapidly causing the temperature of the compost pile to rise further. The third phase marks the fall in temperature as the biooxidative phase ends. In the fourth stage of the composting process, the compost goes through the maturation phase as lignin and other highly resistant compounds are decomposed. It is reported that the composting process must go through the thermophilic phase where the temperature rises above 55°C to sanitise the composting material from pathogens (Gea et al. 2005). Microbial activity plays an important role in the transformation of organic material into stabilised and sanitised composting product (Zeng et al. 2009). Ekinchi et al. (2006) mention that to obtain a successful stabilise and sanitised compost product requires maintaining a suitable environment for the process

which includes the moisture content, oxygen level, C: N ratio, nutrients and temperature given.

Compost protects soil against erosion, enhances soil water retention, reduces soil compaction and acidity, and enhances the soil biological and biochemical activity which establishes an ecological equilibrium in the soil (Sarkar *et al.* 2011). Although compost has its many benefits when applied, large volumes are required to provide sufficient nutrients for plants to grow as compost is low in nutrients. So, compost which is an organic fertiliser should be integrated with chemical fertilisers to give the optimal growth and development when applied to plants. A study by Ahmad *et al.* (2007) showed a higher yield and growth response from wheat and maize when organic and chemical fertiliser were applied together in comparison when the fertiliser was applied individually. The findings from the study implied that a combination of organic and chemical fertiliser could be a more effective and economical way to increase the growth and yield of crop plants than fertilisers alone.

Composting is the most suitable way to manage waste economically and environmentally as it reduces bulk volume of oil palm waste into useful compost which could improve and sustain the soil fertility (Mukhlis et al. 2013). For the composting experiment that was carried out in this study, an indigenous microbial consortium of cellulolytic microorganisms was inoculated into the composting material in the beginning of the process. OPEFB was also sprayed with partially treated POME throughout the process. Partially treated POME was added into the compost to maintain the optimum moisture and provide nutrients to the composting material throughout the experiment. A consortium of cellulolytic microorganisms was inoculated into the composting material to accelerate the composting process of oil palm empty fruit bunches (OPEFB). According to a previous study by Zainudin et al. (2013), indigenous cellulolytic and hemicellulolytic bacteria from untreated raw POME was co-composed with OPEFB to reduce the composting period. This co-composting experiment was completed in 40 days, shorter in comparison to 60-90 days as reported by Baharuddin et al. (2009). Baharrudin et al. (2009) composted OPEFB with partially treated POME that did not contain many indigenous microorganisms, thus a longer period could be required to complete the composting process. Apart from the inoculation of lignocellulosic microorganisms, Mukhlis et al. (2013) mentioned the importance of selecting effective

lignocellulolytic microorganisms as this could be a crucial step to have a successfully accelerated composting of OPEFB.

1.2.5 Cellulolytic Microorganisms

Cellulolytic bacterium such as Bacillus could be readily isolated from agricultural environments such as soil, compost and animal waste, and often shown to produce both cellulase and xylanase enzymes (Kim *et al.* 2012). Bacillus are gram positive bacteria that are rod-shaped from the member of phylum Firmicutes. *Bacillus* spp. could often be encountered constantly in soil environment, and grown rapidly when cultured in liquid medium, producing spores that are resistant (Shoda 2000). Spores formed by the *Bacillus* spp. can tolerate organic solvents, UV light and high-temperature environment making it able to endure adverse environmental conditions (Arrebola *et al.* 2010).

Bacillus spp. are known to produce various types of enzymes which can be used in various industries. Bacillus altitudinis and Bacillus licheniformis had been reported to have efficient production of cellulase when incubated in media supplemented with cellulose (Sreeja et al. 2013). In a previous study by Lokhande and Pethe (2017), Bacillus thuringiensis was tested to have the maximum cellulolytic activity when compared to the other 36 bacterial species isolated in the study. Khianngam et al. (2014) isolated 10 cellulase producing microorganisms from oil palm meal which is an oil palm by-product that is rich in fiber and used for the production of the animal feed. Eight out of these 10 isolates belonged to the Bacillus spp. while the remaining 2 were from Paenibacillus sp. and Lysinibacillus sp. respectively. In addition to the ability to produce cellulase, Bacillus spp. has also been described to be able to produce hemicellulase apart from cellulase, making them suitable for the composting of oil palm empty fruit bunch (OPEFB) (Kim et al. 2012). An article by Archana and Satyanarayana (1997) describe the ability of Bacillus altitudinis to produce xylanase enzyme during a soil state fermentation where it was grown on moist solid surface. Bacillus cereus have also showed the ability to hydrolyse carbon sources of CMC and xylan when tested on agar plate assays (Sen et al. 2012). Aside from cellulase and hemicellulase production, *Bacillus* spp. has a secondary metabolite that enhances the growth of plants and response defence systems in the host plant (Raupach & Kloepper 1998; Arrebola et al. 2010).

Fungi are also known to have the ability to degrade lignocellulose as they have efficient enzyme structures which branches onto two enzyme systems: the hydrolytic system, and

the oxidative and ligninolytic system (Sánchez 2009). The enzyme hydrolase produced by certain fungi can degrade polysaccharide, lignin and open phenyl rings. *Phanerochaete chrysosporium* is an example of a known lignocellulosic fungal strain. A study that was carried out by Martinez *et al.* (2004) tested the ability of *P. chrysosporium* to degrade lignocellulose efficiently. From the study, *P. chrysosporium* was able to degrade all major parts of the plant cell. There are only a few known fungi that are able to completely degrade lignin and leave behind crystalline cellulose with a bleaches appearance giving it the term 'white rot'. White rot fungi have the ability to degrade lignin as it involves an extracellular enzyme system which comprises of peroxidase of lignin, laccase and manganese and also hydrogen producing oxidase (Kirk *et al.* 1987). They are commonly found to be an inhabitant of the forest floor among the rotting wood.

Penicillium species is another white rot fungus that has the ability to degrade cellulose, hemicellulose and lignin naturally. Rodriguez *et al.* (1996) tested the ability of the *Penicillium* strains isolated from the forest where all the *Penicillium* strains were showed to be able to degrade cellulose, hemicellulose, and fractions of lignin. In the study of Rodriguez *et al.* (1996), *Penicillium* spp. exhibited to be one of the most powerful lignin degraders among several fungal strains tested. In another study, *Penicillium* was reported to exert high activity of lignocellulose when degrading barley straw (Rosgaard *et al.* 2006). *Penicillium* spp. were also tested for the production of cellulolytic and xylanolytic enzyme where it was compared to *Trichoderma reesei* by Krogh *et al.* (2004) and Penicillium showed to have better higher production activity. Chang *et al.* (2012) reported *Penicillium rolfsii* showed lignocellulolytic ability as it was able to hydrolyse oil palm residue from oil palm trunk to produce bioethanol when compared to other isolated fungus. *Penicillium citrinum* had also shown to produce high amounts of endoglucanases and exoglucanases that are responsible for the composting of cellulose (Suely *et al.* 2014).

Brown rot fungus usually just degrades the cellulose and hemicellulose of cell walls, leaving lignin undigested (Hatakka & Hammel 2010). During the breakdown of cellulose and hemicellulose in moist wood by brown rot fungus, hydrogen peroxide is released which causes the decomposition to increase making the wood appear shrunken with cubicle patterned cracks in a discoloured brown form, hence giving it the name 'brown rot'. *Aspergillus* species are examples of brown rot fungi. *Aspergillus* spp. could produce high level of xylanase which increased the bleachability of eucalyptus pulp in an

environmental process (Khonzue *et al.* 2011). In the previous study, *Aspergillus nomius* exhibited the ability to produce a cellulolytic and xylanolytic enzyme to degrade untreated oil palm trunk (Ang *et al.* 2015). *Aspergillus niger* was used a positive control for the cellulolytic assay during this thesis study as it is a known cellulase and xylanase producer which are usually used as an industrial microbial strain to produce large amounts of these enzymes (Biswas *et al.* 2014). *A. niger* is thermophiles that in temperature range of $35 \pm 5^{\circ}$ C. *A. niger* showed the ability to produce cellulolytic enzymes in an earlier study as an aid for the fermentation of waste (Santos *et al.* 2011). Thus, *A. niger* was chosen as the positive control for the cellulolytic assay as it is widely used for its cellulolytic enzyme production (Pandey *et al.* 2000).

Soft rot fungus produces cellulase that breaks down cellulose in wood which decolourises and causes crackling pattern to form which has similar appearance to brown rot fungus (Schmit 2006). It is typically found in the Ascomycetes sub-division. Soft rot fungus are less aggressive decomposers when compared to white rot fungus (Vane *et al.* 2005). Soft rot commonly occurs in wood that are exposed to extreme environments (Blanchette *et al.* 2004). Environments that are too wet usually inhibit the growth of brown and white rot fungus but does not limit the growth of soft rot fungus. Soft rot fungus usually favour conditions that are high in moisture and low in lignin content (Levy 1966).

1.2.6 Mechanism of Cellulolytic Microorganisms

Cellulose is usually degraded by cellulolytic enzymes produced from cellulolytic microorganisms. The cellulolytic enzyme consists of three basic type of enzymes which are endoglucanases, exoglucanases and ß-glucosidases (Khianngam *et al.* 2014). Endoglucanase produces oligosaccharides when it catalyses the initial disruption of the internal bonds that are within the cellulose crystalline structure. Endoglucanases are followed by exoglucanases, exoglucanases attack the nonreducing end of oligosaccharide chains that was produced by endoglucanase creating disaccharides. Disaccharides which are the cellobiose fragments of the cellulose is further hydrolysed by ß-glucosidases into glucose, hence completing the hydrolysis of cellulose into glucose (Ang *et al.* 2015).

Hemicellulose in plant cell walls consist of different polysaccharides, xylan represents the major component among the polysaccharides present in hemicellulose (Uffen 1997). According to Li *et al.* (2014), the hemicellulose layer in the oil palm empty fruit bunches (OPEFB) consist mostly of xylan. Hence, cellulolytic microorganisms that produce

xylanases were screened for to degrade the hemicellulose in OPEFB during the composting experiment in this study. Xylanases cleave the 1,4- β -D-xylosidic linkages in xylan randomly, producing xylose, arabinose, galactose and mannose (Collins *et al.* 2005). These sugars produced from the degradation are used by microorganisms for metabolism (Bisaria & Ghose 1981).

1.2.7 Screening for Cellulolytic Microorganisms

Carboxymethylcellulose (CMC) agar and Remazol Brilliant Blue-R (RBBR) xylan agar were chosen to be used for the initial screening of cellulolytic microorganisms from the collected soil samples. For CMC agar, pure isolates were cultured onto CMC agar, stained with 1% Congo red dye and destained with 1M NaCl to detect for visible clear zones. Previous studies have shown that Congo red has a better effectiveness to distinguish cellulase activity on solid medium (Gohel *et al.* 2014). During the staining of the CMC plates, the 1% Congo red dye binds with the polysaccharide which is CMC and forms a visible complex. Cellulases produced by the cellulolytic microbe inoculated into the plates break down the polysaccharide present in the agar medium, replacing the surrounding area of the isolate with smaller monosaccharides and disaccharides (Gupta *et al.* 2012). Therefore, a visible clear zone is formed around the cellulolytic microorganisms as they are unable to bind with the monosaccharides and disaccharides properly, which indicates the presence of cellulolytic bacteria.

Isolates that showed positive (zone-clearing) cellulolytic properties were then cultured on dyed xylan agar. The xylan was dyed with Remazol Brilliant Blue- R (RBB-R) before the screening. Those showed visible clear zones would indicate the presence of xylan degrading microorganisms (Meddeb-Mouelhi *et al.* 2014). Isolates with the ability to produce xylanase were able to degrade the dyed xylan present in the agar medium and form clear zones around the area. The formation of the clear zone on the RBBR-R xylan agar indicated the presence of the cellulolytic microbe that produces xylanase.

Isolated microbial strains which showed positive result in cellulase and xylanase productions were further tested their efficiency by measuring the reducing sugar production. There are two known methods for detecting the reducing sugars in cellulose and hemicellulose activity which are a 2-cyanoacetamide method and the 3,5-dinitrosalicylic acid (DNS) method. DNS is an alkaline solution that reacts with reducing sugars to form 3-amino-5-nitrosalicylic acid that can be read at 510 nm (Breuil & Saddler

1985). For this study, the 2-cyanoacetamide method was chosen to test for the reducing sugar produced by the isolated strains. Jurick *et al.* (2012) tested and found that the 2-cyanoacetamide method is capable of detecting the reducing sugars produced by cellulase as efficient as the DNS assay. Unlike the DNS method, 2-cyanoacetamide method does not require the use of potential carcinogenic and corrosive chemicals; therefore, the assay was not required to be conducted in the fume hood (Jurick *et al.* 2012). The 2-cyanoacetamide method was chosen for its accurate, rapid and non-toxic detection of cellulose and hemicellulose activity. Another advantage of the 2-cyanoacetamide method is that other enzyme activities can also be easily detected by changing the substrate in the growth medium. Carboxymethylcellulose (CMC) was used as a substrate in this study for detecting cellulase activity while xylan was used as a substrate to detect xylanase activity of the isolated microorganisms.

1.2.8 Plant Growth Promoting Microorganisms

Nitrogen, phosphorus, and potassium are the main essential nutrients required by plants to have a normal functioning cell system (Sharma & Chetani 2017). Hence, sufficient amounts of nutrient must be supplemented to plants to ensure optimum growth (Chen 2006). To reduce the adverse effect of chemical fertilisers, new agricultural practices have been developed where agricultural sectors have been creating a mix of chemical fertilisers with compost to help boost the plant growth. Soil samples from the natural environment normally contain plant growth-promoting microorganisms such as nitrogen-fixing microorganisms, phosphate solubilising microorganisms, potassium solubilising microorganisms, and indole acetic acid (IAA) producing microorganisms (López-Mondéjar et al. 2016). The presence of these microorganisms is known to aid in enhancing the plant growth when incorporated into the compost to form a bio-fertiliser. For the last couple of decades, plant growth-promoting microorganisms have been used increasingly in several regions of the world as it significantly increases the growth and yield of crops with its inoculation (Chen 2006). Most essential soil nutrients remain in insoluble forms in soils that are unavailable to plants causing nutrient deficiency (Nautiyal 1999). Therefore, plant growth promoting microorganisms play a vital role in converting the nutrients that are in insoluble forms into forms that the plants can absorb. For this study, isolated cellulolytic microorganisms were tested for plant growth promoting ability to enhance the compost produced.

Nitrogen is available molecularly from the atmosphere. However, it cannot be directly assimilated by plants. Nitrogen can only become available for plants through biological process that only nitrogen fixing microorganisms nitrogen can perform (Franche et al. 2009). Shortage of nitrogen availability to plants causes the formation of yellow leaves due to the insufficient supply of chlorophyll. The growth of the plants will also be affected as it will appear stunted as nitrogen is and essential nutrients required for cell division and enlargement. Bacillus spp., Rhizobia spp., Frankia spp. and Penicillium spp. are some of the examples of nitrogen fixing microorganisms (Moore & Becking 1963; Harper & Lynch 1984). In this study, Jensen's medium was used to screen the isolated cellulolytic microorganisms for the presence of nitrogen-fixing microorganisms. For Jensen's medium, nitrogen-fixing microorganisms grow well on nitrogen-free medium (Ahmad et al. 2008). The nitrogen-fixing microorganisms use the nitrogen gas available from the environment to perform cell protein synthesis. The cell protein produced from the synthesis mineralizes into the soil around the plants after the cells have died and therefore, contributes to the nitrogen availability to the plants. Due to the absence of nitrogen source in the agar medium, only nitrogen-fixing microorganisms will grow on the medium. To further confirm the presence of nitrogen-fixing activity, the medium with the nitrogen-fixing microbe could be flooded with bromothymol blue as a colour pH indicator (Gothwal et al. 2008). Bromothymol blue have a greenish appearance when it is in its original neutral pH state, yellow in the presence of acid and blue in an alkaline condition. For this study, the change of green to blue indicates the presence of nitrogenfixing activity by the isolated microorganisms.

Decaying dead plant debris releases phosphate to the soil in organic and inorganic forms that are not readily available for plants to absorb. Organic forms are commonly referred to living substances which contains carbon, proteins, lipids and vitamins while inorganic forms are usually found in non-living substances such as decaying plants which holds minerals (Dibb 2002). Phosphate containing chemical fertilisers that are applied to soil also becomes immobilised and unavailable to plants (Nautiyal 1999). This phenomenon is due to the strong bonds that phosphate has with calcium and magnesium in alkaline condition while having the same strong formation bonds with iron and aluminium in acidic soils (Sridevi *et al.* 2013). Due to the rapid formation of the strong bonds of other elements, plants are not able to absorb the nutrients causing a deficiency in phosphate. Phosphate solubilising plays an important role in maintaining the phosphate balance for

plants. Deficiency in phosphate causes plants to have smaller leaves, weak steam and slower growth development. Some examples of phosphate solubilising strains are *Pseudomonas* spp., *Bacillus* spp., *and Rhizobium* spp. (Peix *et al.* 2001). For this study, phosphate solubilising microorganisms were screened using Pikovskaya's agar media. Calcium phosphate added gave the agar medium a cloudy powdery appearance as it acted as the phosphate for the phosphate solubilising microorganisms to use as an energy source. From the assay, microorganisms that solubilise the phosphate present around its vicinity producing organic acids and showing a halo zone indicated phosphate solubilising activity.

Potassium is one of the three essentials nutrients required by plants. Potassium is available in the soil, however, like phosphate, it is also bound to other minerals making it unavailable for the plant. Potassium solubilising microorganisms can solubilise potassium through their hydroxyl and carboxyl groups that chelate the cations bound to phosphate thus releasing the low molecular weight soluble organic acids that are available for the plant (Chen *et al.* 2006). Potassium is an important nutrient as it helps in the stomatal activity, photosynthesis, protein synthesis and transport of water and nutrient in plants (Zhao *et al.* 2001). Some examples of potassium solubilising microorganisms are *Bacillus* spp. and *Pseudomonas* spp. (Sheng 2005). Aleksandrow agar can be used to screen for potassium solubilising microorganisms in this study. Potassium alumino silicates present in the agar medium acts as the source for potassium salts for the potassium solubilising microorganisms to secrete organic acids. From the test, clear zones that formed around the cellulolytic isolate inoculated into the medium indicate potassium solubilisation activity.

Indole acetic acid (IAA) is an auxin which is a plant hormone that affects the growth of plants. IAA is a product of L-tryptophan that can be produced by plant growth promoting microorganisms (Mohite 2013). IAA induces the growth of longer roots with increased number of hair roots which helps increase the nutrient uptake by plants (Datta & Basu 2000). *Bacillus* spp., *Pseudomonas* spp., *Acetobacter* spp., *Rhizobium* spp., *Aspergillus* spp., and *Penicillium* spp. are some strains that known to produce IAA (Yadav *et al.* 2011; Mohite 2013). During this study, cellulolytic isolates were grown in base medium supplemented with L-tryptophan and tested using Salkowski reagent to determine the concentration of indole-3-acetic acid produced. The IAA production ability by the

cellulolytic microorganisms was judge based on the amount of IAA produced where a higher concentration of IAA gave a more concentrated pink colour.

1.3 Research Aims and Objectives

The aim of this research was to screen and characterise cellulolytic microorganisms isolated from soil samples taken within the region of Sarawak capable of producing cellulase and hemicellulase (xylanase) for use and application as microbial consortium in the composting of oil palm empty fruit bunch (OPEFB) and evaluating the effectiveness of the compost produced. The specific objectives set out to achieve the research aim are:

i. To isolate and characterise cellulolytic microorganisms from soil samples collected in Sarawak;

ii. To evaluate the efficiency of isolated cellulolytic microorganisms in the formulation and development of indigenous cellulolytic microbial consortium; and

iii. To investigate the effectiveness of the indigenous microbial consortium developed in the composting of OPEFB into compost through in-vessel composting method and the compost produced in growth enhancing of oil palm seedlings via pot trial method.
1.4 Thesis Outline

This thesis presented is divided into 5 chapters which are listed as follow:

Chapter 1- Introduction and Literature Review

Chapter 1 provides a brief introduction to the background and a broad review of essential literature related to the study which was reported and described in previous studies by other researchers. The aim and objectives of the research were also conferred in this chapter.

Chapter 2- Screening and Characterisation of Cellulolytic Microorganisms from Sarawak Soil

Chapter 2 describes a detailed screening, isolation and identification of cellulolytic microorganisms which were obtained using selective media. This chapter focuses on the isolation of cellulase and xylanase producing microorganisms from soil samples taken from various location in Sarawak. Isolated cellulolytic microorganisms were also tested for plant growth promoting ability in this chapter.

Chapter 3- Selection of Cellulolytic Microorganisms Based On Enzyme Activity and Optimisation of Selected Microorganisms

Chapter 3 present details on the efficiency of the isolated cellulolytic microorganisms in producing cellulase and xylanase activity using 2-cyanoacetamide method. The purpose of this chapter was to evaluate the effectiveness of the isolated cellulolytic strains as potential candidates to form an indigenous microbial consortium for the composting of OPEFB. The isolates capable of producing comparable cellulolytic activity with positive control strain were selected and used for subsequent experiments. This chapter also focuses on the optimisation of the selected cellulolytic microorganisms to form the microbial consortium.

Chapter 4- Production of Compost Via In-Vessel Composting of Oil Palm Empty Fruit Bunch Using Cellulolytic Microorganisms and Its Enhancement in the Growth of Oil Palm Seedlings

Chapter 4 describes the production of compost by composting of OPEFB using indigenous microbial consortium containing selected cellulolytic microorganisms through in-vessel composting method. During this chapter, the efficiency of the compost in the enhancement of the growth of oil palm seedlings was also described. The knowledge obtained from the study can lead to further investigation along this line of research such as large-scale field application using the indigenous microbial consortium of cellulolytic microorganisms produced in this study.

Chapter 5- General Conclusions, Recommendations, and Future Works

Chapter 5 provides a succinct overview of the findings from the experimental studies presented. Perspective on future research possibilities within this field as well as recommendations is conferred in this chapter.

Chapter

2

SCREENING AND CHARACTERISATION OF CELLULOLYTIC MICROORGANISMS FROM SARAWAK SOIL

2.1 Introduction

The growing oil palm industry in Malaysia has been ever increasing over the last decade due to escalating requirement of palm oil (Rosenani *et al.* 2016). The amount of waste produced by the oil palm industry has been accumulating causing an environmental hazard. Oil palm empty fruit bunch (OPEFB) is one of the major produced by the oil palm industry. OPEFB is a lignocellulose waste produced abundantly in the oil palm industry that consists mainly of cellulose, hemicellulose, and lignin (Mukhlis *et al.* 2013). Ways to tackle the growing waste production is very much needed.

Fungi and bacteria are the main natural agents of cellulose degradation (Jahangeer *et al.* 2005). Cellulolytic microorganisms were targeted in this study due to its ability to degrade cellulose in OPEFB. Microorganisms are favoured for degrading cellulose over the use of chemical treatments due to the harmful effects of the chemical treatments might inflict on the environment (Li *et al.* 2014). Cellulolytic microorganisms that produce cellulase and hemicellulase have exhibited the ability in previous studies to degrade cellulose into simple sugars (Li *et al.* 2014).

In this chapter, soil samples collected from several locations within Sarawak were screened for cellulolytic microorganisms. Soil are chosen as source for obtaining cellulolytic microorganisms as it contains an immense diversity of microorganisms that have yet to be discovered (Torsvik & Øvreås 2002). Cellulolytic microorganisms can be found naturally in the forest soil as they usually consist of the microbial community that degrades most of the organic material that settles on the forest floor (López-Mondéjar *et al.* 2016). Selective media was used in this study to isolate cellulolytic microorganisms from soil samples. Carboxymethylcellulose (CMC) agar was used to detect microorganisms that produced cellulase while dyed xylan agar was used to detect microorganisms which produce xylanase. Cellulolytic microorganisms were selected in this study as a potential candidate to degrade the cellulose and hemicellulose in the OPEFB waste.

The experimental focus of this chapter was to screen, isolate and identify cellulolytic microorganisms from soil samples taken from the forest grounds of Borneo as a potential source to degrade OPEFB waste produced from the oil palm industry.

2.2 Materials and Methods

2.2.1 Sampling Locations and Collections

Forest soil was targeted in this study as cellulolytic microorganisms can be found naturally in the forest ground (López-Mondéjar *et al.* 2016). Upon the authorisation by Sarawak Biodiversity Center, fresh soil samples were collected from 4 targeted locations within the region of Sarawak. A total of 12 soil samples, 3 soil samples from each targeted location were collected in January 2016. Each sample was collected aseptically using sterile tools and placed into its own individual resealable bag which was stored in a polystyrene box containing ice packs at the collection site. The soil samples were then kept in 4°C cold room for not more than 3 days before performing the isolation procedures. The targeted sites are as follow:

- (i) Melikin, Serian.
- (ii) Semonggok Forest Reserve, Kuching.
- (iii) Kubah National Park, Kuching.
- (iv) Kampung Panchor Hot Spring, Kuching.

2.2.2 Screening for Potential Microorganisms from Soil Samples

Agar plate assay was used to the initial screening for potential cellulolytic, nitrogen-fixing, phosphate solubilising, potassium solubilising, and indole acetic acid (IAA) producing microorganisms from collected soil samples during this research. Preparation of agar plates for the assays can be found in Appendix Section 1.

2.2.2.1 Serial Dilution

Serial dilution technique was used during the initial screening for cellulolytic microorganisms. A ten-fold serial dilution was carried out for each soil sample using sterilised deionised water.

2.2.2.2 Screening for Potential Cellulolytic Microorganisms Using Selective Media(I) Screening for Cellulolytic Microorganisms

Carboxymethylcellulose (CMC) agar was used to screen for cellulolytic microorganisms. The dilution series of the diluted soil samples chosen for the initial screening was 10⁻³, 10⁻⁴, 10⁻⁵, and 10⁻⁶. Dilution 10⁻³, 10⁻⁴, 10⁻⁵, and 10⁻⁶ was chosen for the initial screening because dilution of 10⁻¹ and 10⁻² gave an overgrowth of

microorganism on medium plates. From the diluted soil samples prepared, $100 \ \mu L$ of diluted soil samples were spread onto the CMC agar evenly. The spread plates were sealed to avoid contamination and incubated under aerobic conditions at $37^{\circ}C$ for 3 days. After incubation, the plates were stained with 1% Congo red dye (Fisher Scientific, USA) for around 30 minutes followed by destaining process using 1M sodium chloride (NaCl) for approximately 15 minutes. The plates were then observed for formation of clear zones around the colonies visible on the CMC plates. CMC plates that showed clear zone were kept in 4°C for isolation purposes.

(II) Screening for Xylanase Producing Degrading Microorganisms

RBBR-xylan agar was used to screen for xylanase producing microorganisms. Pure isolated cellulolytic microorganisms that were obtained from the soil samples using CMC agar which showed potential in degrading cellulose were cultured onto the xylan agar. Plates were sealed to avoid contamination and incubated under aerobic conditions at 37°C for 3 days. After 3 days of incubation, the plates were observed for the formation of clear zones around the colonies. Plates that showed clear zone were noted down. The screening analysis for xylanase producing microorganisms were performed in triplicates for each isolate.

(III) Screening for Nitrogen-Fixing Bacteria

Jensen's medium was used to screen for nitrogen fixing bacteria. Pure cellulolytic isolates were inoculated onto fresh Jensen's medium agar and sealed to avoid contamination. The plates were incubated under aerobic conditions at 37°C for 3 days. After 3 days of incubation, the plates were observed for the formation of clear zones around the colonies. Plates that contained cellulolytic microorganisms that showed clear zone were noted down. The screening analyses for nitrogen fixing microorganisms were performed in triplicates for each isolate tested on.

(IV) Screening for Phosphate Solubilising Bacteria

Pikovskaya agar was used to screen for nitrogen fixing bacteria. Pure cellulolytic isolates were inoculated onto fresh Pikovskaya agar plate. The inoculated were sealed to avoid contamination and incubated under aerobic conditions at 37°C for 3 days. After 3 days of incubation, the plates were observed for the formation of clear zones around the colonies. Plates that contained cellulolytic microorganisms that showed

clear zone were noted down. The screening analyses for phosphate solubilising microorganisms were performed in triplicates for each isolate tested on.

(V) Screening for Potassium Solubilising Bacteria

Aleksandrov agar was used to screen for nitrogen fixing bacteria. Pure cellulolytic isolates were inoculated onto fresh Aleksandrov agar and sealed to avoid contamination. The inoculated plates were incubated under aerobic conditions at 37°C for 3 days. After 3 days of incubation, the plates were observed for the formation of clear zones around the colonies. Plates that contained cellulolytic microorganisms that showed clear zone were noted down. The screening analyses for potassium solubilising microorganisms were performed in triplicates for each isolate tested on.

(VI) Screening for Indole Acetic Acid (IAA) Producing Microorganisms

Isolated cellulolytic microorganisms were tested for the ability to produce indole acetic acid (IAA) using the Swalkowski method. Broth cultures of pure cellulolytic microorganisms were inoculated into basic growth medium with an addition of tryptophan (1 g/L). They were incubated at 37°C for 4 days in an incubator shaker (Sartorious Stedim Biotech, Germany). After incubation, the cultures were spun down using a centrifuge machine and 1 mL of the supernatant was transferred aseptically into a universal bottle with proper labelling. In the darkness of the fume hood, 1.5 mL of Swalkoski reagent (2% 0.5 FeCl₃ in 35% HClO₄ solution) was pipetted into each universal bottle containing the supernatant. The solutions were mixed well and were left to incubate in the dark for 75 minutes before reading it at 525 nm using a spectrophotometer. If the solution turns pink, it indicates production of IAA. However, if the solution turns yellow, it shows that no IAA production occurred. The screening analysis for IAA producing microorganisms were performed in triplicates for each isolate tested on, the results of the study were presented as mean values for the three separate plates.

2.2.3 Isolation of Pure Culture

Each strain that exhibited clear zones surrounding its colony on carboxymethylcellulose (CMC) agar plate was isolated and transferred to its own fresh agar medium. For each bacterial strain, it was picked using sterilised loops, while for fungal strain, isolation was performed using autoclaved 5-mm straw. Subsequent sub-culturing was performed until pure isolates were obtained. The pure isolates on the CMC agar plates were stained and destained to re-confirm the ability of the isolates to degrade CMC. Isolates that exhibited the formation clear zones were kept in 4°C for further analysis. Isolates that did not show clear zones were discarded by decontamination using an autoclave (Hirayama, Japan).

2.2.4 Storage and Preservation

2.2.4.1 Bacterial Culture

Long term storage using glycerol stock was used in this study for maintenance and preservation of the isolated cellulolytic bacteria. For the maintenance of the bacterial culture in glycerol stock, 800 μ L of overnight grown cultures in nutrient broth (NB) (Himedia, India) were inoculated into 2.0-mL cryogenic vials (Nest Biotechnology, China) containing sterilised 800 μ L of 60% glycerol (HMBG, Germany) to obtain a final glycerol concentration of 30% (v/v). The stocks were mixed carefully by pipetting the mixture using a micropipette (Eppendorf, Germany). The cryogenic vials were labelled, sealed with paraffin film, and placed into a cryo box and kept in a -80°C freezer (Thermo Scientific, USA). Long term storage for each bacterial strain was deposited in 10 replicates. When reviving the stored bacteria, a sterile toothpick or inoculation loop was used to scrape off the solid ice in the cryogenic vials and transfer culture onto fresh nutrient agar (NA) (Himedia, India) which would then be incubated at 37°C.

For short term storage of the bacterial culture, the pure isolates were grown on NA (Himedia, India) plates for 24 hr and stored at 4°C until further requirement for the next experiment. Short term storage for each bacterial strain was also in 10 replicates.

2.2.4.2 Fungal Culture

Long term storage using glycerine agar was used in this study for maintenance and preservation of the isolated cellulolytic fungus. For the maintenance of the fungal culture in glycerine agar, 50 mL of glycerol stock was incorporated into the 1 L of potato dextrose agar (PDA) (Himedia, India) before autoclaving. The autoclaved PDA and glycerol stock solution (HMBG, Hamburg, Germany) was aseptically dispensed into a sterile 90-mm petri dish in a laminar flow cupboard and allowed to cool and harden for 45 minutes before keeping it aside. Pure fungal cultures were cultivated onto their own respective plates for 7 days at 37°C, allowing enough mycelium to develop before storing them. Straws 5 mm in diameter were cut into smaller size slightly shorter than the depth of 2.0 mL cryogenic vial, autoclaved and oven dried. The fungal cultures were stored by plugging the freshly grown cultures into the sterilised straws using forceps until the straw was filled. Once the straw was filled with the mycelium, it was placed into the sterilised 2.0-mL cryogenic vials (Nest Biotechnology, China). The cryogenic vials were labelled and placed into a cryo box where it was kept in the -80°C freezer (Thermo Scientific, USA). Long term storage for each fungal strain was deposited in 10 replicates. When reviving the stored fungus, a sterile toothpick or inoculation loop was used to poke out the fungal cultures out from the straws onto the fresh PDA medium and incubated at 37°C.

Fungal cultures were also preserved using barley grains for storage of 6 months. Clean universal bottles were first filled with 5 g of organic barley and rinsed twice with distilled water. The distilled water was drained and replaced with 1.5 mL of Potato dextrose broth (PDB) (Himedia, India). The universal bottles containing the barley and PDB were autoclaved twice before allowing them to cool prior to inoculation. A straw of 5 mm diameter was used to transfer two 7-day old fungal cultures into the universal bottles containing the sterilised barley and PDB. They were allowed to grow for 2 weeks before placing them in the 4°C fridge. Storage for each fungal strain in barley was performed in triplicates. When reviving the stored fungus, a sterile forceps was used to transfer two grains of barley colonized with the fungal culture onto a sterile PDA medium.

For short term storage of fungal culture, the pure cultures were grown on PDA plates for 3 days and stored at 4°C until required. Short term storage for each fungal strain was in 10 replicates.

2.2.5 Molecular Identification

2.2.5.1 Preparation of Culture Broth for Molecular Identification

(I) Bacterial Culture

Cellulolytic bacterial isolates were inoculated into sterilised nutrient broth (Himedia, India) and shaken overnight at 37°C in an incubator shaker (Sartorious Stedim Biotech, Germany).

(II) Fungal Culture

Cellulolytic fungal isolates were inoculated into sterilised potato dextrose broth (Himedia, India) and shaken overnight at 37°C in an incubator shaker (Sartorious Stedim Biotech, Germany).

2.2.5.2 DNA Extraction

The freeze-and-thaw method (Griffin *et al.* 2002) was used to lyse the isolated microorganisms. From the incubated broth, 1000 μ L of the culture broth were transferred into sterilised 1.5-mL microcentrifuge tubes respectively and centrifuged for 10 minutes at 7500 rpm using a microcentrifuge (Eppendorf, Germany). The supernatant was decanted carefully to avoid the disposal of the pellet. When a pellet was observed, it will be re-suspended in 1 mL of sterilised deionised water by mixing using a micropipette. The mixture was then mixed using a vortex (Barnstead Thermolyne, USA) to ensure the pellet was completely re-suspended. The suspension was then subjected to the freeze and thaw cycle of -80°C freezer (Thermo Scientific, USA.) and hot water bath (Memmert, USA). A total of 4 cycles were carried out in a programme illustrated in Table 2.1.

After subjecting the sample to freeze and thaw, it was spun down at 10,000 rpm for 5 minutes using a microcentrifuge (Eppendorf, Germany). The supernatant from the sample was carefully transferred into a fresh sterile microcentrifuge tube, labelled, and stored in the -20°C until further requirement. Only the supernatant was kept because the pellet that contains mostly cellular debris would be decontaminated and discarded.

Freeze samples at -80°C using the -80°C freezer						
Thaw samples at 85°C using a hot water bath						
Freeze samples at -80°C using the -80°C freezer						
Thaw samples at 85°C using a hot water bath						

Table 2.1. DNA extraction - Freeze-and-thaw cycles.

2.2.5.3 DNA Amplification

The microbe's DNAs obtained from the extraction process were amplified by using polymerase chain reaction (PCR). MyTaq Red Mix (Bioline, USA) were used along with primers (IDT, USA) to prepare the master mix for PCR amplification. The PCR master mix contained the following: MyTaq Red Mix (25μ L), forward primer (1μ L, 20μ M), reverse primer (1μ L, 20μ M), and sterilised MilliQ water (18μ L).

For the bacterial DNA samples, universal primer 8 F and 1541 R was used, while for the fungal DNA samples, universal primer ITS 4 and ITS 5 was chosen. The concentrated primers were mixed with sterile distilled water according to manufacturer's instructions. The primer sequences of the primers used are listed in Table 2.2.

For bacterial DNA samples	
8 F	5'-AGA GTT TGA TCC TGG CTC AG-3'
1541 R	5'-AAG GAG GTG ATC CAG CCG CA-3'76
For fungal DNA samples	
ITS 4	5'-TCC TCC GCT TAT TGA TAT GC-3'
ITS 5	5'-GGA AGT AAA AGT CGT AAC AAG G-3

Table 2.2. Primer sequences.

The prepared master mix was aliquot into labelled sterilised PCR tubes using a micropipette. Each PCR tube contained 45 μ L master mix and 5 μ L of crude DNA extract was added into its own respective tube. DNA amplification was performed in a PCR machine (Eppendorf, Germany) which was pre-programmed with the following cycling conditions: initial denaturation at 95°C for 1 minute, followed by 29 cycles of denaturation process at 95°C for 15 seconds, annealing process at 55°C for 15 seconds

and extension process at 72°C for 15 seconds, followed by a final elongation process at 72°C for 5 minutes. The final PCR products were hold on at 4°C.

The final PCR product was analysed by gel electrophoresis. Each PCR product was pipetted into wells made from 1% (w/v) agarose gel (Vivantis Technologies, USA) with 3μ L of Redsafe (iNtRON Biotechnology, South Korea) for staining of the nucleic acid, allowing the visualization under UV light. The gel electrophoresis was run at 120V for 30 minutes for the bacterial PCR product while 100V for 50 minutes for the fungal PCR product. The voltage and time to run for the bacterial and fungal PCR product were different due to the difference in the base pair, bacteria have about 1500 bp while fungus has 600-700bp. The Bio- Rad Gel Documentation EQ System for DNA/RNA Gel Photos was used to observe and capture the gel image of the PCR product.

2.2.5.4 DNA Analysis

PCR product that showed clear bands in the gel electrophoresis was sent for DNA sequencing. The sequencing of the DNA of the PCR product was performed by First Base Laboratories Sdn Bhd, Selangor, Malaysia. The identification of the microorganisms was then identified by matching the unknown gene sequence obtained to known gene sequences in the GenBank database using Basic Local Alignment Search Tool (BLAST) at the National Centre for Biotechnology Information (NCBI). Molecular Evolutionary Genetics Analysis Version 7.0 (Mega 7) was used to create a phylogenetic tree by using the maximum likelihood method based on Tamura-Nei model (Tamura & Neil, 1993).

2.3 Results and Discussion

2.3.1 Isolation of Celluloytic Microorganisms

Forest soil was targeted as it contains diverse microorganisms that are responsible for producing enzymes that decompose the cellulose and hemicellulose present in the soil. As reported by López-Mondéjar (2016), approximately 12% of the bacterial colonies isolated from forest soil samples showed to have cellulase producing ability.

The microorganism of interest was distinguished by its ability to form clear zones around the colony that were cultured on the selective media. For the initial screening of cellulolytic microorganisms, diluted soil samples were first cultured onto carboxymethylcellulose (CMC) agar to detect cellulase producing microorganisms. Cultures that exhibited clear zone (Figure 2.1) was isolated and subsequent sub-cultured onto fresh growth medium until pure culture was acquired. These pure isolates were then re-cultured onto CMC agar in triplicates to confirm its cellulase production. Isolates that showed visible clear zones during the CMC plate assay were further cultured onto xylan agar to test if it could degrade the dyed xylan (Figure 2.2), suggesting the presence of xylanase.



Figure 2.1 Clear zones indicating the presence of cellulase in CMC agar (A: before staining) and (B: after staining).



Figure 2.2 Pure isolates showing xylan-degrading activities on dyed xylan agar: (A) control; (B) clear zone formed by bacterial isolates; and (C) fungal isolates when xylanase could be produced.

Isolates which exhibited clear zone in the plate assays expressed a strong evidence of positive results. Isolates which showed clear zone formation on the CMC agar and xylan agar were inoculated onto new CMC plates and kept in glycerol stock which was placed in the -80° freezer for long term storage. A total of 24 cellulolytic isolates were isolated from the soil samples collected from the targeted region which showed positive results in degrading cellulose while 10 out of the 24 isolates demonstrated to degrade xylan. A summary of the ability of the isolates to degrade cellulose and hemicellulose can be seen in Table 2.4.

2.3.2 Screening for Plant Growth-Promoting Microorganisms

N-fixing microorganisms: Jensen's medium was used to screen the isolated cellulolytic microorganisms for the presence of nitrogen-fixing microorganisms. Bromothymol blue in its original pH state turns from green (Figure 2.3A) to blue (Figure 2.3B) when in an alkaline condition which indicates the presence of nitrogen-fixing activity. Bromothymol blue changes from green to yellow when exposed to an acidic environment (Figure 2.3C). The colour change from green to blue indicates the presence of nitrogen-fixing activity by the isolated microorganisms. The results showed that 5 out of 24 isolates showed positive results in performing nitrogen fixing (Table 2.3).

		Degradation	n properties				
	Code		Hemi-	N-	PO4 ³⁻ -	К-	.
		Cellulose	cellulose	fixing	solubilizing	solubilizing	IAA
	S 1	+	-	-	-	-	+
	S2	+	+	-	+	-	+
	S3	+	+	-	-	-	+
	S4	+	-	-	-	-	+
	S5	+	-	-	+	-	+
ites	S6	+	-	+	-	-	+
isola	S 7	+	-	-	-	-	+
terial	S 8	+	+	+	-	-	+
Bact	S9	+	+	-	-	-	+
	S10	+	-	+	+	+	+
	S11	+	-	-	-	-	+
	S12	+	-	-	-	-	+
	S13	+	-	-	-	-	-
	S14	+	+	-	-	-	-
	F1	+	-		-	-	+
	F2	+	-	+	-	-	+
	F3	+	-	-	-	-	+
S	F4	+	+	-	-	-	-
solat	F5	+	+	-	-	-	-
ngali	F6	+	+	+	-	-	-
Fu	F7	+	-	-	-	-	-
	F8	+	+	-	-	-	+
	F9	+	-	-	-	-	+
	F10	+	÷	-	-	-	+

Table	2.3	Summary	of	isolated	cellulolytic	microorganisms	with	cellulolytic
degrad	ling	properties a	nd	growth p	romoting pro	operties.		



Figure 2.3 Cellulolytic microorganisms grown on Jensen's medium with bromothymol blue dye – (A) neutral state – green; (B) blue – alkaline, and (C) yellow – acidic.

Cellulolytic isolates which exhibited the presence of activity of interest were noted down. From the isolated 24 isolates, five isolates (S6, S8, S10, F2 and F6) showed to have the N-fixing abilities, three isolates (S2, S5 and S10) with phosphate-solubilising property, one isolate (S10) with potassium-solubilising property, and 16 isolates (S1-S12, F1-F3, and F8) with the abilities to produce IAA (Table 2.3).

Phosphate-solubilising microorganisms: Pikovskaya agar was used for the detection of phosphate-solubilising microorganisms. Phosphate is available in both organic and inorganic forms in soil (Gupta *et al.* 2012). However, plants are only able to utilize phosphate that is in the readily available inorganic phosphate form. Therefore, phosphate-solubilising microorganisms play a vital part in converting the phosphate present in the soil into readily available form for the uptake by the plants. As it can be seen from Figure 2.4 (A), phosphate-solubilising activity was detected from the isolated cellulolytic microorganisms when a clear zone was formed around the vicinity of the isolate due to the solubilisation of phosphate. Only 3 out of 14 bacterial isolates showed phosphate-solubilising properties (Table 2.3). None was observed in the fungal isolates.

Potassium-solubilising microorganisms: Aleksandrow agar was chosen as a selective medium to detect potassium-solubilising activity. Potassium-solubilising microorganisms convert insoluble potassium available in the soil into a form that plants can access. As reported by Prajapati and Modi, the use of potassium-solubilising microorganisms were found to resolve potassium from insoluble minerals, therefore, it has been used as an agricultural improvement (2012). From Figure 2.4 (B), clear zones were formed around

isolates that solubilise potassium within its vicinity. From this experiment, only one isolate could solubilise potassium (Table 2.3).



Figure 2.4 Cellulolytic Microorganisms grown on Pikovskaya agar (A) and Aleksandrow agar (B) showed clear zone indicates solubilising property.

IAA-producing microorganisms: According to Mohite (2013), indole acetic acid (IAA) producing microorganisms play an important role in stimulating and facilitating plant growth. Growth promoting microorganisms are able to produce IAA where it is a product of L-tryptophan metabolism. Swalkowski method was used to test the ability of the isolated cellulolytic microorganisms to produce IAA. Such microbe can be detected when the supernatant and reagent turns pink.

From this experiment, 16 isolates out of the 24 isolated cellulolytic microorganisms showed positive results in producing IAA as shown in Table 2.4. Based on the absorbance reading at 525 nm, F8 have the highest production of IAA (Abs = 0.270), followed by S3 (Abs = 0.170) and S2 (Abs = 0.168).

	Bacterial is	olates	Fungal isolates								
Isolate codes	Production of IAA	Absorbance reading (Abs)	Isolate codes	Production of IAA	Absorbance reading (Abs)						
S 1	+	0.141 ± 0.002	F1	+	00.143±0.002						
S2	+	0.168 ± 0.001	F2	+	0.052 ± 0.003						
S 3	+	0.170 ± 0.002	F3	+	0.117±0.003						
S4	+	0.156 ± 0.003	F4	-	-						
S5	+	0.157 ± 0.002	F5	-	-						
S 6	+	0.109 ± 0.005	F6	-	-						
S 7	+	0.110 ± 0.001	F7	-	-						
S 8	+	0.149 ± 0.001	F8	+	0.270 ± 0.002						
S9	+	0.109 ± 0.003	F9	-							
S10	+	0.135±0.009	F10	-							
S11	+	0.059 ± 0.005									
S12	+	0.111 ± 0.002									
S13	-	-									
S14	-	-									

Table 2.4 Indole acetic acid (IAA) produced by cellulolytic microorganisms.

Absorbance value expressed as mean \pm standard deviation (n=3).

2.3.3 Molecular Identification of the Isolated Bacteria and Fungus

From the initial screening, 14 bacterial isolates and 10 fungal isolates were identified using molecular identification via DNA sequencing. In the molecular approach, the crude DNA of the 24 isolates was successfully extracted using the freeze and thaw method. Universal primers, forward primer 8 F (5'-AGA GTT TGA TCC TGG CTC AG-3') and reverse primer 1541 R (5'-AAG GAG GTG ATC CAG CCG CA-3') was chosen for the bacterial isolates identification while forward primer, ITS 4 (5'-TCC TCC GCT TAT TGA TAT GC-3') and reverse primer ITS 5 (5'-GGA AGT AAA AGT CGT AAC AAG G-3') were used for the molecular identification of the fungal isolates.

Based on the gel electrophoresis results, PCR product size obtained from the bacterial isolates were about 1300-1500 base pairs (Figure 2.5) while the 550-650 base pairs were obtained for the fungal isolates (Figure 2.6).



Figure 2.5 DNA bands of the PCR product extracted from the cellulolytic bacterial isolates. Label M represents 1kB DNA Marker; C represents negative control; and S1-S14 represents the DNA bands produced from the bacterial isolates.



Figure 2.6 DNA bands of the PCR product extracted from the cellulolytic fungal isolates. Label M represents 1kB DNA Marker; C represents negative control; and F1-F10 represents the DNA bands produced from the fungal isolates.

BLAST was used to compare the PCR product sequence obtained in this study to sequences available in the GenBank database, it then calculates the statistical significance of the sequence of the unknown entered to the closest known match sequence (Kent 2002). The sequence matches of the bacterial isolates and the fungal isolates are summarised in Table 2.5 and Table 2.6, respectively.

			Accession		Onorr	
Code	Strain Nama	Classest match	no. of	Base	Query	Similarity
Coue		Closest maten	closest	Pair	(9/)	(%)
			match		(70)	
S1	Bacillus sp. S1	Bacillus subtilis	KY285264.1	1455	94	96
52	Pacillus on S2	Bacillus	VV126612 1	1/1/	00	02
52	<i>Buculus</i> sp. 52	licheniformis	KA420042.1	1414	90	93
62	Pacillus on S2	Bacillus	VV270740 1	1 / 0 1	02	06
55	<i>Buculus</i> sp. 55	amyloliquefaciens	KA3/9/40.1	1461	92	90
S.4	Stanbulo co cours SA	Staphylococcus	VT11007 1	1 / 0 0	20	08
54	Staphylococcus 54	saprophyticus	K 144100/.1	1400	89	98
S5	Bacillus sp. S5	Bacillus cereus	KF500919.1	1427	94	96
56	Pacillus on S6	Bacillus	VV750696 1	1207	02	08
30	<i>Bucilius</i> sp. 30	toyonensis	K1/30080.1	1387	95	90
S7	Bacillus sp. S7	Bacillus subtilis	KY123860.1	1496	90	98
58	Racillus sp. S8	Bacillus	KC172054-1	1500	88	07
30	<i>Ducilius</i> sp. 36	altitudinis	KC172034.1	1500	00	21
S9	Bacillus sp. S9	Bacillus aerius	JX981918.1	1329	97	97
\$10	Pseudomonas	Pseudomonas	KC700300 1	1381	08	00
510	aeruginosa	Aeruginosa	KC790500.1	1301	90	"
S 11	Racillus on S11	Bacillus	E1402045 1	1245	00	05
511	<i>Buculus</i> sp. 511	licheniformis	1 1 4 9 3 0 4 3 . 1	1545	77	95
\$12	Bacillus	Bacillus	KU146560 1	1/31	83	00
512	methylotrophicus	methylotrophicus	KU140300.1	1431	85	77
\$13	Racillus sp S13	Bacillus	KV287785 1	1/150	87	96
515	<i>Duculus</i> sp. 515	atrophaeus	K120//0J.1	1439	0/	70
S14	Bacillus sp. S14	Bacillus subtilis	KX426659.1	1357	95	95

Table 2.5 Molecular identification of the cellulolytic bacterial isolates.

Identification was based on sequencing data obtained using NCBI nucleotide BLAST database.

The closest match of the sequences was based on the query cover percentage (%) in base pairs and identity percentage (%) which were no less than 80% for both criteria. The accession number, base pair, query cover and, similarity of the closest match species to the sequences were recorded in Table 2.5 and Table 2.6.

		Accession no.	Base	Query	Similarity
Code	Closest match	of closest	Pair	cover	(%)
		match	1 411	(%)	(70)
F1	Fusarium fujikuroi	KT192406.1	557	97	99
F2	Fusarium proliferatum	GU066655.1	555	98	99
F3	Fusarium oxysporum	KF751873.1	556	97	99
F4	Penicillium coprophilum	AF033469.1	579	97	99
F5	Trichoderma harzianum	KY750325.1	618	98	99
F6	Penicillium citrinum	KM979730.1	548	99	99
F7	Penicillium oxalicum	GU078430.1	587	96	99
F8	Aspergillus nomius	DQ467991.1	593	98	99
F9	Endomelanconiopsis endophytica	GQ469968.1	579	96	99
F10	Penicillium rolfsii	KM246748.1	581	98	99

Ta	ab	le	2.	6	Μ	0	lec	ul	ar	ic	len	ntif	fic	ati	ioi	1 0	f 1	the	c	ell	ul	ol	yti	ic	fur	ıgal	is	ola	tes.
																							•			0			

Identification was based on sequencing data obtained using NCBI nucleotide BLAST database.

Only isolates which showed sequences that had a similarity of \geq 99% were named using the closest match species name identified using BLAST. In the past, the sequenced isolates which showed 97.0% to 99.5% of similarity when BLAST was named with the closest match (Stackebrandt & Goebel 1994). However in recent years, more papers have recommended and supported the idea where a similarity of \geq 99% was required to name the strain to the closest match species name (Keswani & Whitman 2001; Clarridge & Alerts 2004). Thus, strain name for each isolate which had a similarity of <99% were named after the genus followed by its code name for convenience and were referred as such throughout the thesis. Out of the 14 bacterial isolates that were searched against the GenBank database, 12 isolates were found to be from the *Bacillus* species. The identified strains were listed in Table 2.7. The isolates identified consist of the following genus: *Bacillus, Staphylococcus, Pseudomonas, Fusarium, Endomelanconiopsis, Penicillium, Aspergillus, and Trichoderma*.

No	Label	Domain	Strains identity obtained	Number of isolates	Strain species
1	S1/S7/ S14	Bacteria	Bacillus subtilis	3	Bacillus spp.
2	S2/S11	Bacteria	Bacillus licheniformis	2	Bacillus spp.
3	S3	Bacteria	Bacillus amyloliquefaciens	1	Bacillus spp.
4	S4	Bacteria	Staphylococcus saprophyticus	1	Staphylococcus spp.
5	S5	Bacteria	Bacillus cereus	1	Bacillus spp.
6	S6	Bacteria	Bacillus toyonensis	1	Bacillus spp.
7	S8	Bacteria	Bacillus altitudinis	1	Bacillus spp.
8	S9	Bacteria	Bacillus aerius	1	Bacillus spp.
9	S10	Bacteria	Pseudomonas aeruginosa	1	Pseudomonas spp.
10	S12	Bacteria	Bacillus methylotrophicus	1	Bacillus spp.
11	S13	Bacteria	Bacillus atrophaeus	1	Bacillus spp.
12	F1	Fungus	Fusarium fujikuroi	1	Fusarium spp.
13	F2	Fungus	Fusarium proliferatum	1	Fusarium spp.
14	F3	Fungus	Fusarium oxysporum	1	Fusarium spp.
15	F4	Fungus	Penicillium coprophilum	1	Penicillium spp.
16	F5	Fungus	Trichoderma harzianum	1	Trichoderma spp.
17	F6	Fungus	Penicillium citrinum	1	Penicillium spp.
18	F7	Fungus	Penicillium oxalicum	1	Penicillium spp.
19	F8	Fungus	Aspergillus nomius	1	Aspergillus spp.
20	F9	Fungus	Endomelanconiopsis endophytica	1	<i>Endomelanconiopsis</i> spp.
21	F10	Fungus	Penicillium rolfsii	1	Penicillium spp.

 Table 2.7 The closest match of strains identity obtained for the isolated isolates after

 comparison with GenBank database.

The molecular phylogenetic trees were constructed based on the sequences obtained from cellulolytic and hemicellulolytic bacterial and fungal isolates of the current study and known closely matched species' sequences available in the GenBank. The phylogenetic tree for cellulolytic and hemicellulolytic bacteria and fungi in Figure 2.7 and Figure 2.8 shows the position of the isolates. The phylogenetic trees were generated using Molecular Evolutionary Genetic Analysis version 7.0 (MEGA7) software with the adaption of the maximum likelihood method based on Tamura-Nei model (Tamura & Neil 1993). A phylogenetic tree was constructed to organize the biological data obtained from the isolates to structure classifications and to have an insight of the evolution of the isolates have gone through (Kumar *et al.* 2016). A phylogenetic tree is a diagram that paints the lines of the evolutionary origin of the different species, genes or organisms which derive from a common ancestor (Baum 2008).



Figure 2.7 The molecular phylogenetic tree showing the position of the cellulolytic bacterial isolates.

*Data based on their partial sequence which was constructed using the Molecular Evolutionary Genetic Analysis version 7.0 (MEGA7) software with the adaption of the maximum likelihood method based on Tamura-Nei model.



Figure 2.8 The molecular phylogenetic tree showing the position of the cellulolytic fungal isolates.

*Data based on their partial sequence which was constructed using the Molecular Evolutionary Genetic Analysis version 7.0 (MEGA7) software with the adaption of the maximum likelihood method based on Tamura-Nei model.

2.4 Summary

This chapter focused on the screening and characterization of cellulolytic microorganisms. The cellulolytic microorganisms were successfully isolated from the soil samples collected from the local targeted area using selective media such as Carboxymethylcellulose (CMC) agar and Remazol Brilliant Blue-R (RBBR) xylan agar. The desired microorganisms were distinguished by its ability to exhibit visible clear zones around the isolates that were cultured on the selective media. Out of the 24 isolates studied, all 24 isolate was able to produce cellulose, while only 10 isolates were produced xylanase.

The cellulolytic microorganisms isolated out from the soil samples were further analysed for their ability to be plant growth-promoting microbes – based on their abilities to make available three of the plant macronutrients namely the nitrogen (N), phosphate (P) and potassium (K), as well as their abilities to produce indole acetic acid (IAA), a plant growth enhancing hormone. Jensen's medium was used to detect microorganisms that were able to perform nitrogen fixation. Pikovskaya agar was chosen for the detection of phosphate-solubilising activity by the cellulolytic microorganisms while Aleksandrow agar was used as a detector for potassium-solubilising microorganisms. The isolated microorganisms were also tested for IAA production using the Salkowski method. From the 24 isolates, five isolates showed to have the abilities to perform nitrogen fixing, three isolates were able to solubilise phosphate, one isolates exhibited the ability to solubilise potassium, and 16 isolates produced IAA.

The molecular identification of the 24 isolates was identified by using the freeze and thaw method for the DNA extraction and polymerase chain reaction (PCR) for the amplification of the DNA. Based on the comparison with the GenBank database, the identified isolates were classified to be members of the genus of *Bacillus, Staphylococcus, Pseudomonas, Fusarium, Endomelanconiopsis, Penicillium, Aspergillus and, Trichoderma.* Further evaluations for the 24 different strains were carried out and the results would be described in the following chapters. The efficiency of the 24 different strains as potential cellulolytic and hemicellulolytic microbes to degrade oil palm empty fruit bunches (EFB) was evaluated in Chapter 3.

Chapter

3

SELECTION OF CELLULOLYTIC MICROORGANISMS BASED ON ENZYME ACTIVITY AND OPTIMISATION OF SELECTED MICROORGANISMS

3.1 Introduction

Environmentally friendly ways to overcome the increasing oil palm empty fruit bunch (OPEFB) waste produced by the growing oil palm industry in Malaysia must be seek. Composting of OPEFB is recommended to manage wastes that are large in volume (Zainudin *et al.* 2013). Composting reduces the bulk volume of OPEFB waste into useful compost which is economically and environmentally friendly (Mukhlis *et al.* 2013). Nevertheless, as OPEFB is lignocellulosic material, a longer time would be required to degrade completely (Baharuddin *et al.* 2009). OPEFB contains roughly 53% cellulose, 30% of hemicellulose and 17% lignin (Baharuddin *et al.* 2009). Hence, the addition of cellulolytic microorganisms would help to accelerate the composting of OPEFB (Zainudin *et al.* 2013).

Microbes that are cellulolytic have the potential to degrade OPEFB as they produce certain enzymes such as cellulase and hemicellulase. The cellulase and hemicellulase produced by theses microbes have the ability to help in the hydrolysing process of the cellulose and hemicellulose in OPEFB into simple sugars (Ang *et al.* 2015). In this chapter, isolated cellulolytic microorganisms that were obtained from collected soil samples that were discussed in Chapter 2 were tested for its ability degrade cellulose and hemicellulose using the 2-cyanoacetamide method. The 2-cyanoacetamide method can be used to detect glucose level produced by the cellulolytic microorganisms when it degrades cellulose and hemicellulose into simpler sugars (Jurick *et al.* 2012). Carboxymethylcellulose (CMC) was used as a substrate in this study for detecting cellulase activity while xylan was used as a substrate to detect hemicellulase activity in the cellulolytic isolates. Isolates which showed the highest cellulase and hemicellulase activities were chosen for further testing as a potential candidate to form an indigenous microbial consortium of degrading microorganisms to degrade OPEFB waste for further experiments.

Antagonism among the selected cellulolytic strains that were chosen based on their degrading abilities was evaluated and discussed in this chapter. The selected strains were grown together to detect any mutual inhibitions in a test for potential antagonism. This test was conducted to prevent any growth inhibition that to be inflicted among the selected cellulolytic microorganisms when inoculated together as a consortium (Ghai *et al.* 2007).

A study by Hamzah *et al.* (2012) indicated that physical parameters such as initial medium pH and incubation temperature would play important roles to promote microbial biomass production. Each of the selected cellulolytic strains were optimised for its ideal growth condition in different parameters. The selected strains were placed into differently tuned growth mediums and temperature environments to find its ideal growth condition based on its growth rate which would be the number of cells that were produced per volume of growth medium.

The aim of this chapter was to analyse the efficiency of the selected microorganisms for its cellulase and hemicellulase activities. Strains which showed to have the highest activity were further tested for antagonism activity and optimised to find its ideal incubation condition. Final selected strains were used to form an indigenous microbial consortium for further investigations.

3.2 Materials and Methods

3.2.1 Measurement of Enzyme Activity of Isolated Cellulolytic Microorganisms

3.2.1.1 Detection of Glucose Level Produced by Cellulolytic Microorganisms

The cellulase and hemicellulase activities of the isolated cellulolytic microorganism were detected using 2-cyanoacetamide method adapted from Jurick *et al.* (2012). This method is based on measurement of glucose level liberated from the Carboxymethylcellulose (CMC) and xylan substrates by cellulases and hemicellulases respectively.

(I) Preparation of Standard

Glucose standards were first prepared by dissolving glucose (Sigma-Aldrich, USA) into 50 mL sodium acetate (Sigma-Aldrich, USA) buffer pH5.0 with the concentration of 0, 5, 25, 50, 100, 200, 400, and 800 mg/mL⁻¹. Each concentration was measured in triplicates into clean sterile universal bottles with a corresponding blank. The glucose standard with 0 mg/mL⁻¹ concentration was used as a blank for the assay. The blank value for the standard was subtracted from the readings to adjust the background. The water bath was first set to 80°C. From the prepared glucose solution, 200 μ L of it was added to 400 μ L of 1% 2-cyanoacetamide (Acros Organics, USA) solution and 2mL of 100mM borate (Sigma-Aldrich, USA)

with pH9.0 in sterile universal bottles. The solution was boiled for 10 minutes in the water bath (Memmert, USA) after being mixed thoroughly using a vortex (Barnstead Thermolyne, USA). The boiled solution was allowed to cool to room temperature and read at 276 nm using a multiplate reader (BioTek Synergy HT, USA) to determine its absorbance.

(II) Analysis On the Glucose Level Produced by Cellulolytic Microorganisms

For the determination of sugars produced by the microorganisms in the different substrate, the selected microorganisms were first cultured into carboxymethylcellulose (CMC)/xylan broth media. The basic broth medium used can be referred to Table 2.1 where 8 g/L of CMC/xylan substrate was measured depending on the requirement. Overnight bacterial broth culture were inoculated into the CMC/xylan broth and incubated for 3 days before performing the cellulase/hemicellulase bioassay. As for fungus, 5 days old fungal broth culture was inoculated into the CMC/xylan broth and incubated for 5 days before performing the cellulase/hemicellulase bioassay. The cultures were incubated in the condition of 37°C with shaking at 120 rpm in an incubator shaker (Sartorious Stedim Biotech, Germany). After incubation, 200 µL of the culture was added to 400µL of 1% 2-cyanoacetamide solution and 2 mL of 100 mM borate with pH9.0 in sterile universal bottles. The solution was boiled for 10 minutes in the water bath after being mixed thoroughly using a vortex. The boiled solution was allowed to cool to room temperature and read at 276 nm using a multiplate reader (Sartorious Stedim Biotech, Germany) to determine its absorbance. CMC broth was used as a blank for cellulase bioassay while xylan broth was used for the hemicellulase bioassay. The absorbance readings obtained from the cellulase/hemicellulase bioassay were used to determine the glucose produced by the cellulolytic microorganisms. The absorbance readings from the bioassay and the standard curve of glucose were used to calculate the by cellulolytic microorganisms. The glucose produced the isolated cellulase/hemicellulase analyses were performed in triplicates for each sample.

3.2.2 Screening for Antagonistic Activity of Selected Strains

Antagonistic activity of the selected based on the bioassay for cellulase and hemicellulase activity were tested for antagonism against each other. Antagonistic activity was experimented based on the protocol explained by Ghai *et al.* (2007). The selected strains used for further investigation are as follow:

- (i) *Bacillus* sp. S2 (Bacterium)
- (ii) *Bacillus* sp. S8 (Bacterium)
- (iii) *Penicillium citrinum* F6 (Fungus)
- (iv) *Penicillium rolfsii* F10 (Fungus)
- (v) *Penicillium coprophilum* F4 (Fungus)

Antagonism properties of selected strains were tested by using the dual culture technique. Fresh nutrient agar (NA) plates were chosen to test for the antagonism activity. The NA plates were divided into two sections where each selected strain was grown with each other. Selected bacterial strains were streak onto one side of the agar plates using a sterile loop while 5 mm diameter of 5 days old mycelia was placed on the other side of the agar plate. The plates were incubated at 37°C for 3 days and observed for antagonistic activity by the selected strain on each other. The antagonistic activity analysis was performed in triplicates for each plate.

3.2.3 Optimisation of Bacterial and Fungal Strains

Selected cellulolytic strains were cultured in different conditions to test for its optimal growth environment. Fresh broth cultures for the selected cellulolytic strains were prepared before carrying out the experiment. Bacterial strains were cultured by inoculating them into individual 10 mL nutrient broth media while fungal strains were inoculated into individual 10 mL potato dextrose broth media. The cultured nutrient broth was incubated overnight while the fungal strains were grown for 5 days in an incubator shaker with shaking at 120 rpm with the temperature set at $37^{\circ}C \pm 1^{\circ}C$.

3.2.3.1 pH

The influences of different pH ranging from pH5.0 to pH8.0 with an interval of 0.5 were tested on the selected cellulolytic strains. The initial pH of the growth medium used was attuned with the use of 1 N NaOH and 1 N HCl. The bacterial strains were grown overnight in tuned nutrient broth incubated at $37^{\circ}C \pm 1^{\circ}C$ with an agitation of 120 rpm while the fungal strains were grown for 5 days in tuned potato dextrose broth incubated

at the same temperature and agitation as the bacterial strain. The optical density of the bacterial cultures was read at 600 nm in triplicates using spectrophotometer (Thermo Scientific, USA). Sterilised clean nutrient broth was used as a blank for the setting of the spectrophotometer. For the fungal cultures, the experimental broth medium was dried in the oven to obtain its mycelial dry weight g/L. The technique for obtaining the mycelial dry weight for detecting the growth of fungal culture was modified from the method mentioned by Zhu *et al.* (2008). Fresh PDB growth medium was used as a reference blank for the fungal cultures.

3.2.3.2 Temperature

The optimal temperature of the selected cellulolytic strains was tested by incubating the selected cellulolytic strains in different temperature ranging from 35 to $45^{\circ}C \pm 2$ with an interval of 5°C and detecting the influence of it. The bacterial strains were grown overnight in nutrient broth incubated at $37^{\circ}C \pm 1^{\circ}C$ with an agitation of 120 rpm while the fungal strains were grown for 5 days in potato dextrose broth incubated at the same temperature and agitation as the bacterial strain. The optical density of the bacterial cultures was read at 600 nm in triplicates using spectrophotometer (Thermo Scientific, USA). Sterilised clean nutrient broth was used as a blank for the setting of the spectrophotometer. For the fungal cultures, the experimental broth medium was dried in the oven to obtain its mycelial dry weight g/L. The technique for obtaining the mycelial dry weight for detecting the growth of fungal culture was modified from the method mentioned by Zhu *et al.* (2008). Clean PDB growth medium was used as a reference blank for the fungal cultures.

3.2.4 Microbial Growth Rate of Selected Microorganisms

3.2.4.1 Bacterial Strain

Fresh broth cultures for the selected bacterial cellulolytic strains were prepared before carrying out the experiment. Bacterial strains were cultured by inoculating them into individual 10mL nutrient broth media. The cultured nutrient broth was incubated overnight in an incubator shaker and shaken at 120 rpm with the temperature set at $37^{\circ}C \pm 1^{\circ}C$. The optical density of the cultures for this experiment was read at 600 nm using spectrophotometer (Thermo Scientific, USA). Sterilised clean nutrient broth was used as a blank for the setting of the spectrophotometer.

(I) Indirect Growth Measurements

To begin the indirect growth measurement of the selected bacterial strains, 10 mL of overnight culture prepared was poured into its own individual shake flask containing 90 mL of sterilised nutrient broth. It was mixed well and its initial optical density, time zero, was taken immediately and noted down. The optical density reading of each bacterial strain was taken every 1 hr interval in triplicates for the first 8 hr right after the inoculation. The final optical density reading was taken at time 24 hr. For each reading, 1000 μ L of the culture broth was pipette into its own cuvette and discarded immediately after taking its reading. For each time point, the optical density reading of the cultures was taken in triplicates.

(II) Direct Growth Measurements

The colony forming unit of the selected bacterial strains was obtained through the direct growth measurement method and total plate count. The direct growth measurement was carried out along with the indirect growth measurements. From time zero to time 8 hr, 1 mL of culture broth from each bacterial strain cultured in shake flask was taken every 1hr interval and a ten-fold dilution was carried out using sterilised deionised water. To perform the ten-fold dilution, 9 mL of sterilised deionised water was allocated into seven, 15-mL centrifuge tube for each sample taken from the shake flask culture, and labelled 10^{-1} , 10^{-2} , 10^{-3} , 10^{-4} , 10^{-5} , 10^{-6} , and 10⁻⁷ respectively. The 1 mL culture broth taken from the shake flask culture was pipetted into tube 10⁻¹ and mixed well by pipetting up and down with a minimum of 5 times. From tube 10⁻¹, 1 mL of the mixture was transferred to the tube labelled 10^{-2} , and mixed well with the same method as mentioned before. These steps were repeated until tube 10⁻⁷. For the dilution 10⁻⁵, 10⁻⁶, and 10⁻⁷, 100 µL was transferred respectively onto each nutrient agar plate in triplicates, and spread evenly using a sterile cotton swab. The plates were incubated at 35°C for one day in the incubation room. In the second day, the distinct colonies on each plate were counted. The colony forming unit per millilitre (CFU/mL) was calculated according to the equation 3.1 below:

Equation 3.1 CFU/mL = a number of colonies per mL plated/ Dilution.

3.2.4.2 Fungal Strain

(I) Indirect Growth Measurements

For the indirect growth measurement of the selected fungal strains, the dry weight of the mycelium produced from the incubating the fungal strains in broth was observed. The indirect growth measurement of the fungal strains was modified from the protocol explained by Zhu *et al.* (2008). Potato dextrose broth (PDB) was first prepared by sterilizing 100 mL of in 250 mL baffled flasks. The sterilised PDB were inoculated with the selected 5-day old fungal agar plate culture. For the transfer of fungal cultures into PDB, the straw technique was used to transfer one 5-mm diameter fungal culture into the broth. The inoculated baffled flask was incubated at 35°C and shaken at 120 rpm for 5 days. After 5 days of incubation, the mycelial biomass from the broth culture was harvested by vacuum filtration with pre-fried and weighed Whatman No.1 filter paper. The mycelium on the filter paper was dried at 80°C for 24 hr and weight using a weighing balance. The growth of each fungal strain was estimated by the dry weight of the mycelium. Indirect measurement of the selected fungal strains was carried out in triplicates.

(II) Direct Growth Measurements

Direct growth measurement of each selected fungal cellulolytic strain was carried out by measuring its growth diameter on potato dextrose agar over the span of 7 days. Fresh fungal culture plates were first prepared by inoculating pure cultures onto individual potato dextrose agar plates using the sterile straw technique and grown for 5 days at 35°C in the incubator. For the direct growth measurement of fungal strains, the sterile straw technique was adapted to transfer 5mm diameter of fungus from the fresh culture prepared onto sterile potato dextrose agar plates and incubated at 35°C. The growth diameter of the fungus during incubation day 5 was observed. Direct measurement of the selected fungal cellulolytic strains was carried out in triplicates.

3.2.5 Data Analysis

The bioassay for detecting the glucose level produced by cellulolytic microbes, optimisation and microbial growth rate of selected microbial strains were performed in triplicates. The results of these studies were presented as mean values for the triplicates.

3.3 Results and Discussion

3.3.1 Measurement of Enzyme Activity by Selected Cellulolytic Microorganisms

The strains isolated from soil samples that were identified were tested for its cellulose and hemicellulose activity using the 2-cyanoacetamide method to detect the glucose level produced. Carboxymethylcellulose (CMC) was used as a substrate in this study for detecting cellulose activity while xylan was used as a substrate to detect hemicellulase activity in the isolated microorganisms. A standard curve was first created using glucose with acetate buffer of pH5.0 to make a concentration of 5, 25, 50, 100, 200, 400, 800 and 1000 mg mL⁻¹. Acetate buffer used for the construction of the standard curve as it is a commonly used buffer for enzyme assay including cellulase (Jurick *et al.* 2012). Construction of the standard curve yielded a R^2 value of 0.9985 and a linear equation of y = 0.0015x + 0.0083. Standard curve of glucose constructed using the 2-cyanoacetamide method can be referred to in Appendix Section 2.

The 2-cyanoacetamide bioassay was used to detect the reducing sugar produced by cellulase and hemicellulase producing microorganisms. The glucose value produced by the isolated microbial strains was calculated based on the equation y = 0.0015x + 0.0083 and absorbance reading obtained from the bioassay. *Aspergillus niger* (NBRC 33023) was used as positive control for this study due to it is commonly used to produce cellulase and hemicellulose commercially (Ang *et al.* 2015). Figure 3.1 summarises the reducing sugar produced by each isolated strain from cellulose and hemicellulose respectively.

Cellulose is the most abundant carbohydrate found in nature and is a major structural material in plant cell walls (Chen 2014). About 45-50% of the plant cell wall of the oil palm empty fruit bunch (OPEFB) consists of cellulose (Baharuddin *et al.* 2009). Therefore, the CMC which derived from cellulose was used as a substrate to detect cellulase activity of the isolated microbial strains to seek out which strains has the best degrading capability to degrade cellulose. As illustrated in Figure 3.2, isolated strains that produced the most amount of glucose from degrading CMC was *Bacillus* sp. S2 (185.13 mg mL⁻¹) followed by *Penicillium citrinum* F6 (162.24 mg mL⁻¹), *Penicillium rolfsii* F10 (160.47 mg mL⁻¹), *Bacillus* sp. S8 (140.47 mg mL⁻¹), and *Aspergillus nomius* F8 (129.80 mg mL⁻¹). *Bacillus* sp. S7 and *Fusarium oxysporum* F3 showed to have the least glucose production which was 3.13 mg mL⁻¹ during the cellulase bioassay. For the cellulase activity, *Bacillus* sp. S2 and *Bacillus* sp. S8, *P. citrinum* F6, and *P. rolfsii* F10

showed the ability to produce higher amount of glucose than the positive control, *A. niger* which had the glucose value of $134.02 \text{ mg mL}^{-1}$.



■ Cellulase ■ Hemicellulase

Figure 3.1 Glucose produced by cellulase and hemicellulase producing microorganisms as detected using 2-cyanoacetamide method.

The hemicellulose of empty fruit bunches in oil bunch consists mainly xylan (Li, Ng & Wu 2014). Therefore, the bioassay for testing hemicellulase activity of the isolated strains was carried by replacing the substrate to xylan. The hemicellulase bioassay showed only 10 out of the 24 different strains having the ability to degrade both CMC and xylan which was shown in Figure 3.1. Among the 10 strains tested for its hemicellulase activity, *Penicillium rolfsii* F10 exhibited the highest level of glucose at 153.58 mg mL⁻¹, followed by *Bacillus* sp. S8 (144.47 mg mL⁻¹), *Penicillium citrinum* F6 (128.46 mg mL⁻¹), *Bacillus* sp. S2 (124.45 mg mL⁻¹), *Aspergillus nomius* F8 (120.47 mg mL⁻¹), *Penicillium coprophilum* F4 (107.80 mg mL⁻¹), *Trichoderma harzianum* F5 (77.13 mg mL⁻¹), *Bacillus* sp. S9 (63.13 mg mL⁻¹), *Bacillus* sp. S14 (48.47 mg mL⁻¹), and lastly, *Bacillus* sp. S3 (35.80 mg mL⁻¹). *A. niger* which was used as the positive control for the hemicellulase activity obtained a glucose value of 184.02 mg mL⁻¹ which was higher than all the strains tested.
From the bioassay for testing the cellulase and hemicellulase activity of the isolated microorganisms, the 14 cellulolytic isolates which produced the lowest glucose level from degrading CMC was excluded from further analysis. Cellulose composes the first layer of the oil palm fibre waste, therefore, the microorganisms with the highest cellulose degrading ability were prioritised. After confirming 10 cellulolytic microorganisms with the highest cellulose degrading abilities, the hemicellulase productions of these isolates were observed. From the observation, the top five cellulolytic strains which produced among the highest cellulase activities were also found to exhibit the high hemicellulase productions. Therefore, the top five isolated strains namely *Bacillus* sp. S2 and *Bacillus* sp. S8, Penicillium citrinum F6, Aspergillus nomius F8, and Penicillium rolfsii F10 showed the most prominent results in producing the highest cellulase and hemicellulase activities during the experiment were chosen as to form an indigenous microbial consortium for the purpose of composting of oil palm empty fruit bunches. Indigenous consortium of a mixed bacterial and fungal strains was often chosen as it seemed more effective in composting waste such as reducing the time required for such process (Mukhlis et al. 2013).

The chosen strains consisted of a mix population of bacterium and fungus from the genus *Bacillus, Penicillium* and *Aspergillus* which have been discussed in Chapter 2 for their genetic identification. Similar species had also been previously studied by other researchers on cellulolytic activities. *Bacillus* spp. isolated from agriculture environments had been reported to exhibit the ability to convert cellulose and hemicellulose into soluble sugars (Kim *et al.* 2012). *Penicillium* spp. had also been studied more recently as it has been described to be a promising strain in the bioethanol industry for its capability in producing cellulases (Schneider *et al.* 2016; Amore *et al.* 2013). *Aspergillus* spp. are known cellulolytic microorganisms as they have been industrially used to produce large amounts of cellulolytic strains were therefore in line with the previous findings, and further test were therefore conducted before using them as inoculum to accelerate the composting process.

3.3.2 Screening for Antagonism Activity Among Selected Strains

Dual culture plate assay was adapted to be used for this study to test for the presence of antagonism among the selected cellulolytic strains. Dual culture plate assay is an established technique used to discover antagonism among strains. The selected strains, Bacillus sp. S2 and Bacillus sp. S8, Penicillium citrinum F6, Aspergillus nomius F8, and Penicillium rolfsii F10 were tested for the antagonism effect on each other. The selected strains were cultured among each other on fresh nutrient agar (NA) and incubated for 3 days at 37°C. The growth of the selected strains when grown with each other was visually observed from the dual plate assay. The dual plate assays were observed for the presence of growth inhibition. As shown in Figure 3.2, no antagonistic activities were observed on each selected strain when grown together. The selected cellulolytic strains Bacillus sp. S2 and Bacillus sp. S8, P. citrinum F6, A. nomius F8, and P. rolfsii F10 showed no antagonistic activity against each other. The results obtained served as an indication that the selected isolated strains had no antagonism properties. Therefore, the selected strains were of potential to be used as the cellulolytic degrading microorganism consortia in the following in-vessel composting study discussed in Chapter 4 as it will not inflict any growth inhibition when formed as the microbial consortium.



Figure 3.2 Antagonistic activity evaluations were performed for the 5 selected cellulolytic strains: *Bacillus* sp. S2, and *Bacillus* sp. S8, *P. citrinum* F6, A. *nomius* F8, and *P. rolfsii* F10.

3.3.3 Optimisation of Bacterial and Fungal Strains

A study by Hamzah *et al.* (2013) indicated that physical parameters such as initial medium pH and incubation temperature play an important role to promote microbial biomass production. For this study, the selected degrading microorganisms *Bacillus* sp. S2 and *Bacillus* sp. S8, *Penicillium citrinum* F6, *Aspergillus nomius* F8, and *Penicillium rolfsii* F10 were tested for their optimum pH and incubation temperature by incubating the selected microorganisms in growth mediums tuned to different pH and different temperature conditions.

3.3.3.1 рН

(I) Bacterial Strain

The optimisation of pH for the selected bacterial strain Bacillus sp. S2 and Bacillus sp. S8 was carried out. The selected strains were tested for the optimum growth rate when incubated overnight in nutrient broths with pH ranging from pH5.0 to pH8.0 with an interval of 0.5 as illustrated in Figure 3.3. The growth medium was attuned with the use of 1 N NaOH and 1 N HCl. The absorbance of overnight culture that was incubated in different pH nutrient broth mediums was read using a spectrophotometer at a wavelength of 600 nm. Fresh nutrient was used as a blank for adjusting the background. The optimal incubation pH of growth medium for the selected microorganisms is shown in Figure 3.4. The maximum absorbance reading that could be observed from the assay was produced by Bacillus sp. S8 with the absorbance value of 1.519 when incubated at pH6.0. However, Bacillus sp. S8 also exhibited the lowest absorbance value of 0.577 when grown at pH8.0. It can be said that Bacillus sp. S8 grows the best when grown in the growth medium of pH6.0 when incubated in different pH condition. From the pH optimisation of *Bacillus* sp. S2, the strain also exhibited to prefer the growth condition of pH 6.0 giving an absorbance reading of 1.188. From the graph bar of Bacillus sp. S2, Bacillus sp. S2 showed to favour the growth conditions that have a pH value of lower than pH6.0 as it grows better in acidic conditions compared to the alkaline condition. From the pH optimisation of the selected bacterial strains, it can be concluded that both bacterial strains Bacillus sp. S2 and *Bacillus* sp. S8 favoured the growth medium condition of pH6.0±0.5.



Figure 3.3 The selected bacterial strain (A) *Bacillus* sp. S2 and (B) *Bacillus* sp. S8 exhibited higher growth rates when placed in growth condition of pH6.0.



Figure 3.4 Effect of different pH on the growths for selected bacterial strain *Bacillus* sp. S2 and *Bacillus* sp. S8.

(II) Fungal Strain

The pH optimisation for the three selected fungal strain *P. citrinum* F6, *A. nomius* F8, and *P. rolfsii* F10 was carried out. The selected strains were tested for the optimum growth rate when incubated for 5 days in potato dextrose broth (PDB) ranging from pH5.0 to pH8.0 with an interval of 0.5 as shown in Figure 3.5. From visual observation, it can be seen that more mycelium is produced when the fungal strains were incubated in their optimum pH condition. The PDB growth medium for the fungal strains was attuned with the use of 1 N NaOH and 1 N HCl. The mycelial dry weight obtained from the assay was tabulated in Figure 3.6. The dry weight of fungal grown in PDB obtained from the analysis was used to adjust the background weight.



Figure 3.5 Selected fungal strain (A) *P. citrinum* F6, (B) *A. nomius* F8, and (C) *P. rolfsii* F10 showed to favour the growth condition of pH6.5 where more mycelia were produced.



Figure 3.6 Effect of different pH on the growths of selected fungal strain *P. citrinum* F6, *A. nomius* F8, and *P. rolfsii* F10.

From the pH optimisation of the fungal strains, *A. nomius* F8 produced the heaviest mycelial dry weight of 19.17g after incubation at pH6.0 for 5 days in the incubator shaker followed by *P. rolfsii* F10 which exhibited a mycelial dry weight of 18.67g when grown in pH6.0. From this information, it shows *A. nomius* F8 and *P. rolfsii* F10 favours the pH growth condition of pH6.0. From the graph bar of *A. nomius* F8, *A. nomius* F8 grows better in conditions above pH6.0 than when placed in growth conditions below pH6.0. This displays *A. nomius* F8 prefers to grow in more alkaline

conditions compared to the acidic condition it was placed in. When the pH optimisation was tested on *P. citrinum* F6, the fungal strain exhibited a preference of growth condition with pH6.5 producing a mycelial dry weight of 13.93 g/L. However, it showed the lowest mycelial dry weight of 11.77 g/L when grown in pH8.0. From the optimisation of the fungal strain, it can be concluded that the selected fungal strains preferred to be grown in growth condition with pH6.0 \pm 0.5.

3.3.3.2 Temperature

(I) Bacterial Strain

The optimisation for the selected bacterial strain *Bacillus* sp. S2 and *Bacillus* sp. S8 was carried out. For the initial stage of optimisation of the selected bacterial strains, the bacteria were allowed to grow for overnight in pH6.0 nutrient broth media at temperatures ranging from 35°C to 45°C with an interval of 5°C as it can be seen in Figure 3.7. The optimal incubation temperature obtained from the assay for the selected bacterial strains was shown in Figure 3.8.



Figure 3.7 The selected bacterial strain (A) *Bacillus* sp. S2 and (B) *Bacillus* sp. S8 exhibited higher growth rates when placed in a 45°C condition based on higher absorbance readings.



Figure 3.8 Effect of different temperature on the growth for selected bacterial strain *Bacillus* sp. S2 and *Bacillus* sp. S8.

From the observation of the temperature optimisation for the selected bacteria strains, it can be observed that the bacterial strain grew better when placed in thermophilic condition. Both selected bacterial strains exhibited a higher absorbance reading when incubated at the growth condition of 45°C where *Bacillus* sp. S2 gave an absorbance reading of 1.535 while *Bacillus* sp. S8 had an absorbance reading of 1.891. *Bacillus* sp. S2 showed the lowest absorbance reading of 1.189 when grown at the temperature of 35°C. From the temperature optimisation assay for the selected microorganisms, it can be concluded that *Bacillus* sp. S2 and *Bacillus* sp. S8 favours the growth condition of 45°C as it shows a higher growth rate when incubated in that temperature condition.

(II) Fungal Strain

The optimisation of temperature for the selected fungal strains *P. citrinum* F6, *A. nomius* F8, and *P. rolfsii* F10 was carried out. The selected fungal strains were incubated in pH6.0 potato dextrose broth (PDB) medium placed at different temperatures ranging from 35°C to 45°C with an interval of 5°C for 5 days in an incubator shaker. As can be seen from Figure 3.9, it can be visually observed the

mycelium produced from the fungal strains when placed in different temperatures differ in size and volume. *P. citrinum* F6 produced fine mycelium when grown in 35°C while *A. nomius* F8 and *P. rolfsii* F10 exhibited larger sized mycelium. Mycelium was present in both *A. nomius* F8 and *P. rolfsii* F10 5-day old cultured baffled flask when grown in temperature of 40°C. This indicates the capability of *A. nomius* F8 and P. *rolfsii* F10 to grow in higher temperature growth condition. The dry weight of the mycelium produced by the fungal strains from 5 days of incubation in different temperatures was illustrated in Figure 3.10.



Figure 3.9 Selected fungal strain (A) *P. citrinum* F6, (B) *A. nomius* F8, and (C) *P. rolfsii* F10 showed to favour an incubation temperature of 35°C where more mycelia were produced.



Figure 3.10 Effect of different temperature on the growth of selected fungal strains *P. citrinum* F6, *A. nomius* F8, and *P. rolfsii* F10.

From the temperature optimisation of the selected fungal strains, *A. nomius* F8 exhibited to have the heaviest mycelial dry weight of 20.41 g per 100 mL when incubated in the growth condition of 35°C followed by *P. rolfsii* F10 with mycelial dry weight of 19.09 g per 100 mL. *A. nomius* F8, and P. rolfsii F10 produced mycelium when grown at the temperature of 40°C indicating the ability of the two out of the three selected fungal strains to grow at a higher temperature. However, *P. citrinum* F6 was not able to grow when it was placed in the growth condition of 40°C and 45°C. *P. citrinum* F6 produced a mycelial dry weight of 14.12 g when incubated at 35°C. From the optimisation of the selected fungal strain, *P. citrinum* F6, Aspergillus *nomius* F8, and *P. rolfsii* F10, it can be concluded that the three selected strains favoured the growth conditions of 35°C.

3.3.3.3 Optimal Growth Conditions of Selected Microorganisms

The optimal growth condition of the 5 selected microorganisms was tested based on the condition of pH and temperature. From the pH optimisation assay of the selected strains, the data obtained from it exhibited that both selected bacterial and fungal strains favoured the growth conditions of pH6.0 \pm 0.5 when incubated in tuned basic broth mediums that consist of different pH conditions ranging from pH5.0 to pH8.0. During the optimisation of temperature for the selected strains, the two selected bacterial strains showed to favour the growth of condition of thermophilic temperature of 45°C while 2 out of the 3 selected fungal strains preferred to be incubated in the temperature condition of 35°C. After observing data from all optimisation, the growth rate of the microorganisms can be seen to slowly increase after each element was adjusted to the optimum condition. The summary of the increasing growth rate of the selected microorganisms can be seen from Table 3.1 for the selected bacterial strain and Table 3.2 for the selected fungal strain.

Table 3.1	Summary	of increasing	growth	rate o	f selected	bacterial	strains	after
optimisati	on.							

	Bacillus sp. S2 (Bacteria),	Bacillus sp. S8 (Bacteria),	
	Absorbance reading (Abs.)	Absorbance reading (Abs.)	
NB pH7.3, 37°C	0.901 ± 0.006	0.589 ± 0.011	
NB pH6.0, 37°C	1.188 ± 0.010	1.118 ± 0.002	
NB pH6.0,45°C	1.535 ± 0.004	1.891 ± 0.005	

	P. citrinum F6,	A. nomius F8,	P. rolfsii F10,	
	Mycelial Dry	Mycelial Dry	Mycelial Dry	
	Weight (g)	Weight (g)	Weight (g)	
PDB pH5.0, 37°C	12.15 ± 0.01	15.73 ± 0.05	15.65 ± 0.04	
Р DB р Н6.0, 37°С	13.01 ± 0.09	19.17 ± 0.01	18.67 ± 0.05	
PDB pH6.0, 35°C	14.12 ± 0.04	20.41 ± 0.01	19.09 ± 0.01	

Table 3.2 Summary of increasing growth rate of selected fungal strains after optimisation.

3.3.4 Microbial Growth Rate of Selected Microorganisms

The growth of the selected cellulolytic bacterial and fungal strain was analysed. A different method of the test was used due to the growth difference of bacteria and fungus. The aim of this experiment was to observe the growth rate of the selected strain and also the amount of cell that will be inoculated into the in-vessel composting that will be discussed in Chapter 4.

3.3.4.1 Bacterial Strain

Evaluations of the growth kinetics of the selected bacterial strains *Bacillus* sp. S2 and *Bacillus* sp. S8 were conducted based on indirect growth measurement technique and the direct growth measurement technique. For the indirect growth measurement technique, the bacterial strains were grown in their own individual flask containing 90 mL of nutrient broth and 10 mL overnight broth culture. The flask was incubated at 35°C in the incubator shaker and observed. A spectrophotometer was used to measure the turbidity of the culture at 600 nm wavelength. During the direct growth measurement technique, the viable count or plate count of the bacterial strains was carried out as can be seen in Figure 3.11. The growth kinetics characteristics of the selected bacterial strains were recorded in Table 3.3.



Figure 3.11 Viable count or plate count was used for direct growth measurement technique of the selected bacterial strains (A) *Bacillus* sp. S2 and (B) *Bacillus* sp. S8.

Table 3.3 Growth kinetics characteristics of the selected bacterial strainsBacillus sp. S2 and Bacillus sp. S8 grown in NB at 35 °C, 120 rpm.

Strain	Specific growth rate, k [hr-1]	Doubling time, Dt [hr]	Absorbance value of bacteria at 24 hr [OD ₆₀₀]	Colony forming units at 24 hr [CFU/mL]
Bacillus sp. S2	0.2102	3.30	1.279	$4.04x10^{7}$
<i>Bacillus</i> sp. S8	0.3311	2.10	1.580	$4.52x10^{7}$

The growth kinetic characteristics for the selected bacterial strain *Bacillus* sp. S2 and *Bacillus* sp. S8 was evaluated based on spectrophotometer assay and total plate count techniques. The formula shown in Equation 3.2 was used to calculate the specific growth rate of the bacterial strains based on the absorbance reading obtained from the spectrophotometer assay.

Equation 3.2
$$\mu = \frac{ln \frac{OD_{600}t_2}{OD_{600}t_1}}{t_2 - t_1}$$

The specific growth rate is a quantitative measurement used to measure the increase of cell mass over a specific time. *Bacillus* sp. S8 showed to have a higher a specific growth rate of 0.1209 hr^{-1} than *Bacillus* sp. S2. The doubling time for each bacterial strain was also studied by calculating the doubling time based on the value obtained from the specific growth rate. The doubling time is the period of time required for the bacterial strain to double its quantity. The doubling time of the bacterial strain is proportional to the specific growth rate. In other words, the higher the specific growth rate, the shorter

the doubling time is required by the bacterial strain to double its quantity size. From Table 3.3, *Bacillus* sp. S8 showed to have a shorter doubling time of 1.20 hr than *Bacillus* sp. S2. This shows that *Bacillus* sp. S8 takes 2.10 hr to double its population while *Bacillus* sp. S2 requires 3.30 hr to double up its quantity, which is 1.20 hr slower than *Bacillus* sp. S2. The absorbance reading of the two selected bacterial strains along with the colony forming unit after 24 hr of incubation were recorded in Table 3.3. *Bacillus* S8 showed to have a higher absorbance and colony forming unit after 24 hr of incubation than *Bacillus* sp. S2. From this study, it can be concluded that *Bacillus* sp. S8 has a higher growth rate than *Bacillus* S2.

3.3.4.2 Fungal Strain

Evaluations of the growth kinetics of the three selected fungal strains *Penicillium citrinum* F6, *Aspergillus nomius* F8, and *Penicillium rolfsii* F10 were conducted based on indirect growth measurement technique and the direct growth measurement technique. For the indirect growth measurement technique, the fungal strains were grown in their own individual flask containing 90 mL of potato dextrose broth and 10 mL 5-day old broth culture. The flask was incubated at 35°C in the incubated in an incubator shaker for 5 days as can be seen in Figure 3.12 and the mycelium biomass was obtained by filtering it using filter paper. The dry weight of the mycelium was obtained after 24 hr of oven dry. During the direct growth measurement technique, a sterile 5mm straw was used to transfer 5 days old agar plate culture onto fresh potato dextrose agar (PDA). The inoculated PDA plates were incubated at 35°C for 5 days. On the 5th day of incubation, the diameter of the fungal culture was measured and recorded. The growth kinetics characteristics of the 3 selected fungal strains were recorded in Table 3.4.

The growth kinetics characteristics for the selected fungal strains *P. citrinum* F6, A. *nomius* F8, and *P. rolfsii* F10 were evaluated based on the dry weight of the mycelium obtained from 5 days old culture and the growth diameter of the 5-day old fungal culture on PDA. From Table 3.4, *A. nomius* F8 showed to have the highest value of mycelial dry weight with the value of 20.58 g from the assay while *P. citrinum* F6 has the lowest mycelial dry weight value of 14.79 g. This shows that *A. nomius* F8 grows faster than the other selected fungal strains, therefore, producing more mycelium in a shorter period of time. *A. nomius* F8 also exhibited higher growth rate on PDA plate, having a diameter of 8.4cm which is higher than *P. citrinum* F6 and *P. rolfsii* F10 by 2.6 cm and 1.2 cm

respectively. From this study, it can be concluded that *A. nommius* F8 has the highest growth rate followed by *P. rolfsii* F10 and *P. citrinum* F6.



Figure 3.12 The dry weight of the mycelium obtained from the 5-day old culture broth of (A) *A. nomius*, (B) *P. citrinum*, and (C) *P. rolfsii* was used for the indirect growth measurement technique to measure the fungal growth.

Table 3.4 Growth kinetics characteristics of the selected fungal strains, P. citrinumF6, A. nomius F8, and P. rolfsii F10 grown in PDB at 35 °C, 120 rpm.

Studin	Mycelial Dry Weight	Growth Diameter at Day 5 (cm)	
Strain	after 5 days (g)		
Penicillium citrinum F6	14.79 ± 0.63	5.8 ± 0.12	
Aspergillus nomius F8	20.58 ± 0.66	8.4 ± 0.24	
Penicillium rolfsii F10	19.37 ± 0.42	7.2 ± 0.19	

3.3.5 Selected Final Strains for Development of Indigenous Microbial Consortium

The selected cellulolytic strains intended for the formulation of indigenous microbial consortium for in-vessel composting were pre-tested for their abilities to secrete cellulases and hemicellulases based on the glucose level detected from the degradation of carboxymethylcellulose (CMC) and xylan respectively. The selected strains were also tested for antagonism effect among each other to prevent the possibility of mutual growth inhibition during in-vessel composting. These selected strains were chosen based on their abilities to degrade CMC and xylan as discussed earlier (Figure 3.13). The selected strains had exhibited high glucose level among the 24 cellulolytic strains isolated from soil, and had a comparable result to the positive control *Aspergillus niger* (NBRC 33023). The selected cellulolytic strains also showed no antagonistic activity among each other. Hence, these five strains were confirmed to be used as indigenous microbial consortium for the composting of oil palm empty fruit bunch (OPEFB).





Figure 3.13 The selected 5 strains (A) *Bacillus* sp. S2, (B) *Bacillus* sp. S8, (C) *P. citrinum* F6, (D) *A. nomius* F8, and (E) *P. rolfsii* F10 were chosen for the formulation of the degrading microorganisms consortia for the in-vessel composting discussed in Chapter 4.

3.4 Summary

The study of Chapter 3 was focused on the cellulase and hemicellulase enzyme productions by the selected cellulolytic microorganisms, mutual antagonism within these selected microorganisms, and also the growth rate of the respective microorganisms. The cellulase and hemicellulase activities of the isolated microorganisms were successfully confirmed using the 2-cyanoacetamide method. After performing the cellulase and hemicellulase assays, Bacillus sp. S2, Bacillus sp. S8, Penicillium citrinum F6, Aspergillus nomius F8, and Penicillium rolfsii F10 were chosen as they exhibited among the highest glucose levels. These selected cellulolytic microorganisms were used to form an indigenous microbial consortium for the composting of oil palm empty fruit bunch(OPEFB) during an in-vessel composting in Chapter 4. Antagonism evaluations were performed among these strains where all the respective strains were paired up and cultured onto the same nutrient agar plate and observed for any mutual growth inhibition. From the evaluation, no antagonistic activities were present in any of the selected strains when paired and grown together. Investigations of the optimum growth condition for the selected microorganisms were also performed. The selected microorganisms were grown in different pH ranging from pH5.0 to pH8.0 with an interval of 0.5. They were also grown in a different temperature of 35°C, 40°C, and 45°C. From the analysis, both selected bacterial and fungal strains showed preference to be grown in a pH of pH6.0. The bacterial strains showed to favour growth in thermophilic temperature of 45°C while the fungal strains preferred to grow in 35°C. However, two out of three fungal strains, namely A. nomius F8 and P. rolfsii F10, were able to grow in 40°C. Upon further investigation of the bacterial strains, Bacillus sp. S8 showed to have the higher growth rate when compared to Bacillus sp. S2. For the fungal strains, A. nomius F8 showed to have a higher growth rate, followed by P. rolfsii F10, and P. citrinum F6. The efficiency of the five selected strains was tested in Chapter 4 in the in-vessel composting experiment.

Chapter

4

PRODUCTION OF COMPOST VIA IN-VESSEL COMPOSTING OF OIL PALM EMPTY FRUIT BUNCH USING CELLULOLYTIC MICROORGANISMS AND ITS ENHANCEMENT IN THE GROWTH OF OIL PALM SEEDLINGS

4.1 Introduction

The oil palm industry has become one of the most important economic contributors in Malaysia (Abidemi *et al.* 2006). As a result, the oil palm industry produces a large amount of solid waste which consists of lignocellulosic materials such as fronds, trunks, and empty fruit bunches (Harun *et al.* 2013). Oil palm empty fruit bunches (OPEFB) is one of the major dry biomass wastes produced by the oil palm industry (Abdullah & Sulaiman 2013). OPEFB are usually burned to generate power for mills, however, this act has caused environmental pollution. Therefore, sustainable waste management for the reduction of this environmental hazardous waste must be explored.

Composting of agroindustrial wastes such as OPEFB had been suggested to be the most practical method for managing the disposal large volume of these wastes (Zainudin *et al.* 2013). Composting is biodegradation of organic waste turns into stable and humified organic material by a mixed microbial population that occurs under controlled conditions distinguishes itself from natural rotting (Sarkar *et al.* 2011).

Composting is the most suitable way to manage waste economically and environmentally as it reduces bulk volume of oil palm waste into useful compost which improves and sustain soil fertility (Mukhlis *et al.* 2013). According to Baharuddin, *et al.*, OPEFB is a lignocellulosic material that usually contains 53% cellulose, 30% hemicellulose and 17% lignin which makes it require a longer composting period (2009). Cellulolytic microorganisms are the key to depolymerise the lignocellulosic material in OPEFB at a faster rate (Mukhlis *et al.* 2013). Thus, the addition of effective cellulolytic microorganisms to the compost would accelerate the composting process. In this composting experiment, the indigenous microbial consortium containing a variety of cellulolytic microorganisms as discussed in Chapter 3 were inoculated into the OPEFB composting material for in-vessel composting. The objective of the experiment was to test the efficiency of the indigenous microbial consortium in the composting of OPEFB. The compost produced from the experiment was further analysed for its effectiveness in the enhancement of the growth of oil palm seedlings via small scale pot trial.

Oil palm seedling has been cultivated using topsoil as a growing medium during the nursery stage. However, using top soil as a growth medium for long term is not practical as growing oil palms would require more nutrients as they grow (Rosenani *et al.* 2016). For the optimum production yields from crops, sufficient and balanced nutrients must be

available for the plants' absorption. Therefore, fertilisers are required to supplement the nutrients already present in soil (Chen 2006). Chemical fertilisers are usually used as an essential nutrition for the crops to get a higher yield. However, in recent years, the yields have become almost static where no crop yields have increased even with the application of the optimum level of chemical fertilisers (Ahmad *et al.* 2007). Furthermore, high energy cost inputs are required for the production of chemical fertiliser. From this scenario, it is highly likely that the use of compost accompanied with chemical fertiliser could be an effective alternative approach to improving the crop yields. According to Ahmad *et al.* (2008), fortification of compost with chemical fertiliser enhances the effectiveness of both compost and chemical fertiliser as it reduced the amount of fertiliser required and also improves the quality of the compost. Apart from that, compost can also help to improve the organic matter status and physiochemical properties of the soil when applied together with chemical fertiliser which gives an addition of nutrients to the plants (Ahmad *et al.* 2007).

Several papers had reported the application of compost to crop productions that exhibited beneficial effects (Gil *et al.* 2008). Compost had been established to have potential ability to enhance water retention, reduce soil compaction and acidity, and also enhance the biological activities in the soil (Sarkar *et al.* 2011). However, compost is comparatively low in nutrient content, so larger volumes are required to provide sufficient nutrients to the growth of plants (Chen 2006). Traditionally, farmers would have to add several tonnes per hectare of compost to their plants due to the poor nutrient of compost (Ahmad *et al.* 2008). In recent practices, it is rather common to integrate compost with chemical fertiliser. An evaluation of soil quality by Dutta *et al.* (2003) showed to a positive effect on the soil health and hence plant health when organic fertiliser and chemical fertiliser were integrated together.

For this chapter in the research study, the indigenous microbial consortium created that contained the selected cellulolytic microorganisms isolated from the collected soil samples were tested for its efficiency in degrading OPEFB during the in-vessel composting study. This experiment was carried out to successfully create an indigenous microbial consortium of degrading microorganisms that can help degrade OPEFB waste in a shorter period of time to produce compost. The compost was further tested for its effectiveness in the enhancement of the growth of oil palm seedlings via small scale pot

trial. The aim was to develop an effective soil supplement by combining compost and chemical fertiliser to increase the growth of oil palms seedlings.

4.2 Materials and Methods

4.2.1 In-Vessel Composting

The composting experiment was conducted in a small-scale in-vessel system placed on black plastic covering in an empty land located in Kuching, Sarawak. A total of nine 200-L vessels were used for the composting of the oil palm empty fruit bunch (OPEFB) and partially treated palm oil mill effluent (POME). The composting vessels were bought and fabricated by a local company named Lien Ann Engineering Sdn Bhd, Kuching, Sarawak. Holes were drilled at the sides of the composting vessels holes to ensure proper aeration and doors latch was made to the vessels to obtain samples and data throughout the composting study. Stands with wheels were also custom made by Lien Ann Engineering Sdn Bhd, composting vessels were placed on top of the roller stand to allow proper rotation of the vessels.



Figure 4.1 Composting vessels were fabricated with door latches (A) and placed on stand stands with rollers (B) to allow easy rotation.

4.2.1.1 Preparation of The Commercial Microbial Agents

The positive control used in this study was a commercially available composting indigenous microbial consortium constituted of *Bacillus* spp. which were *B. cereus, B. subtilis,* and *B. thuringiensis.* The degrading microbe's consortium was obtained in a concentrated dry powder form that was stored at room temperature. For the revival of the

commercial microbial agents, 1.5 g of the concentrated powder was inoculated into 6 replicates of 1500 mL sterilised nutrient broth (Himedia, India) in baffled shake flask (Pyrex, USA), separately. After incubating the indigenous microbial consortium for overnight in the incubator shaker (Sartorious Stedim Biotech, Germany) with a condition of 37°C with shaking at shaken at 180 rpm, they were allocated into three 3-L beakers, each beaker (Duran, UK) contained 500 mL of degrading microbe from each baffled shake flask. They were mixed thoroughly to produce 3 sets of 3-L positive control degrading microbe's consortia for each positive control composting vessel in the composting study.

4.2.1.2 Preparation of Indigenous Microbial Consortium

The following five cellulolytic microorganisms were selected to be used as the indigenous microbial consortium for the composting study:

- (i) *Bacillus* sp. S2 (Bacterium)
- (ii) *Bacillus* sp. S8 (Bacterium)
- (iii) *Penicillium citrinum* F6 (Fungus)
- (iv) Aspergillus nomius F8 (Fungus)
- (v) *Penicillium rolfsii* F10 (Fungus)

The five selected cellulolytic microorganisms were first grown individually in 180 mL fresh basic broth. The cellulolytic bacterial strains were grown in nutrient broth (Himedia, India) while fungal strains were incubated in potato dextrose broth (Himedia, India) with pH6.0 in 250 mL baffled shake flask (Pyrex, USA) incubated at 35°C shaken with agitation of 120 rpm. The bacterial cellulolytic microorganisms were grown overnight while the fungal cellulolytic microorganisms were grown for 5 days before inoculating them into 1620 mL sterilized broth tuned to pH6.0 in baffled shake flask (Pyrex, USA), separately. The freshly grown cellulolytic microorganisms in 180 mL broth medium were poured directly into 1620 mL individual growth medium to ensure a standardized amount of inoculum is inoculated into the broth to ensure consistency of inoculation. The bacterial cellulolytic microorganisms were grown for 5 days at 35°C with shaking at 180 rpm in the incubator shaker (Sartorious Stedim Biotech, Germany). After incubating the cellulolytic microorganism was measured and poured into three 3-L beakers (Duran, UK) to create a consortium of

cellulolytic microorganisms. They were mixed thoroughly to produce 3 sets of 3-L manipulated variable cellulolytic indigenous microbial consortium for each manipulated composting vessel in the composting study.

4.2.1.3 Transportation of Inoculum to In-Vessel Composting Site

The inoculum for the in-vessel composting was packed into labelled plastic bottles with the caps screwed tightly and sealed with paraffin film. The bottles were wipe down with 70% ethanol and placed into a Styrofoam box with sponges surrounding the bottles to prevent the bottles with the inoculums from toppling over and leaking. The Styrofoam box was taped with a hazard warning sign accompanied with the material transfer form and biosafety level of containment. The name and telephone number of the person responsible for handling the inoculums were also written onto the material transfer form.

4.2.1.4 Raw Materials

The composting material, oil palm empty fruit bunch (OPEFB) and partially treated palm oil mill effluent (POME) were obtained from the sponsoring company, Daitoku Sdn Bhd. The OPEFB were obtained after oil palm fruits were removed from the fresh fruit bunch (FFB) during the milling process (threshing). After the threshing process, the OPEFB were shredded into approximately 5 cm length loose fibrous material by using a shredder machine. The palm oil mill effluent used in this study was collected from an anaerobic pond which was used to moisten the OPEFB.

The raw OPEFB and partially treated POME were analysed for its physio-chemical properties before the experiment. Freshly shredded EFB was sent to i-Testchem Laboratory Services, Kota Samarahan, Kuching to analysed for moisture, pH, organic carbon, nitrogen, phosphate, potassium, magnesium and its C: N ratio while freshly collected partially treated POME was analysed for its pH, nitrogen, chemical oxygen demand (COD), biological oxygen demand (BOD), oil and grease, total solids, phosphate, and potassium.

4.2.1.5 Preparation of The Composting Vessels and Experimental Design

For the composting raw materials, 30 kilograms of oil palm empty fruit bunch (OPEFB) was weighed and placed into each individual composting vessel. Approximately 10 L of partially treated palm oil mill effluent (POME) was sprayed evenly onto the EFB into each composting vessel. The substrate ratio of OPEFB to partially treated POME was made to 3:1 as the method was adapted from a related study by Mohammad *et al.* (2014).

After ensuring that the OPEFB was evenly moistened by the partially treated POME, 3 L of inoculum was inoculated into each vessel. Three test groups were carried for the composting study as shown in Table 4.1, each experimental group was carried out in triplicates where the composting vessels were arranged in a completely randomised design.

Experimental group	Description
Control	OPEFB, partially treated POME, sterilised broth.
Commorgial migraphial agants	OPEFB, partially treated POME, commercial microbial
Commercial interoblat agents	strains.
Indigenous microbial	OPEFB, partially treated POME, cellulolytic indigenous
consortium	microbial consortium that was discussed in Chapter 3.

 Table 4.1 Experimental groups for the in-vessel composting study.

4.2.1.6 Composting Analysis Throughout the Study

A moisture level of 65-75% was maintained throughout the composting process by spraying partially treated palm oil mill effluent (POME) on the oil palm empty fruit bunch (OPEFB) when necessary. To ensure proper aeration was provided, the compost vessels were turned over every 2 days to allow air to enter the compost and give proper mixing. The C: N ratio and temperature of compost were monitored throughout the 60-day composting process. A hygrometer was used to check the temperature and humidity of the surrounding area.

4.2.1.7 Sampling

A 500g compost sample was collected at different locations in the composting vessel: bottom, core, and surface. The samples collected in triplicates were air dried and kept in sterile clear polyethene resealable bag at 4°C.

4.2.1.8 Sample Analysis

Compost samples were sent to i-Testchem Laboratory Services, Kota Samarahan, Kuching to analysed for the C: N ratio every 15 days throughout the 60 days in-vessel composting study. Final compost sample collected at the end of the composting study was analysed for moisture, pH, organic carbon, nitrogen, phosphate, potassium, and magnesium.

4.2.2 Pot Trial

4.2.2.1 Preparation of Additive

Chemical fertiliser - Chemical fertiliser chosen for this pot trial study was a standard palm oil fertiliser formula used in Malaysia with the nutrients ratio of 14:7:9 (Rosenani *et al.* 2016).

Compost - Compost was obtained from the compost produced during the in-vessel composting study discussed in Chapter 4.

4.2.2.2 Preparation of Oil Palms and Experimental Design

Oil palm seedlings that were 5-month old of age were selected for the pot trial study. The 5-month old oil palm seedlings were purchased from Sarjoh Plantation Nursery Sdn. Bhd, Kuching, Sarawak and transplanted into a soil mixture that consists of top soil and sand with the ratio of 3:1 (Tchapda & Oscar 2016). The soil used to plant the oil palms were bought from Pelangi Landscaping and analysed for its physio-chemical properties. The oil palms were watered with approximately 1.5 L of tap water for each plant daily for irrigation and monitored for a month before beginning the treatment.

After 30 days, 40 oil palms with the best condition were selected and treated with different treatments. The selected oil palms were uprooted and replanted into new bigger polybags. Pot trial experiment was conducted in a black netting roofing shed to assess the effect of compost and chemical fertiliser on the growth of 6 months old oil palms. The selected oil palms were divided into four different treatment groups with 10 oil palms seedlings per treatment group for the pot trial study as shown in Table 4.2.

The oil palms with different treatments were arranged in the net house in a completely randomised design with 10 replicates for each treatment with ambient light and temperature provided. For each pot, approximately 1.5 L of tap water was watered to each pot daily in the morning for irrigation.

Experimental Group	Description	Amount of additive		
Control	The oil palms were planted without the presence of any added nutrients.	No additives were added.		
Chemical fertiliser	The oil palms were fertilised with chemical fertiliser.	20 g of NPK 20-10-10 chemical fertiliser.		
Compost	The oil palms were fertilised with compost only.	100 g of enriched compost.		
Compost-chemical mixed fertiliser	The oil palms were fertilised with a mix of compost and chemical fertiliser.	10 g of enriched compost and 5 g of NPK 14:7:9.		

Table 4.2 Treatment groups for pot trial study.

4.2.2.3 Application of Additive

Chemical fertiliser group - The chemical fertiliser was measured out beforehand and poured onto the top layer of the soil after transplanting.

Compost group- The compost was incorporated in the soil for each pot by mixing it with the top 10 cm soil at the time of transplanting.

Compost-chemical mixed fertiliser group- The compost was supplemented with chemical fertiliser by integrating the measured compost and chemical fertiliser together. It was then incorporated into the soil by mixing it with the top 10 cm soil at the time of transplanting.

4.2.2.4 Oil Palm Data

(I) Initial Physical Examination

The oil palms initial physical examination was carried out and recorded before the start of the pot trial. The measurements and observations were:

- (i) The number of opened leaves was counted.
- (ii) The height of the trees where it was measured by beginning from the base of the stem which was right above the soil surface to the first leaf internode that was encountered along the stem of the oil palm (Figure 4.2)
- (iii) The measurement of the stem girth was performed by using a cotton string to measure the girth diameter of the oil palm bulb (Figure 4.2).

4.2.2.5 Observations of Oil Palms Seedlings at End of Pot Trial Experiment

Oil palm plants were measured for its physical properties (number of leaves, height and girth measurement of oil palm) and harvested at the 6th month of experiment for analysis. Freshly harvested oil palm leafs were sent to Sarawak Plantation Services Sdn. Bhd. in Kuching, Sarawak to performed foliar analysis in accordance to MS 677: Parts III, IV, V, VI, VII: 1980 to analysed for the percentage of nitrogen, phosphate, potassium, magnesium, calcium, and boron based on the dry leaf matter (*Malaysian Standard* nd.).



Figure 4.2 Technique used to measure the height (A) and stem girth (B&C) of the oil palm.

4.3 Results and Discussion

4.3.1 In-Vessel Composting

4.3.1.1 Characteristics of Raw Materials and Final Compost

Freshly shredded oil palm empty fruit bunch (OPEFB) contains a relatively large amount of cellulose and lignin. Therefore, requiring a longer period of time to decompose compared to other composting material (Baharuddin *et al.* 2009). Throughout the composting process, different colour and texture of the OPEFB inoculated with commercial microbial agents and indigenous microbial consortium was observed as it slowly degraded (Figure 4.3). The OPEFB was light brown and in the form of long rough fibrous material at the beginning of the experiment. The OPEFB slowly turned darker in colour and shorter in length as it degraded throughout the experiment. Matured compost at the end of the experiment exhibited a blackish appearance accompanied with a texture similar to soil.



Figure 4.3 Physical changes of oil palm empty fruit bunch at Day 0(A), Day 20(B), and Day 60(C).

Oil palm empty fruit bunches (OPEFB) has a low moisture content but high carbon to nitrogen ratio (C: N ratio) while partially treated palm oil mill effluent (POME) has the opposite characteristic as can be observed from Table 4.3. Thus, by mixing these two materials together, it can provide adequate nutrients and moisture content for microorganisms to grow and degrade the composting material. According to Baharuddin *et al.*, OPEFB and partially treated POME complement each other during the co-composting as the high carbon and low nitrogen content available in the OPEFB balances the partially treated POME which has low carbon but high nitrogen content (2009). The value of the parameters recorded in Table 4.3 showed some variations in the chemical

oxygen demand (COD), biological oxygen demand (BOD), oil and grease and total solids. The variations could be due to the unpredictable weather condition and inconsistent discharge from mill operations (Yacob *et al.* 2005).

Throughout the study, partially treated POME was added to the composting material (OPEFB) to maintain a moisture level of 65-75%. According to Liang and Das (2003), the ideal range for moisture content in a compost is 60-75% as it provides maximum microbial activities. Therefore, partially treated POME was added to each OPEFB in the composting vessels every subsequent 3 days to replenish the water loss due to the evaporation to the environment to maintain the microbial activities (Baharuddin *et al.* 2010). Partially treated POME was added to composting vessel until day 50 to avoid the final compost to be too wet. The final compost sample collected on day 60 from all the experimental groups had a moisture content of 60-65% (Table 4.3).

Each composting vessel was turned over every two days to maintain proper distribution of the moisture and build-up of heat during the composting period. Turning of the composting vessel was also carried out to provide a proper supply of aeration for the composting materials. The composting materials in the vessel were made sure to have proper aeration to prevent the slowdown of decomposition due to low oxygen level. Low oxygen level increases the opportunity of the composting material to absorb ammonia which will lead to immobilisation (Baharuddin *et al.* 2010).

	Raw ma	aterials	Final Compost			
Danamatans	Shuaddad	Partially		Commercia	Indigenous	
r ar anneter s	Shreudeu	treated	Control	l microbial	Microbial	
	OPEFB	POME		agents	consortium	
Moisture (%)	18.97 ± 1.33	95.30 ± 2.10	60.1 ± 11.40	62.43 ± 5.29	64.67 ± 1.29	
рН	5.57 ± 0.31	7.70 ± 0.40	7.47 ± 0.29	8.77 ± 0.47	8.50 ± 0.10	
Carbon (%)	39.87 ± 0.21	19.10 ± 1.90	32.93 ± 3.05	30.27 ± 2.99	30.70 ± 1.87	
Nitrogen (%)	1.33 ± 0.12	2.80 ± 0.60	2.01 ± 0.25	2.25 ± 0.28	3.04 ± 0.37	
C:N ratio	30.05 ± 2.58	7.90 ± 3.60	17.00 ± 1.15	13.33 ± 1.53	10.00 ± 1.73	
COD		1338.00				
$(mg L^{-1})$	-	± 226.64	-	-	-	
BOD		263.00	-	-	-	
$(mg L^{-1})$	-	± 15.92				
Oil and		10.00				
grease	-	18.00	-	-	-	
$(mg L^{-1})$		± 2.20				
Total solid		2744.00		-		
$(mg L^{-1})$	-	± 159.63	-		-	
Composition of nutrients and metal elements (%)						
Phosphate	0.15 ± 0.00	4.23 ± 0.60	0.25 ± 0.04	0.28 ± 0.06	0.46 ± 0.07	
Potassium	0.64 ± 0.01	1.80 ± 0.30	1.17 ± 0.25	1.31 ± 0.06	1.14 ± 0.13	
Magnesium	0.20 ± 0.01	0.90 ± 0.10	0.40 ± 0.05	0.55 ± 0.05	0.60 ± 0.05	

Table 4.3 Properties of shredded OPEFB, partially treated POME and final compostfor the three experimental test groups at Day 60.

4.3.1.2 Physiochemical and Biochemical Changes in Composting

(I) Temperature Variation

The most important indicator that can be used to check the efficiency of the composting process is temperature (Li *et al.* 2008). The ambient temperature around the composting site was measured to be around 24-30°C throughout the study. Figure 4.4 shows the temperature profile of the compost vessels thought the 60-day composting study. The average temperature of the oil palm empty fruit bunch (OPEFB) and partially treated palm oil mill effluent (POME) in the composting vessel at the start of the compost study after inoculation was approximately 21°C. On

the second day of fermentation, the temperature of the composting vessels that contained the commercial microbial agents and indigenous microbial consortium increased sharply to about 65°C. The temperature of the compost spiked up due to the explosive growth of the mixed population and active microbial decomposition of the raw material generating heat. During this phase, the dominant microorganisms attack the soluble, readily biodegradable compounds which are relatively small size with high content of available nutrients (Fourti *et al.* 2008). The composting vessel that was inoculated with the commercial microbial agents and indigenous microbial consortium maintained in a thermophilic phase from day 3 to day 28 with a temperature around 40-63°C. According to Wong *et al.* (2001), most studies on composting had reported that the optimum temperature range for effective decomposition of raw material was 50-70°C where 60°C was regarded as the most satisfactory level.





Figure 4.4 Compost temperature profile for 60 days of in-vessel composting.

Several reports have claimed the temperature of the composting process must reach a temperature of 55°C and above to kill pathogens and sanitize the compost (Gea *et al.* 2005). According to Hoitink and Fahy (1986), diseases caused by pathogens can be controlled through composting as pathogens in the waste would be killed. The composting process of the composting vessel inoculated with commercial microbial agents and indigenous microbial consortium met the sanitary requirement without requiring extra heat application as it remained in the thermophilic phase for approximately 25 days of the experiment. Baharuddin et al. (2010) stated that a temperature of 45-55°C must be maintained in a composting process to ensure maximum biodegradation of the raw materials. During the thermophilic stage, cellulose and hemicellulose of the OPEFB would be broken down by the thermophilic bacteria and heat-tolerant fungi. The temperature of the composting vessel inoculated with commercial microbial agents and indigenous microbial consortium decrease gradually after thermophilic phase and entered a curing phase. During the curing phase, lignin would be slowly degraded by the mesophilic microorganisms in the compost. Throughout the composting study, the composting vessel that was used as the control entered the lower range of thermophilic phase from day 3 to day 9 with the highest temperature of 44°C recorded during day 9 before falling back into the mesophilic phase where a temperature around 22-38°C was maintained until the end of the experiment. This was mainly because no microorganisms were inoculated into it. Therefore, not much microbial activity was observed in the control vessel. The temperature of all the experimental group remained the same after day 40 despite turning of the vessel for another 20 days.

(II) Carbon to Nitrogen Ratio Observation

The nitrogen content of the entire experimental group increased gradually throughout the composting process while the carbon content of the compost decreased throughout the treatment. The nitrogen content of the indigenous microbial consortium group rose the highest among the entire experimental group where it was originally 1.0% and increased to 3.0%. It was followed by the commercial microbial agents experimental group where the nitrogen content increased from 1.0% to 2.3%. The control group had the lowest increase in nitrogen content which was 1.0% to 2.0%. The carbon content of the composting material of the composting material was decreased from 40.0% to 33.0% (control), 30.0% (commercial microbial agents), and 31.0% (indigenous microbial consortium) whereby indigenous microbial consortium had the most decrease. The phenomenon of the increase of nitrogen

content and a decrease of carbon could be attributed from the microbial activity on the cellulose substrate and nitrogen, which increased the microbial protein and humic substance (Thambirajah *et al.* 1995). A lower carbon content that is accompanied with a high nitrogen value gives a low C: N ratio for the compost.

Wong *et al.* (2001) mentioned that the carbon to nitrogen ratio (C: N ratio) can be used as one of the indicators to determine the maturation of the compost. C: N ratio of a compost can also be used in determining the stability apart from the maturity of the compost (Baharuddin *et al.* 2010). The C: N ratio was calculated from the carbon and nitrogen content of the composting material from the composting study was used to measure the rate of decomposition of the raw materials. The C: N ratio analysed from the samples collected from the compost vessels on specific time point was illustrated in Figure 4.5. The experimental group for commercial microbial agents and indigenous microbial consortium showed to decreased in C: N ratio and reach a stable condition of 13:1 and 12:1, respectively. An article by Mohammad *et al.* (2014) mentions that compost with a C: N ratio of not higher than 20 could be considered to have reached a satisfactory maturation level. However, Jiménez and García (1992) reported that compost that exhibited a stable ratio of 15:1 and below is often preferable.

The duration of composting was focused on as it is a critical criterion in order to make the composting method fast and feasible for the management of oil palm waste. The C: N ratio profile of the composting experiment using indigenous microbial consortium exhibited to satisfactory results in which better maturity level in a shorter period of time were achieved, which was in line with to other related studies. Studies by other researchers reported by Zainudin *et al.* (2013) to obtain a C: N ratio of 12.4 after 40 days of composting OPEFB with cellulolytic and hemicellulolytic rich untreated rich POME while Baharuddin *et al.* (2009) managed to get a C: N ratio of 12.7:1 after 60-day of co-composting OPEFB with partially treated POME. The indigenous microbial consortium in this experiment exhibited to have a conspicuous result as it exhibited a C: N ratio of 12:1 after 30-day composting when compared to the other researchers study showing its ability to degrade OPEFB effectively.



Figure 4.5 Compost C: N ratio profile throughout 60 days of in-vessel composting.

(III) Effect of pH Throughout the Composting Study

Figure 4.6 shows the pH profile of three experimental groups throughout the 60- day in-vessel composting study. The pH of the composting material for the commercial microbial agents and indigenous microbial consortium gradually increase and remained in the pH range of 8.8 to 9.8 from day 3 to day 24. The pH of the commercial microbial agents and indigenous microbial consortium experimental groups increased in the initial stage of the composting as the temperature rises. The increment in pH might due to the increase in ammonia level produced by the biochemical reactions of composting materials that contained nitrogen (Baharuddin et al. 2010). Additionally, the partially treated palm oil mill effluent (POME) supplemented to the oil palm empty fruit bunch (OPEFB) could have also contributed to the slightly alkaline condition. POME has a higher pH than OPEFB (Table 4.2), therefore the pH of the composting material increased slightly over time and until reaching a stable pH. Furthermore, bacteria that often dominates during the thermophilic phase in the composting process are not acid tolerant (Baharuddin et al. 2010). For this study, the pH for the

commercial microbial agents and indigenous microbial consortium finally decrease to a stable pH of 8.8 and 8.5, respectively in the compost. Fully developed compost have been reported to have a pH value of 8-9 (Sundberg *et al.* 2004). Therefore, the compost produced by commercial microbial agents and indigenous microbial consortium would deem to have reached a matured stage.



Figure 4.6 Compost pH profile for 60-day of in-vessel composting.

(IV) Nutrient Changes of the Compost Produced

For this study, the percentage of nitrogen, phosphate, potassium, and magnesium present in the compost at the initial and final stage were analysed. The increment of the nutrients in the compost for all experimental group during the composting study can be seen from Figure 4.7.

All four nutrients tested from the collected compost samples showed an increase in nutrients percentage in for each experimental group. All experimental groups exhibited an increase in nutrient content as these could had been contributed by the nutrients available in the partially treated POME that was sprayed onto the composting material throughout the composting study. Partially treated POME contains appreciable amounts of nitrogen, phosphate, potassium, and magnesium which are vital nutrient elements for the growth of plants (Bala *et al.* 2014). Partially treated POME used in the experiment was tested to have 2.80% of nitrogen, 4.23% of phosphate, 1.80% of potassium and 0.90% of magnesium. Nonetheless, part of the nutrients from partially treated POME could also be used for the growth of microorganisms during the composting process (Baharuddin *et al.* 2010).





Nutrients, % over dry matter

Figure 4.7 Nutrient increments of the composts after 60 days of in-vessel composting.

The compost vessels inoculated with the indigenous microbial consortium exhibited to have the highest increase in nitrogen (1.71%), phosphate (0.32%), potassium (0.50%), and magnesium (0.40%) contents among the different experimental groups. The selected cellulolytic microorganisms which consisted of *Bacillus, Penicillium* and *Aspergillus* species were chosen to form the indigenous consortium showed to have nitrogen-fixing and phosphate solubilising as studied and discussed in Chapter 2. These indigenous microbes could have contributed in the increased of nutrient level in the compost.

The nutrients available in the compost that was inoculated with the commercial microbial agents showed lowest nutrient increments of nitrogen (0.68%), phosphate (0.10%), potassium (0.53%), and magnesium (0.20%) among the three experimental groups. The commercial microbial agents used constituted of Bacillus species microbes. Bacillus spp. are found to be one of the most dominant group that contributes highly to the breakdown of cellulose in oil palm empty fruit bunch (OPEFB) as it takes up the nutrient existing in the composting material rapidly (Krishnan *et al.* 2016). Due to the quick uptake of nutrients, the commercial microbial agents could have contributed to the lower nutrient percentage.

The control of the composting study had an increase of nutrients by 0.92% in the nitrogen, 0.13% in phosphate, 0.67% potassium, and 0.35% in magnesium. The control of the experimental group had a higher nutrient content than the commercial microbial agents group in the experiment as no microbial inoculant was inoculated into the control group. Thus, nutrients were only used slightly since no extra microbes was inoculated into it resulting in a higher nutrient increment than the commercial microbial agents group.

The nutrient analysis results of the indigenous microbial consortium experimental groups obtained from the composting study showed a more prominent nutrient increment compared to the other experimental group, hence making it a potential indigenous microbial consortium for the composting of OPEFB in large scale.
4.3.2 Pot Trial

For the end this study, the aim was to develop an effective soil supplement by combining compost and chemical fertiliser to increase the growth of oil palms seedlings. This was to reduce the dependency on chemical fertiliser. 6-month old oil palms were treated with 4 different treatments as shown in Table 4.2. No further nutrients were added to the oil palm seedlings after the commencement of the experiment. The oil palms were watered daily to overcome the water loss due to evaporation (Baharuddin *et al.* 2009). Sufficient water was provided to the oil palm seedlings as water deficiency may cause growth rate to decrease and also an uneven distribution of nutrients in different parts of plants (Cheng *et al.* 2011). Furthermore, fertilisation is most effective when the oil palms are not water-stressed (Sands & Mulligan 1990). Throughout this experiment, the increases in height, diameter of stem, and number of leaves of the oil palm seedlings were observed (Figure 4.1). The results obtained from the study will be presented and discussed in the following sections.

4.3.2.1 Physical Observations After 6-Month of Pot Trial Experiment

(I) Leaf

The number of fully opened leaves of the oil palms was counted as it was useful as one of the indicators to observe the improvement in the growth of treated oil palms. The number of leaves on the oil palms was counted in 3rd and 6th month. Only leaves that have fully opened were counted while the leaves that have not opened were left out of the calculation. The average number of leaves that were counted on the oil palms was tabulated in Table 4.4.

P value of each treatment group was calculated using one-way ANOVA that is available IBM SPSS Statistics 23. From the observation of Table 4.4, only oil palms that were treated with compost group and compost-chemical mixed fertiliser group showed to have significant differences to p-value ($p \le 0.05$) at the 3rd month of treatment. The oil palms treated with chemical fertiliser did not show a significant difference to the control group (p > 0.05) during this period. The oil palms treated with chemical fertiliser and the control group produced 3 new leaves each during the 3rd month. As for 6th month, the treatment groups and the new leaf count (in parenthesis) were: chemical fertiliser (5 leaves), compost group (8 leaves) and compost-chemical mixed fertiliser group (8 leaves). These showed that oil palms treated with compost produced more new leaves than the untreated oil palm (control).

Average Numbers of New Fully Opened Leaves of Oil Palm Seedlings (cm)			
Control	Chemical fertiliser	Compost	Compost-
			chemical mixed
			fertiliser
3 ± 0.00 b	3 ± 0.00 b	5 ± 0.32 a	5 ± 0.19 a
$4\pm0.42\boldsymbol{b}$	$5\pm0.48 a$	8 ± 0.00 a	$8\pm0.00\mathbf{a}$
	age Numbers of N Control $3 \pm 0.00\mathbf{b}$ $4 \pm 0.42\mathbf{b}$	age Numbers of New Fully OpenedControlChemical fertiliser $3 \pm 0.00\mathbf{b}$ $3 \pm 0.00\mathbf{b}$ $4 \pm 0.42\mathbf{b}$ $5 \pm 0.48\mathbf{a}$	age Numbers of New Fully Opened Leaves of Oil PaControlChemical fertiliserCompost $3 \pm 0.00\mathbf{b}$ $3 \pm 0.00\mathbf{b}$ $5 \pm 0.32\mathbf{a}$ $4 \pm 0.42\mathbf{b}$ $5 \pm 0.48\mathbf{a}$ $8 \pm 0.00\mathbf{a}$

 Table 4.4 Numbers of new fully opened leaves of oil palms over a 6-month period.

** Note:

Each value is the mean \pm standard deviation of the 10 oil palms from each experimental group.

Each value that is significantly different from p value where $p \le 0.05$ is labelled with the letter 'a' while the label letter 'b' indicates a value that does not have a significant difference to the p-value.

This study showed that that compost may enable more new leaves to be produced. Compost, either used *per se* or in combination with chemical fertiliser, would both enhance newer leaf growth as early as 3rd month, while the chemical fertiliser may only enhance such growth in the 6th months.

(II) Height

During the pot trial study, the height of the oil palm was used as one of the indicators to detect the growth of the trees. The height of the treated oil palm was observed and measured using a normal tape measure shown in Figure 4.1. The height obtained from the observation was recorded in unit cm. The height of oil palms was recorded every three months throughout the pot trial study. The data obtained from the experiment was tabulated in Table 4.5. P value of each treatment group was calculated using one-way ANOVA that is available IBM SPSS Statistics 23.

In the 3rd month of treatment, chemical fertiliser, compost and compost-fertiliser groups exhibited significantly differences with the p-value where $p \le 0.05$ were noted

when compared to the control group. Control group did not show a significant difference to the p-value in the 3rd month of treatment. The control group took a longer period of time to exhibit a significant difference to the p-value only at the 6th month.

Height of Oil Palm Seedlings (cm)				
	Control	Chemical fertiliser	Compost	Compost- chemical mixed fertiliser
3 rd Month	$2.98\pm0.42\textbf{b}$	$5.27 \pm 0.63 \mathbf{a}$	$3.92\pm0.48 \textbf{a}$	$4.52\pm0.26\textbf{a}$
6 th Month	$3.96 \pm 0.47 \mathbf{a}$	10.35 ± 0.42 a	$8.72 \pm 1.33 \mathbf{a}$	$9.67 \pm 0.32 \mathbf{a}$
** Note:				

Table 4.5 Height of oil palms over a 6-month period.

Each value is the mean \pm standard deviation of the 10 oil palms from each experimental group.

Each value that is significantly different from p value where $p \le 0.05$ is labelled with the letter 'a' while the label letter 'b' indicates a value that does not have a significant difference to the p-value.

In the 3rd month, chemical fertiliser group showed average height of 5.27 ± 0.63 cm. It was followed by oil palms treated with compost-chemical mixed fertiliser (4.52 ± 0.26 cm), compost (3.92 ± 0.48 cm) and lastly, control (2.98 ± 0.42 cm). All treatment groups, be it compost, chemical fertiliser and their combination, show to improve the height at 3rd month. During the 6th month of pot trial, except for the control group, all groups showed an increase of approximately 5 cm from the 3rd month. Although the chemical fertiliser group showed the higher increase, it is not significantly different from the compost and compost-chemical mixed fertiliser group. In this experiment, the growths of plant height were equally good in all treatment groups. If data were to be obtained beyond 6th month, a clear effect of compost on height could be concluded.

(III) Girth

Stem girth is an important aspect of oil palms as it is during assessment as it shows the potential number of fruit bunches an oil palm can carry at one time (Abidemi, Akinrinde & Obiqbesan 2006). Hence, the changes in the stem girth were measured and used as one of the growth indicator for the oil palms during this pot trial study. A cotton string was used to obtain the stem girth of the oil palm. The cotton string length that went around the stem base was measured with a measuring tape. The same tape measure was used for the measurement of height and stem girth. The data obtained from the study was tabulated in Figure 4.6. P value of each treatment group was calculated using one-way ANOVA that is available IBM SPSS Statistics 23.

Stem Girths of Oil Palm Seedlings (cm)				
	Control	Chemical fertiliser	Compost	Compost- chemical mixed fertiliser
3 rd Month	$5.37\pm0.37\boldsymbol{b}$	$5.54\pm0.48\textbf{b}$	7.47 ± 0.55 a	$8.11 \pm 0.14 \mathbf{a}$
6 th Month	$8.46\pm0.28 a$	$13.72\pm0.68 a$	$15.62 \pm 0.69 \mathbf{a}$	16.82 ± 0.28 a
** Note:				

Table 4.6 Stem	girths of oil	palms over a	6-month period.
----------------	---------------	--------------	-----------------

Each value is the mean \pm standard deviation of the 10 oil palms from each experimental group.

Each value that is significantly different from p value where $p \le 0.05$ is labelled with the letter 'a' while the label letter 'b' indicates a value that does not have a significant difference to the p-value.

Both the compost group and compost-chemical mixed fertiliser group showed fast effects in stem girth growth as significant stem growths were noted in 3^{rd} month of pot trial with significant p-values (p ≤ 0.01) in comparison to control group. The control and chemical fertiliser group however only showed significant differences (p ≤ 0.01) later at 6^{th} month towards the end of the pot trial.

Oil palms treated with compost produced in this study thus showed to enhance the growth of stem girth better than the other experimental groups. Oil palms treated with compost-chemical mixed fertiliser exhibited a good growth of stem girth of 16.82 ± 0.28 cm at the 6th month which were almost doubled the sizes of the stem girths of the control group. The control group that had no fertiliser showed the smallest stem girths when compared to the other experimental groups throughout the study. From the study, the compost group exhibited to a better girth growth when compared to the chemical fertiliser group. The mix of compost and chemical fertiliser exhibited a better source of nutrients to the oil palm seedling for stem girth growth.

(IV) Nutrient in Oil Palm Leaf

The measurement of nutrient levels in oil palm leaf or foliar analysis was used as an indicator to test the efficiency of the treatment on the growth of the treated oil palms. Foliar analysis was used to determine whether the fertilisers applied were utilised by the oil palms (Pushparajah 1979). Freshly harvested leaves were sent to Sarawak Plantation Services Sdn. Bhd. in Kuching, Sarawak to performed foliar analysis in accordance to MS 677: Parts III, IV, V, VI, VII: 1980 to analysed for the percentage of nitrogen, phosphate, potassium, and magnesium based on the dry leaf matter. From the sampling of the oil palms leaves, the foliar analysis obtained were tabulated in Table 4.7.

Nutrient in Oil Palm Leaf (%)				
	Control	Chemical fertiliser	Compost	Compost- chemical mixed fertiliser
Nitrogen	$1.506 \pm 0.03 \mathbf{b}$	$1.834 \pm 0.01 \mathbf{a}$	$1.927 \pm 0.05 \mathbf{a}$	$2.011 \pm 0.01 \mathbf{a}$
Phosphate	$0.121\pm0.03\textbf{b}$	$0.142\pm0.03 \textbf{a}$	$0.154 \pm 0.02 \mathbf{a}$	$0.166\pm0.01 \textbf{a}$
Potassium	$1.285\pm0.03\textbf{b}$	$1.786 \pm 0.02 \mathbf{a}$	$1.469 \pm 0.05 \mathbf{a}$	$1.606 \pm 0.02 \mathbf{a}$
Magnesium	$0.160\pm0.04 \textbf{b}$	$0.179 \pm 0.03 \boldsymbol{a}$	$0.305 \pm 0.05 \mathbf{a}$	$0.236 \pm 0.01 \mathbf{a}$
** Note:				

Table 4.7 Nutrient levels in oil palm leaves over a 6-month period.

Each value is the mean \pm standard deviation of the 10 oil palms from each experimental group.

Each value that is significantly different from p value where $p \le 0.05$ is labelled with the letter 'a' while the label letter 'b' indicates a value that does not have a significant difference to the p-value.

P value of each treatment group was calculated using one-way ANOVA that is available IBM SPSS Statistics 23. Treated oil palms (chemical fertiliser, compost, and compost-chemical mixed fertiliser) exhibited to have a significant ($p \le 0.01$) difference from the control group in all the nutrients tested from the foliar analysis.

Leaves from oil palms treated with chemical, compost, and compost-chemical mixed fertiliser were 1.83%, 1.93% and 2.01% respectively, all significantly higher in nitrogen percentage than control which was 1.51%. There was no significant difference among each of the three treatment groups, albeit the compost seemed to provide higher nitrogen nutrient than chemical fertiliser and could have help to contribute to the highest nitrogen level recorded in the compost-chemical mixed fertiliser group. Thus, compost could probably comparable to chemical fertiliser in supplementing nitrogen to oil palm. Nitrogen is one of the major nutrients required by oil palms which known to be responsible for increasing the growth and fruiting. Furthermore, nitrogen also known to promote the growth of leaf and number of bunches with a heavier weight.

In the foliar analysis for phosphate content, the findings were similar to nitrogen content that all treatments groups were significantly different from the control, albeit there was no significant difference among the treatment groups. The phosphate contents for chemical, compost and chemical-compost mixed fertiliser groups were 0.14%, 0.15% and 0.17% respectively, while the control had 0.12% phosphate content as it was not supplemented with any additional nutrients. Although couldn't be proven significant, the phosphate content was higher for the compost group if compared to chemical fertiliser group, and this could have contributed to the highest phosphate level in the compost-chemical mixed fertiliser group. This observation is similar to nitrogen content. This result thus indicating that compost may substitute part of the phosphate requirement by oil palm which is conventionally being provided by chemical fertiliser. Phosphate is one the nutrients required by oil palms for

improvement in the bunch production rate and weight, thus, increasing the yield that could obtain from the oil palm fruit bunch.

The potassium contents in the leaves for the treatment groups were significantly different to control, albeit there was no significant difference noted among the treatment groups. The potassium contents for chemical, compost and chemical-compost mixed fertiliser groups were 1.79%, 1.47% and 1.61% respectively, while the control had 1.29% potassium content. Unlike nitrogen and phosphate, the potassium content was actually lower for compost group if compared to chemical fertiliser group, although the results could not be proven significant. This lower potassium could have pulled down the potassium level in compost-chemical mixed fertiliser group. Thus, compost produced in this study may only be able to provide potassium at a lower level to the oil palm. Potassium is an essential nutrient to oil palm as it helps the oil palm in a similar way with phosphate as it also increases the yield of the oil palm.

As for magnesium, the results were similar to nitrogen and phosphate where compost was shown to provide high level of such nutrient, although the results could not be proven significant. The magnesium contents for chemical, compost and chemical-compost mixed fertiliser were 0.18%, 0.31% and 0.24% respectively, all significantly more than the 0.16% recorded for the control.

It is interesting to note that potassium may play a role in the increment of height in oil palm seedlings in this study. Chemical fertiliser, which showed the largest height increment among the different treatments (Table 4.5), also exhibited to have the highest potassium percentage from the foliar analysis (Table 4.7).

Nonetheless, from the foliar analysis, even though chemical fertiliser showed to have contributed to the highest percentage of potassium, oil palms treated with compost gave exceptional results in high levels of nitrogen, phosphate, and magnesium when compared to chemical fertiliser group and control. The compost seemed to be an equally good source of nutrients to be supplemented to the oil palm seedlings in comparison to chemical fertiliser. From this study, it can also be seen that compost-chemical mixed fertiliser exhibited an exceptional result as it may have combined the good properties of both chemical fertiliser and compost.

(V) Physical Observation of Oil Palms After 6-Month of Experiment

The pot trial experiment was carried out in an open environment at a vacant field. The condition of the surrounding environment was similar to a plantation estate. Unpredictable weather was inflicted onto the oil palms during the trial. Thus, the results obtained from the experiment might be influenced by such factors. The observation of the physical appearance of the oil palm was recorded at the end of the 6-month timeframe of the pot trial. The physical appearance of the treated oil palms (chemical fertiliser, compost, and compost-chemical mixed fertiliser) were placed next to the untreated oil palm (control) for comparison purposes as illustrated in Figure 4.8.



Figure 4.8 Physical appearances of oil palms treated with different treatments which were Chemical fertiliser (A), compost (B), and compost-chemical mixed fertiliser (C) after 6 months of the experiment.

Oil palms treated with the control group were not treated with any additional nutrient. As can be seen in Figure 4.8, oil palms which did not receive any additional nutrient were much smaller in size. It also can be visually seen to be smaller when placed next to treated oil palms. This experiment supported the notion that sufficient and balanced nutrient must be supplemented for the optimum growth of oil palm as suggested by Chen (2006). Chemical fertiliser showed to have the highest height increment when shown in Table 4.5. The physical appearance of oil palms treated with chemical fertiliser appeared to be taller in height when compared to oil palms treated with compost. Oil palms treated with compost however appeared to be physically larger due to larger stem girth and leafier. Oil palms treated with compost mixed with chemical fertiliser group. The oil palms treated with compost had leafier and larger stem than the chemical fertiliser group. Hence, from the pot trial study, it

can be said that the compost produced in this study showed to have a better enhancement on the overall growth of oil palms in stem girth and leaves. Conversely, the integration of chemical fertiliser with compost had a better influence on the height of oil palms.

A clearer effect of the compost could be concluded if data were to be obtained beyond 6th months. The compost enriched with chemical fertiliser would continuously show a better influence on the growth of oil palms. Nutrients in chemical fertiliser are often washed away by rain falls shortly after being applied (Sharma & Chetani 2017). Unlike chemical fertilisers, compost conditions the soil to enable the release nutrients to plants slowly over the time span of months and years (Singh *et al.* 2010).

4.3.2.2 Treatment Which Showed Potential as A Soil Supplement

Oil palms that were 6 months of age were treated with four different treatments for a 6month period. The four groups were the control group (no fertiliser), chemical fertiliser group, compost group, and compost-chemical mixed fertiliser group. From the pot trial experiment, oil palms treated with the compost produced in this study gave an exceptional result when compared to the positive control (chemical fertiliser) and negative control (no fertiliser). The compost-chemical mixed fertiliser group, followed by compost group, showed to have the best influence on increasing the growth of the oil palm seedlings. This development could be due to the growth enhancement abilities of the cellulolytic microorganisms that were inoculated into the compost group during the composting experiment. These cellulolytic microorganisms which showed various growth enhancement abilities were discussed in Chapter 2. The mix population of bacteria and fungi from the genus Bacillus, Penicillium and Aspergillus were used in the indigenous microbial consortium for the composting of oil palm empty fruit bunch (OPEFB). Bacillus spp. were known to have growth enhancement abilities (Arrebola et al. 2010; Raupach & Kloepper 1998). The Bacillus sp. S2 and S8 isolated in this study exhibited nitrogen fixing and phosphate solubilising ability respectively. Both these Bacillus spp. and fungal isolate Aspergillus sp. F8, and Penicillium sp. F10 had also shown to produce indole acetic acid (IAA) when tested using the Swalkowski method. Lastly, Penicillium sp. F8 used in this study had shown nitrogen fixing ability.

Even though compost group and compost-chemical mixed fertiliser group showed a lower height increment than the chemical fertiliser group of 1.63 cm and 0.68 cm based on the

mean height respectively, both compost group and compost-chemical mixed fertiliser group exhibited to have the better influence on stem girth, leaf production, as well as demonstrating high level of essential nutrient contents in the foliar analysis. Thus, the compost produced by this indigenous microbial consortium could also be an effective soil supplement. By supplementing chemical fertiliser with compost, the oil palm seedlings had exhibited slightly better results than the compost when applied *per se*.

In this pot trial study, the integration of compost and chemical fertiliser was shown to improve the growth of oil palm seedling in different aspects. Related studies had also reported that mixing chemical fertiliser with compost would give positive effect to the soil and the plants (Dutta *et al.* 2003; Ahmad *et al.* 2008). The oil palms treated with compost-chemical mixed fertiliser group would be leafier and having thicker stem girth. Hence, the oil palms treated with compost-chemical mixed fertiliser fruit bunches (Abidemi *et al.* 2006). The integration of compost and chemical fertiliser could also help reduce the dependency of chemical fertilisers by 50% for the cultivation of oil palms.

4.4 Summary

The research study of Chapter 4 was focused on the testing the efficiency of the indigenous microbial consortium that was composed of the selected cellulolytic microorganisms isolated from the collected soil samples in degrading oil palm empty fruit bunches (OPEFB) into compost. OPEFB and the indigenous microbial consortium was composed along with partially treated palm oil mill effluent (POME) that helped moisten the compost and supply nutrients for the degrading microorganisms to thrive. The composting study was carried with three different experimental groups which were the control group, commercial microbial agents group, and indigenous microbial consortium group. The composting study was carried out using an in-vessel composting method that studied for 60 days. The composting vessel was turned and sprayed with partially treated POME three times a week to ensure sufficient aeration and moisture level was provided for the growth activities of the degrading microorganisms. Different parameters were analysed throughout the compost study such as moisture, pH, temperature, C: N ratio, and nutrient changes. The pH, temperature, and C: N ratio were used to measure the composting rate of the raw materials. The indigenous microbial consortium group showed to have a comparable result to the commercial microbial agents group. The indigenous microbial consortium group entered the thermophilic phase on the second day along with the commercial microbial agents group and successfully obtained a C: N ratio of 12:1 and 13:1 respectively on day 30 which indicated the compost have reached a matured level. The indigenous microbial consortium group also exhibited higher nutrient increments of 35% N, 57% P, and 9% Mg when compared to the commercial microbial agents group. From the results obtained from the composting experiment, it can be said that an effective indigenous microbial consortium of degrading microorganisms was successfully created in this study as it degrades OPEFB waste into compost group with the help of their enzymes in a shorter period of time of 30days.

In the second half of this chapter, the efficiency of the compost group produced during the composting experiment waste tested for its enrichment potency in the growth of oil palm seedlings. Oil palms seedling that was 6-month of age were obtained and treated with four different treatments namely control, chemical fertiliser, compost, and compostchemical mixed fertiliser. For compost-chemical mixed fertiliser, only half of the amount of the chemical fertiliser and compost group used in chemical fertiliser and compost group were used to integrate it together. The pot trial was carried out within a 6-month timeframe. The different parameter of the oil palms seedlings was observed throughout the study. The number of fully opened leaves, stem girth, and height of the oil palms were measured at 3^{rd} and 6^{th} month. At the end of the 6-month pot trial, the leaves were harvested for foliar analysis to obtain its nitrogen, phosphate, potassium, and magnesium contents. From the study, the control group, which has no added nutrient applied to it, exhibited the lowest growth rate based on its growth parameters in comparison with treatment groups. Chemical fertiliser groups showed highest in plant height when compared to the other treatments, followed by compost-chemical mixed fertiliser. However, compost-chemical mixed fertiliser group showed the highest leaf production and stem girth, both significant at $p \le 0.05$. Compost-chemical mixed fertiliser group also exhibited significant highest nutrient content based on foliar analysis ($p \le 0.01$). Thus, compost-chemical mixed fertiliser group showed to have the ideal overall good influence on the growth of the oil palm seedlings in this study.

Chapter

5

GENERAL CONCLUSIONS, RECOMMENDATIONS, AND FUTURE WORKS

5.1 General Conclusion

5.1.1 Aims of Thesis

The main aim of this thesis was to isolate cellulolytic microorganisms which produce cellulases and xylanases from Sarawak soil for the composting of oil palm empty fruit bunch (OPEFB) waste produced by the oil palm industry. The rapid development of oil palm industry during recent decades has contributed to a significant amount of wastes that would eventually become a major environmental hazard. Ways to treat and dispose of these wastes must be sought after to reduce the harmful effect that it may inflict to the environment. Therefore, this thesis proposed the use of cellulolytic microorganisms to convert OPEFB waste into compost group through composting.

This thesis reports on the approach to isolate and identify the cellulolytic microorganisms obtained from soil samples. The cellulolytic microorganisms were evaluated for its cellulase and xylanase activities through several bioassays. From the results obtained from the bioassays, five cellulolytic strains which showed high cellulolytic activity were selected to create the indigenous microbial consortium for in-vessel composting of OPEFB. The compost produced was evaluated for its effectiveness in growth enhancement of oil palm seedling via a pot trial. This chapter summarises the results obtained during this research study that has been described in details in chapter 2, 3, and 4. Future works and recommendation of this study were also discussed in this chapter.

5.1.2 Screening and Isolation of Cellulolytic Microorganism

Selective media were used to target microorganisms capable of degrading cellulose and xylan. Carboxymethylcellulose (CMC) agar was chosen as the primary screening and isolation for cellulase producing microorganisms. Microorganisms which showed the ability to degrade cellulose were further tested for its ability to degrade xylan using dyed xylan agar. These strains which exhibited cellulolytic activity on both CMC and xylan were further tested for growth promoting abilities such as nitrogen-fixing, phosphate solubilising, potassium solubilising and indole acetic acid production. Molecular identification of these cellulolytic strains were members of the genus of *Bacillus, Staphylococcus, Pseudomonas, Fusarium, Endomelanconiopsis, Penicillium, Aspergillus* and, *Trichoderma*. The identified cellulolytic strains were further tested through several assays.

Cellulase and xylanase activity of these 24 strains were tested using 2-cyanoacetamide method. *Aspergillus niger* (NBRC 33023) was used as a control strain for the assay. From the isolated strains, five strains showed to have produce high level of cellulases and xylanases in comparison to *A. niger* (NBRC 33023). These five strains were paired and tested respectively for mutual inhibition activity where no antagonism was shown. These five selected cellulolytic strains that showed high cellulase and xylanase productions and no mutual antagonism were selected to form an indigenous microbial consortium for the in-vessel composting study. These five strains which formed the indigenous microbial consortium are *Bacillus* sp. S2 (bacterium), *Bacillus* sp. S8 (bacterium), *P. citrinum* (fungus), *A. nomius* (fungus), and *P. rolfsii* (fungus).

5.1.3 Evaluating Efficiency of Cellulolytic Microorganisms as an Indigenous Microbial Consortium in Degrading OPEFB Via In-Vessel Composting Method to Produce Compost

The efficiency of the indigenous microbial consortium comprised of cellulolytic strains was evaluated for composting of oil palm empty fruit bunch (OPEFB) waste. Three different experimental groups were carried out in triplicates via a 60-day in-vessel composting study: control, commercial microbial agents, and indigenous microbial consortium. The composting vessel was turned and sprayed with partially treated POME 3 times a week to ensure sufficient aeration and moisture level was provided for the growth of the microorganisms. Parameters analysed throughout the compost study were moisture level, pH, temperature, C: N ratio, and nutrient changes.

The indigenous microbial consortium showed comparable efficiency to the commercial microbial agents in composting of OPEFB, as both experimental groups progressed rapidly into the thermophilic phase on the second day. These indicated that that the composting was in good progress. Both groups also reached the mature level on day 30 with a C: N ratio of 13:1 and 12:1 respectively. According to Jimennez and Perez (1992), compost which exhibited a stable ratio of 15:1 and below could be considered as a satisfactory maturation level of compost. The indigenous microbial consortium produced the highest nutrient increment during composting when compared to the other experimental groups. The composting experiment showed that indigenous microbial consortium had successfully degraded OPEFB in a shorter period of time of 30 days when compared to the conventional composting processes that normally require 60-90 days.

5.1.4 Small Scale Pot Trial to Evaluate the Growth Enhancement Potency of Compost Group in Oil Palm Seedlings

A pot trial was conducted to evaluate the efficiency of the compost group on the enhancement of the growth of oil palm seedlings. 6-month oil palm (*Elaeis guineensis*) seedlings purchased from Sarjoh plantation, Sarawak. Four different treatment groups were tested namely control, chemical fertiliser, compost, and compost-chemical mixed fertiliser. Compost-chemical mixed fertiliser group consisted of half the amount of the chemical fertiliser used in chemical fertiliser group, and full amount of compost at the same level that would be used in the compost group. The pot trial employed a completely randomised design with 10 replicates for each treatment. This 6-month trial was aimed to observe the number of fully opened leaves, stem girth, and plant height of oil palm seedlings, as these parameters were often used as growth indicators for oil palms. At the end of the pot trial, the leaves of the oil palm seedlings were also sent to analyse the nitrogen, phosphate, potassium, and magnesium contents.

From the pot trial study, control group without any fertiliser exhibited the lowest growth parameters were compared to the other treatment groups. Chemical fertiliser showed to have the highest in plant height, which was then followed by compost-chemical mixed fertiliser. Nonetheless, chemical fertiliser seemed to be inferior in comparison to compost for leaf production, stem girth growth and leaf nutrient level. Oil palm seedling from the compost group showed to have a better growth of leaves and stem girth, and higher leaf nutrient when compared to the chemical fertiliser group. Interestingly, when compost mixed with chemical fertiliser in the compost-chemical mixed fertiliser group, the best growth of leaves and stem girth were recorded. Compost-chemical mixed fertiliser group also exhibited the highest nutrient content of nitrogen, phosphate, potassium, and magnesium in the foliar analysis. Hence, compost-chemical mixed fertiliser group exhibited to have the best overall influence on the growth enhancement of oil palm seedlings when compared to the other treatments in this pot trial.

5.2 Future Works

Through the findings from this study, composting using indigenous microorganisms shows to be a good and sustainable way to recycle oil palm industry waste into useful compost fertiliser. A mixture of compost with chemical fertiliser had been proven to improve the growth of oil palm seedlings in a pot trial. The selected cellulolytic strains of *Bacillus* sp. S2, *Bacillus* sp. S8, *Penicillium citrinum, Aspergillus nomius, and Penicillium rolfsii* had formed an efficient indigenous microbial consortium that could degrade oil palm empty fruit bunch (OPEFB) in an in-vessel composting study in 30 kg scale. Further investigation of the effectiveness of indigenous microbial consortium in degradation of OPEFB into compost group in larger scale should be studied such as the large scale windrow composting plant. The compost produced from the composting process should be further analysed for its potential enhancement in the growth of oil palms in field trials in plantations for a longer period of time.

Additionally, further works on the improvement of the composting process such as incorporation of fly ash into the composting material could be carried out. Fly ash is a waste material produced from coal power due to the production of energy plants through the use of coal, thus often considered an environmentally harmful substance. However, Mupambwa et al. (2015) had claimed that fly ash contains a high amount of essential plant nutrients such as phosphate, but the limited bioavailability of the nutrients itself has prevented the direct implementation in agriculture. If the harmful elements in the fly ash can be minimised while maintaining the phosphate content, it can be used in the agriculture. Fly ash contains very high salinity which in turns is affiliated with high pH. It also incurs boron toxicity which limits microbial activity within the soil when it is assimilated directly to the soil. Although high in phosphate content, fly cash still is lacking of many important plant nutrients and unsuitable in pH. However, a research done by Ravikumar et al. (2008) suggests that it may be possible to incorporate compost along with fly ash with the aid of earthworms as a bio-remediation method to convert the fly ash into beneficial nutrients which include nitrogen and phosphate, it also might help improve the pH balance. Incorporating fly ash into the composting process would be a good way to help reduce more types of waste products while increasing the total phosphate concentration in the compost.

Furthermore, the enhancement of the cellulolytic enzyme ability of the selected cellulolytic microorganisms from the soil samples to form the indigenous microbial consortium during this thesis study could be studied. A paper by Adsul *et al.* (2007) investigated on the improvement of *Penicillium* spp. to stimulate higher cellulase production. *Penicillium* spp. showed an increased production of cellulose enzymes through repeated process of mutating the strain through EMS followed by UV-irradiation. This idea can be adapted to increase the cellulase production of the cellulolytic strains that can produce potent cellulose enzymes would help to break down the OPEFB waste at a faster rate. This will also address the lengthy time needed for composting as it is one of the major limitations of composting process.

5.3 Concluding Statements

This thesis aimed towards isolating indigenous cellulolytic microorganisms that have the abilities to produce cellulases and hemicellulases for the composting of oil palm empty fruit bunch (OPEFB) waste. Selective media were used to screen and isolate a total of 24 cellulolytic microorganisms from soil samples taken from various locations within Sarawak. Several bioassays were also performed to eventually select five effective cellulolytic strains to create an indigenous microbial consortium to degrade OPEFB via in-vessel composting within 30 days. The compost produced was shown to be able to improve oil palm growth based on significant improvement in leaf growth, stem girth and nutrient level, either applied *per se*, or in combination with chemical fertiliser. The aims and objectives of this study have been successfully achieved. The findings obtained from this study suggested that the isolated strains were potential OPEFB degraders to produce compost for the growth enhancement of oil palm seedlings.

References

Abdullah, N & Sulaiman, F 2013, 'The oil palm wastes in Malaysia', *INTECH*, vol. 1, no. 3, pp. 75-100

Abidemi, A, Akinrinde, E & Obiqbesan, G 2006, 'Oil palm (*Elaeis guineensis*) seedling performance in response to phosphorus fertilization in two benchmark soils of Nigeria', *Asian Journal of Plant Sciences*, vol. 5, no. 5, pp. 767–775.

Ahmad, R, Naveed, M, Aslam, M, Zahir, ZA, Arshad, M & Jilani, G 2008, 'Economizing the use of nitrogen fertilizer in wheat production through enriched compost', *Renewable Agriculture and Food Systems*, vol. 23, no. 3, pp. 243–249.

Ahmad, R, Shahzad, S & Khalid, A 2007, 'Growth and yield response of wheat (*Triticum Aestivum L.*) and maize (zea mays 1.) to nitrogen and L-tryptophan enriched compost', *Pakistan Journal of Botany*, vol. 39, no. 2, pp. 541–549.

Alves, PDD, Siqueira, FF, Facchin, S, Horta, CCR, Victória, JMN & Kalapothakis, E 2014, 'Survey of microbial enzymes in soil, water, and plant microenvironments.', *The Open Microbiology Journal*, vol. 8, pp. 25–31.

Amore, AA, Giacobbe, SS & Faraco, VV 2013, 'Regulation of cellulase and hemicellulase gene expression in fungi', *Current Genomics*, vol. 14, no. 4, pp. 230–249.

Ang, SK, Yahya, A, Aziz, SA & Salleh, M 2015, 'Isolation, screening, and identification of potential cellulolytic and xylanolytic producers for biodegradation of untreated oil palm trunk and its application in saccharification of lemongrass leaves', *Preparative Biochemistry & Biotechnology*, vol. 45, no. 3, pp. 279–305.

'Malaysian standard' n.d., *Standards & Industrial Research Institute of Malaysia*, viewed 23 September 2015, http://stg.jsm.gov.my/documents/372014/372056/MS+677++Pt. +I+to+VIII+1980-RECOMMENDED+METHODS+FOR+PLANT+CHEMICAL+AN ALYSIS+(PART+I+TO+VIII)-828067.pdf/8c9dfd3c-36d8-4d16-af7d-60b46b8e4f90?v ersion=1.0&previewFileIndex=.

Archana, A & Satyanarayana, T 1997, 'Xylanase production by thermophilic *Bacillus licheniformis* A99 in solid-state fermentation', *Enzyme and Microbial Technology*, vol. 21, no. 1, pp. 12–17.

Arrebola, E, Sivakumar, D & Korsten, L 2010, 'Effect of volatile compounds produced by Bacillus strains on postharvest decay in citrus', *Biological Control*, vol. 53, no. 1, Elsevier, pp. 122–128.

Baharuddin, AS, Razak, MNA, Hock, LS, Ahmad, MN, Abd-Aziz, S, Rahman, NAA, Shah, UKM, Hassan, MA, Sakai, K & Shirai, Y 2010, 'Isolation and characterization of thermophilic cellulase-producing bacteria from empty fruit bunches-palm oil mill effluent compost', *American Journal of Applied Sciences*, vol. 7, no. 1, pp. 56–62.

Baharuddin, AS, Wakisaka, M, Shirai, Y & Aziz, AS 2009, 'Co-composting of empty fruit bunches and partially treated palm oil mill efluents in pilot scale', *International Journal of Agricultural Research*, vol. 4, pp. 69–78.

Bala, J, Lalung, J & Ismail, N 2014, 'Palm oil mill effluent (POME) treatment microbial communities in an anaerobic digester: a review', *International Journal of Scientific and Research Publications*, vol. 4, no. 6, pp. 1–24.

Bastawde, KB 1992, 'Xylan structure,microbial xylanases, and their mode of action', *World Journal of Microbiology and Biotechnology*, vol. 8, pp. 353–368.

Baum, D 2008, 'Reading a phylogenetic tree: the meaning of monophyletic groups', *Scitable by Nature Education*, https://www.nature.com/scitable/topicpage/reading-a-phylogenetic-tree-the-meaning-of-41956.

Bernal, MP, Sánchez-Monedero, MA, Paredes, C & Roig, A 1998, 'Carbon mineralization from organic wastes at different composting stages during their incubation with soil', *Agriculture, Ecosystems and Environment*, vol. 69, no. 3, pp. 175–189.

Binod, P, Palkhiwala, P, Gaikaiwari, R, Madhavan Nampoothiri, K, Duggal, A, Dey, K
& Pandey, A 2013, 'Industrial enzymes - present status and future perspectives for india', *Journal of Scientific and Industrial Research*, vol. 72, no. 5, pp. 271–286.

Bisaria, VS & Ghose, TK 1981, 'Biodegradation of cellulosic materials: substrates, microorganisms, enzymes and products', *Enzyme and Microbial Technology*, vol. 3, no. 2, pp. 90–104.

Biswas, R, Persad, A & Bisaria, VS 2014, 'Production of cellulolytic enzymes', *Bioprocessing of Renewable Resources to Commodity Bioproducts*, pp. 105–132. Blanchette, R, Held, BW, Jurgens, J, Mcnew, DL, Harrington, TC, Shona, M, Farrell, RL & Duncan, SM 2004, 'Wood-destroying soft rot fungi in the historic expedition huts of Antarctica wood-destroying soft rot fungi in the historic expedition huts of Antarctica', *Applied And Environmental Microbiology*, vol. 70, no. 3, pp. 1328–1335.

Breuil, C & Saddler, JN 1985, 'Comparison of the 3,5-dinitrosalicylic acid and Nelson-Somogyi methods of assaying for reducing sugars and determining cellulase activity', *Enzyme and Microbial Technology*, vol. 7, no. 7, pp. 327–332.

Chang, LK, Arai, T, Kosugi, A, Ibrahim, D & Mori, Y 2012, '*Penicillium rolfsii*, the potential lignocellulolytic fungus on hydrolysis of oil-palm residues from oil palm trunk as a second generation biofuel feedstock', *2nd Annual International Conference Syiah Kuala University 2012 & The 8th IMT-GT Uninet Biosciences Conference.*, vol. 2, no. 1, pp. 22–24.

Chen, H 2014, Biotechnology of lignocellulose: Theory and practice, Springer, Beijing.

Chen, J-H 2006, 'The combined use of chemical and organic fertilizers and/or biofertilizer for crop growth and soil fertility', *International Workshop on Sustained Management of the Soil-Rhizosphere System for Efficient Crop Production and Fertilizer Use*, no. October, pp. 1–11.

Chen, YP, Rekha, PD, Arun, AB, Shen, FT, Lai, WA & Young, CC 2006, 'Phosphate solubilizing bacteria from subtropical soil and their tricalcium phosphate solubilizing abilities', *Applied Soil Ecology*, vol. 34, no. 1, pp. 33–41.

Cheng, S, Hong, CX, Hongo, S, Xin, LT & Yong, X 2011, 'Growth and physiological responses to water and nutrient stress in oil palm', *African Journal of Biotechnology*, vol. 10, no. 51, pp. 10465–10471.

Clarridge, JE & Alerts, C 2004, 'Impact of 16S rRNA gene sequence analysis for identification of bacteria on clinical microbiology and infectious diseases', *Clinical Microbiology Review*, vol. 17, no. 4, pp. 840–862.

Collins, T, Gerday, C & Feller, G 2005, 'Xylanases, xylanase families and extremophilic xylanases', *Federation of European Microbiological Societies Microbiology Reviews*, vol. 29, no. 1, pp. 3–23.

Datta, C & Basu, PS 2000, 'Indole acetic acid production by a rhizobium species from root nodules of a leguminous shrub, *Cajanus cajan*', *Microbiological Research*, vol. 155, no. 2, Urban & Fischer Verlag, pp. 123–127.

Dibb, D 2002, 'Introduction to inorganic and organic nutrients', *Better Crops, International Plant Nutrition Institute*, vol. 86, no. 2, p. 2002.

Ekinci, K, Keener, HM & Akbolat, D 2006, 'Effects of feedstock, airflow rate, and recirculation ratio on performance of composting systems with air recirculation', *Bioresource Technology*, vol. 97, no. 7, pp. 922–932.

Esse, PC, Buerkert, A, Hiernaux, P & Assa, A 2001, 'Decomposition of and nutrient release from ruminant manure on acid sandy soils in the Sahelian zone of Niger, West Africa', *Agriculture, Ecosystems and Environment*, vol. 83, no. 1–2, pp. 55–63.

Evans, BR, Gilman, AK, Cordray, K & Woodward, J 2000, 'Mechanism of substrate hydrolysis by a thermophilic endoglucanase from *Thermotoga maritima*', *Biotechnology Letters*, vol. 22, no. 9, pp. 735–740.

Fang, M, Wong, JWC, Ma, KK & Wong, MH 1999, 'Co-composting of sewage sludge and coal fly ash: nutrient transformations', *Bioresource Technology*, vol. 67, no. 1, pp. 19–24.

Fourti, O, Jedidi, N & Hassen, A 2008, 'Behaviour of main microbiological parameters and of enteric microorganisms during the composting of municipal solid wastes and sewage sludge in a semi-industrial composting plant', *American Journal of Environmental Sciences*, vol. 4, no. 2, pp. 103–110.

Franche, C, Lindström, K & Elmerich, C 2009, 'Nitrogen-fixing bacteria associated with leguminous and non-leguminous plants', *Plant and Soil*, vol. 321, no. 1–2, pp. 35–59.

Gea, T, Artola, A, Sort, X & Sánchez, A 2005, 'Composting of residuals produced in the catalan wine industry', *Compost Science and Utilization*, vol. 13, no. 3, pp. 168–174.

Ghai, S, Sood, SS & Jain, RK 2007, 'Antagonistic and antimicrobial activities of some bacterial isolates collected from soil samples', *Indian Journal Microbiology*, vol. 47, no. 1, pp. 77–80.

'Background document an overview of the palm oil sector: countries and companies', 2015, *Global Palm Oil Conference* 2015, pp. 1–22.

Griffin, DW, Kellogg, CA, Peak, KK, Shinn, EA 2002, 'A rapid and efficient assay for extracting DNA from fungi', *The Society for Applied Microbiology*, pp.210-214

Gupta, M, Kiran, S, Gulati, A, Singh, B & Tewari, R 2012, 'Isolation and identification of phosphate solubilizing bacteria able to enhance the growth and aloin-A biosynthesis of *Aloe barbadensis* miller', *Microbiological Research*, vol. 167, no. 6, Elsevier GmbH., pp. 358–363.

Halsall, I 2011, 'Towards a sustainable energy economy – the role of palm', *Journal of Oil Palm & The Environment*, vol. 2, pp. 1–7.

Hamzah, A, Phan, C-W, Abu Bakar, NF & Wong, K-K 2013, 'Biodegradation of crude oil by constructed bacterial consortia and the constituent single bacteria isolated from malaysia', *Bioremediation Journal*, vol. 17, no. 1, pp. 1–10.

Harper, SHT & Lynch, JM 1984, 'Nitrogen fixation by cellulolytic communities at aerobic-anaerobic interfaces in straw', *Journal of Applied Bacteriology*, vol. 57, no. 1, pp. 131–137.

Hasamudin, WWH & Soom, RM 2002, 'Road making using oil palm fiber', *Malaysian Palm Oil Board Information Series*, vol. 171, no. May, pp. 3–6.

Hassan, MA, Yacob, S, Shirai, Y & Hung, Y 2005, 'Treatment of palm oil wastewaters', *Waste Treatment in the Food Processing Industry*, pp. 101–117.

Hatakka, A 2005, 'Biodegradation of lignin', Biopolymers Online, pp. 129-145.

Hatakka, A & Hammel, K 2010, 'Fungal biodegradation of lignocelluloses', *The Mycota*, pp. 319–340.

Hoitink, HAJ & Fahy, PC 1986, 'Basis for the control of soilborne plant pathogens with compost', *Annual Review of Phytopathology*., vol. 24, no. 58, pp. 93–114.

Ibrahim, CO 2008, 'Development of applications of industrial enzymes from Malaysian indigenous microbial sources', *Bioresource Technology*, vol. 99, no. 11, pp. 4572–4582.

Jiménez, EI & García, VP 1992, 'Determination of maturity indices for city refuse composts', *Agriculture, Ecosystems and Environment*, vol. 38, no. 4, pp. 331–343.

Jeffries, TW 1994, 'Biodegradation of lignin and hemicelluloses', *Biochemistry of Microbial Degradation*, pp. 233–277.

Jørgensen, H, Mørkeberg, A, Krogh, KBR & Olsson, L 2005, 'Production of cellulases and hemicellulases by three Penicillium species: effect of substrate and evaluation of cellulase adsorption by capillary electrophoresis', *Enzyme and Microbial Technology*, vol. 36, no. 1, pp. 42–48.

Jurick, WM, Vico, I, Whitaker, BD, Gaskins, VL & Janisiewicz, WJ 2012, 'Application of the 2-cyanoacetamide method for spectrophotometric assay of cellulose enzyme activity', *Plant Pathology Journal*. vol.11, no.1, pp. 38-41

Kent, WJ 2002, 'BLAT — The BLAST -like alignment tool', *Genome Research*, vol. 12, pp. 656–664.

Keswani, J & Whitman, WB 2001, 'Relationship of 16S rRNA sequence similarity to DNA hybridization in prokaryotes.', *International Journal of Systematic And Evolutionary Microbiology*, vol. 51, no. Pt 2, pp. 667–78.

Khalid, A & Mustafa, W 1992, 'External benefits of environmental regulation: resource recovery and the utilisation of effluents', *The Environmentalist*, vol. 12, no. 4, pp. 277–285.

Khalil, HPS, Hanida, S, Kang, CW & Fuaad, NN 2007, 'Agro-hybrid composite: the effects on mechanical and physical properties of oil palm fiber (EFB)/glass hybrid reinforced polyester composites', *Journal of Reinforced Plastics and Composites*, vol. 26, no. 2, pp. 203–218.

Khianngam, S, Pootaeng-on, Y, Techakriengkrai, T & Tanasupawat, S 2014, 'Screening and identification of cellulase producing bacteria isolated from oil palm meal', *Journal of Applied Pharmaceutical Science*, vol. 4, no. 4, pp. 90–96.

Khonzue, P, Laothanachareon, T, Rattanaphan, N, Tinnasulanon, P, Apawasin, S, Paemanee, A, Ruanglek, V, Tanapongpipat, S, Champreda, V & Eurwilaichitr, L 2011, 'Optimization of xylanase production from *Aspergillus niger* for biobleaching of eucalyptus pulp', *Bioscience, Biotechnology, and Biochemistry*, vol. 75, no. 6, pp. 1129–1134.

Kim, Y-K, Lee, S-C, Cho, Y-Y, Oh, H-J & Ko, YH 2012, 'Isolation of cellulolytic *Bacillus subtilis* strains from agricultural environments.', *International Scholarly Research Notices: Microbiology*, vol. 2012, p. 650563.

Kirk, TK, Gifford, O, Drive, P & Farrell, RL 1987, 'Enzymatic "combustion": the microbial degradation of lignin', *Microbiology*, pp. 465–505.

Krishnan, Y, Bong, CP, Azman, NF, Zakaria, Z, Othman, NA, Abdullah, N, Ho, CS, Lee, CT, Hansen, SB & Hara, H 2016, 'Co-composting of palm empty fruit bunch and palm oil mill effluent: microbial diversity and potential mitigation of greenhouse gas emission'. *Journal of Cleaner Production*, vol. 1, pp. 1-7.

Krogh, KBR, Mørkeberg, A, Jørgensen, H, Frisvad, JC & Olsson, L 2004, 'Screening genus Penicillium for producers of cellulolytic and xylanolytic enzymes.', *Applied Biochemistry and Biotechnology*, vol. 113–116, pp. 389–401.

Kumar, K, Stecher, G, Tamura, K 2016, 'Molecular evolutionary genetics analysis version 7.0 for bigger database', *Molecular Biology and Evolution*.

Lai, Oi-Ming, Tan, Chin-Ping, Akoh, C 2015, *Palm oil: production, processing, characterization, and uses*, Elsevier.

Leton, TG & Stentiford, EI 1990, 'Control of aeration in static pile composting', *Waste Management and Research*, vol. 8, no. 4, pp. 299–306.

Levy, J 1966, 'The soft rot fungi: their mode of action and significance in the degradation of wood', *Advances in Botanical Research*, vol. 2, no. C, pp. 323–357.

Li, Q, Ng, WT & Wu, JC 2014, 'Isolation, characterization and application of a cellulosedegrading strain *Neurospora crassa* S1 from oil palm empty fruit bunch', *Microbial Cell Factories*, vol. 13, no. 1, p. 157. Li, X, Zhang, R & Pang, Y 2008, 'Characteristics of dairy manure composting with rice straw', *Bioresource Technology*, vol. 99, no. 2, pp. 359–367.

Liang, C, Das, KC & McClendon, RW 2003, 'The influence of temperature and moisture contents regimes on the aerobic microbial activity of a biosolids composting blend', *Bioresource Technology*, vol. 86, no. 2, pp. 131–137.

Löbmann, K & Svagan, AJ 2017, 'Cellulose nanofibers as excipient for the delivery of poorly soluble drugs', *International Journal of Pharmaceutics*, vol. 533, pp. 285-297

Lokhande, S and Pethe, A 2017, 'Isolation and screening of cellulolytic bacteria from soil and optimization of cellulase production and activity', *International Journal of Life Sciences*, vol. 5, pp. 277–282.

López-Mondéjar, R, Zühlke, D, Becher, D, Riedel, K & Baldrian, P 2016, 'Cellulose and hemicellulose decomposition by forest soil bacteria proceeds by the action of structurally variable enzymatic systems', *Scientific Reports*, vol. 6, no. 1, Nature Publishing Group, p. 25279.

Madaki, YS & Lau, S 2013, 'Palm oil effluent (POME) from Malaysia palm oil mills: waste or resource', *International Journal of Science, Environment and Technology*, vol. 2, no. 6, pp. 1138–1155.

Mahjoub, R, Yatim, JBM & Mohd Sam, AR 2013, 'A review of structural performance of oil palm empty fruit bunch fiber in polymer composites', *Advances in Materials Science and Engineering*, vol. 2013.

Martinez, D, Larrondo, LF, Putnam, N, Gelpke, MDS, Huang, K, Chapman, J, Helfenbein, KG, Ramaiya, P, Detter, JC, Larimer, F, Coutinho, PM, Henrissat, B, Berka, R, Cullen, D & Rokhsar, D 2004, 'Genome sequence of the lignocellulose degrading fungus *Phanerochaete chrysosporium* strain RP78.', *Nature Biotechnology*, vol. 22, no. 6, pp. 695–700.

Mohammad, N, Alam, MZ & Kabashi, NA 2014, 'Optimization of effective composting process of oil palm industrial waste by lignocellulolytic fungi', *Journal of Material Cycles and Waste Management*, vol. 17, no. 1, pp. 91–98.

Mohite, B 2013, 'Isolation and characterization of indole acetic acid (IAA) producing bacteria from rhizospheric soil and its effect on plant growth', *Journal of soil science and plant nutrition*, vol. 13, no. ahead, pp. 0–0.

Moore, AW & Becking, J 1963, 'Soil science', Nature, vol. 198, pp. 1165-1166.

Malaysian Palm Oil Board (MPOB) 2011, 'Malaysian palm oil industry', viewed 12 April 2016, http://www.palmoilworld.org/about_malaysian-industry.html.

Malaysian Palm Oil Council (MPOC) n.d., 'Malaysian palm oil industry', viewed 13 July 2016, http://www.mpoc.org.my/Malaysian_Palm_Oil_Industry.aspx.

Mukhlis, HMS, Sariah, M,Ismail, MR, Habib, SH & Kausar, H 2013, 'Potential lignocellulolytic Trichoderma for bioconversion of oil palm empty fruit bunches', *Australian Journal of Crop Science*, vol. 7, no. 3, pp. 425–431.

Mutert, E, Fairhurst, TH & Uexküll, HR 1999, 'Agronomic management of oil palms on deep peat', *Better Crops International*, vol. 13, no. 1, pp. 22–27.

Nautiyal, CS 1999, 'An efficient microbiological growth medium for screening phosphate solubilizing microorganisms', *Federation of European Microbiological Societies Microbiology Letters*, vol. 170, no. 1, pp. 265–270.

Pandey, A, Binod, P, Ushasree, MV and Vidya, J 2010, 'Advanced strategies for improving industrial enzymes', *Chemical Industry Digest*, pp. 74–84, http://chemindigest.com/advanced-strategies-for-improving-industrial-enzymes/.

Pandey, A, Soccol, CR, Nigam, P & Soccol, VT 2000, 'Biotechnological potential of agro-industrial residues. I: Sugarcane bagasse', *Bioresource Technology*, pp. 69–80.

Peix, A, Rivas-Boyero, AA, Mateos, PF, Rodriguez-Barrueco, C, Martínez-Molina, E & Velazquez, E 2001, 'Growth promotion of chickpea and barley by a phosphate solubilizing strain of *Mesorhizobium mediterraneum* under growth chamber conditions', *Soil Biology and Biochemistry*, vol. 33, no. 1, pp. 103–110.

Prajapati, KB & Modi, H. 2012, 'Isolation and characterization of potassium solubilizing bacteria from ceramic industry soil', *CIBTech Journal of Microbiology*, vol. 1, pp. 8–14.

Pushparajah, E 1979, 'Leaf analysis and soil testing for plantation tree crops', *International Board for Soil*, pp. 1–12.

Rahman, N, Kin Mun, DL, Kathiravale, S, Tang, KM, S Puvaneswari & Abd Aziz, N 2013, 'Malaysian biomass industry action plan 2020: driving SMEs towards sustainable future', *Journal of Chemical Information and Modeling*, vol. 53, p. 160.

Ramli, R, Junadi, N, Beg, MDH & Yunus, RM 2015, 'Microcrystalline cellulose (MCC) from oil palm empty fruit bunch (EFB) fiber via simultaneous ultrasonic and alkali treatment', *Chemical, Molecular, Nuclear, Materials and Metallurgical Engineering*, vol. 9, no. 1, pp. 8–11.

Raupach, GS & Kloepper, JW 1998, 'Mixtures of plant growth-promoting rhizobacteria enhance biological control of multiple cucumber pathogens.', *Phytopathology*, vol. 88, no. 11, pp. 1158–1164.

Ravikumar, TN, Yeledhalli, NA, Ravi, M V & Rao, KN 2008, 'Physical, physicochemcial and enzymes activities of vermiash compost', *Karnataka Journal of Agricultural Sciences*, vol. 21, no. 2, pp. 222–226.

Rodriguez, A, Perestelo, F, Carnicero, A, Regalado, V, Perez, R, De La Fuente, G & Falcon, MA 1996, 'Degradation of natural lignins and lignocellulosic substrates by soilinhabiting fungi imperfecti', *Federation of European Microbiological Societies Microbiology Ecology*, vol. 21, no. 3, pp. 213–219.

Rosenani, AB, Rovica, R, Cheah, PM & Lim, CT 2016, 'Growth performance and nutrient uptake of oil palm seedling in prenursery stage as influenced by oil palm waste compost in growing media', *International Journal of Agronomy*, vol. 2016.

Rosgaard, L, Pedersen, S, Cherry, JR, Harris, P & Meyer, AS 2006, 'Efficiency of new fungal cellulase systems in boosting enzymatic degradation of barley straw lignocellulose', *Biotechnology Progress*, vol. 22, no. 2, pp. 493–498.

Rupani, PF, Rajeev, PS, Irahim, MH & Esa, N 2010, 'Review of current palm oil mill effluent (pome) treatment methods: vermicomposting as a sustainable practice', *World Applied. Science Journal*, vol. 11, no. 1, pp. 70–81.

Sabrina, DT, Gandahi, AW, Hanafi, MM, Mahmud, TMM & Nor Azwady, AA 2012, 'Oil palm empty-fruit bunch application effects on the earthworm population and phenol contents under field conditions', *African Journal of Biotechnology*, vol. 11, no. 19, pp. 4396–4406.

Sánchez, C 2009, 'Lignocellulosic residues: biodegradation and bioconversion by fungi', *Biotechnology Advances*, pp. 185–194.

Sands, R, Mulligan, D 1990, 'Water and nutrient dynamics and tree growth', *Forest Ecology and Management*, vol. 30, no. 1–4, pp. 91–111.

Santos, TCD, Cavalcanti, IS, Bonomo, RCF, Santana, NB & Franco, M 2011, 'Optimization of productions of cellulolytic enzymes by *Aspergillus niger* using residue of mango a substrate', *Ciência Rural*, vol. 41, no. 12, pp. 2210–2216.

Sarkar, P, Meghvanshi, M & Singh, R 2011, 'Microbial consortium: a new approach in effective degradation of organic kitchen wastes', *International Journal of Environmental Science and Development*, vol. 2, no. 3, pp. 170–174.

Schmit, O 2006, Wood and Tree fungi, Springer, Germany.

Schneider, WDH, Gonçalves, TA, Uchima, CA, Couger, MB, Prade, R, Squina, FM, Dillon, AJP & Camassola, M 2016, '*Penicillium echinulatum* secretome analysis reveals the fungi potential for degradation of lignocellulosic biomass.', *Biotechnology for Biofuels*, vol. 9, no. 1, BioMed Central, p. 66.

Sen, A, Batra, A, L, ADP & Sari, SLA 2012, 'Cellulolytic and hemicellulolytic bacteria from the gut of *Oryctes rhinoceros* larvae', *Test*, vol. 17, no. 1, pp. 26–32.

Sharma, A & Chetani, R 2017, 'A review on the effect of organic and chemical fertilizers on plants', *International Journal for Research in Applied Science & Engineering Technology (IJRASET)*, vol. 5, no. Ii, pp. 677–680.

Sheng, XF 2005, 'Growth promotion and increased potassium uptake of cotton and rape by a potassium releasing strain of *Bacillus edaphicus*', *Soil Biology and Biochemistry*, vol. 37, no. 10, pp. 1918–1922. Shi, W, Norton, JM, Miller, BE & Pace, MG 1999, 'Effects of aeration and moisture during windrow composting on the nitrogen fertilizer values of dairy waste composts', *Applied Soil Ecology*, vol. 11, no. 1, pp. 17–28.

Shoda, M 2000, 'Bacterial control of plant diseases', *Journal of Bioscience and Bioengineering*, vol. 89, no. 6, pp. 515–521.

'Palm oil facts & figures' 2014, Sime Darby Plantation, viewed 22 April 2017, http://www.simedarby.com/upload/Palm_Oil_Facts_and_Figures.pdf

Singh, RP, Ibrahim, MH, Esa, N & Iliyana, MS 2010, 'Composting of waste from palm oil mill: A sustainable waste management practice', *Reviews in Environmental Science and Biotechnology*, vol. 9, no. 4, pp. 331–344.

Sreeja, SJ, Jeba, MPW, Sharmila, JFR, Steffi, TIG and Palavesam. A 2013, 'Optimization of cellulase production by *Bacillus altitudinis* APS MSU and *Bacillus licheniformis* APS2 MSU, gut isolates of fish *Etroplus suratensis*', *International Journal of Advancements in Research & Technology*, vol. 2, no. 4, pp. 401–406.

Sreekala, MS, Kumaran, MG & Thomas, S 1997, 'Oil palm fibers: morphology, chemical composition, surface modification, and mechanical properties', *Journal of Applied Polymer Science*, vol. 66, no. 5, pp. 821–835.

Sridevi, M, Mahalakshmi, M & Ranjan, A 2013, 'Isolation and characterization of phosphate-solubilizing bacterial species from different crop fields of Salem, Tamil Nadu, India', *International Journal of Nutrition, Pharmacology, Neurological Diseases*, vol. 3, no. 1, p. 29.

Stackebrandt, E & Goebel, BM 1994, 'Taxonomic note: a place for DNA-DNA reassociation and 16S rRNA sequence analysis in the present species definition in bacteriology', *International Journal Systematic and Evolutionary Microbiology*, vol. 44, no. 4, pp. 846–849.

Suely, P, Rosa, S, Souza, AL, Tana, A, Nascimento, R, Andrade, EV, Astolfi-filho, S & Nunes-silva, CG 2014, 'Endo and exoglucanases produced by *Penicillium citrinum* isolated from Amazon', *Bio-Medical Centre Proceedings*, vol. 8, no. Suppl 4, p. P179.

Sundberg, C, Smårs, S & Jönsson, H 2004, 'Low pH as an inhibiting factor in the transition from mesophilic to thermophilic phase in composting', *Bioresource Technology*, vol. 95, no. 2, pp. 145–150.

Tamura, K. & Neil, M 1993, 'Estimation of the number of nucleotide substitutions in the control region of mitochondrial DNA in humans and chimpanzees', *Molecular Biology and Evolution*, vol. 10, pp. 512–526.

Tchapda, TD & Oscar, NN 2016, 'Growth parameters of oil palm (*Elaeis guineensis Jacq*.) tree seedlings in response to fertilizer types', *International Journal of Agronomy and Agricultural Research*, vol. 9, no. 1, pp. 55–62.

Thambirajah, JJ, Zulkali, MD & Hashim, MA 1995, 'Microbiological and biochemical changes during the composting of oil palm empty-fruit-bunches. Effect of nitrogen supplementation on the substrate', *Bioresource Technology*, vol. 52, no. 2, pp. 133–144.

Torsvik, V & Øvreås, L 2002, 'Microbial diversity and function in soil: from genes to ecosystems', *Curr Opin Microbiol*, vol. 5, no. 3, pp. 240–245.

Uffen, R 1997, 'Xylan degradation: a glimpse at microbial diversity', *Journal of Industrial Microbiology and Biotechnology*, vol. 19, no. 1, pp. 1–6.

Vane, CH, Drage, TC, Snape, CE, Stephenson, MH & Foster, C 2005, 'Decay of cultivated apricot wood (*Prunus armeniaca*) by the ascomycete *Hypocrea sulphurea*, using solid state 13C NMR and off-line TMAH thermochemolysis with GC-MS', *International Biodeterioration and Biodegradation*, vol. 55, no. 3, pp. 175–185.

Verheye, W 2010, 'Growth and production of oil palm', *Soils, Plant Growth and Crop Production*, p. 32.

Viel, M, Sayag, D, Peyre, A & André, L 1987, 'Optimization of in-vessel co-composting through heat recovery', *Biological Wastes*, vol. 20, no. 3, pp. 167–185.

Vries, RP & Visser, J, 'Aspergillus enzymes involved in degradation of plant cell wall polysaccharides', *Microbiology and Molecular Biology Reviews*, vol. 65, no. 4, pp. 497–522.

Wong, JWC, Mak, KF, Chan, NW, Lam, A, Fang, M, Zhou, LX, Wu, QT & Liao, XD 2001, 'Co-composting of soybean residues and leaves in Hong Kong', *Bioresource Technology*, vol. 76, no. 2, pp. 99–106.

World Wildlife Fund 2002, 'The palm oil industry in Malaysia: from seed to frying pan', no. November, p. 145.

Yacob, S, Hassan, MA, Shirai, Y, Wakisaka, M & Subash, S 2005, 'Baseline study of methane emission from open digesting tanks of palm oil mill effluent treatment', *Chemosphere*, vol. 59, no. 11, pp. 1575–1581.

Yadav, J, Verma, JP & Tiwari, KN 2011, 'Plant growth promoting activities of fungi and their effect on chickpea plant growth', *Asian Journal of Biological Sciences*, vol. 4, no. 3, pp. 291–299.

Yeoh, CY, Chin, NL & Tan, CS 2011, 'Co-composting of palm oil mill wastes', *Journal of Food, Agriculture & Environment*, vol. 9, WFL Publisher, pp. 880-885.

Zainudin, MHM, Hassan, MA, Tokura, M & Shirai, Y 2013, 'Indigenous cellulolytic and hemicellulolytic bacteria enhanced rapid co-composting of lignocellulose oil palm empty fruit bunch with palm oil mill effluent anaerobic sludge', *Bioresource Technology*, vol. 147, Elsevier Ltd, pp. 632–635.

Zeng, GM, Huang, HL, Huang, DL, Yuan, XZ, Jiang, RQ, Yu, M, Yu, HY, Zhang, JC, Wang, RY & Liu, XL 2009, 'Effect of inoculating white-rot fungus during different phases on the compost maturity of agricultural wastes', *Process Biochemistry*, vol. 44, no. 4, pp. 396–400.

Zhao, D, Oosterhuis, DM & Bednarz, CW 2001, 'Influence of potassium deficiency on photozynthesis, chlorophyll content, and chloroplast ultrastructure of cotton plants', *Photosynthetica*, pp. 103–109.

Zhu, Y, Pan, J, Qiu, J & Guan, X 2008, 'Optimization of nutritional requirements for mycelial growth and sporulation of entomogenous fungus *Aschersonia aleyrodis* Webber', *Brazilian Journal of Microbiology*, vol. 39, no. 4, pp. 770–775.

Appendix

Section 1 Media Preparation

Carboxymethylcellulose (CMC) Medium Preparation

CMC medium was prepared using the modified recipe as shown in Table 1. The addition of agar-agar powder was added to the medium only when a solid growth medium was required.

Table 1: Ingredients required for the preparation of CMC agar.

Ingredients	Amount per litre
Magnesium Sulfate Heptahydrate, <i>MgSO</i> ₄ .7 <i>H</i> ₂ <i>O</i>	0.5 g
Potassium chloride, KCl	1.0 g
Ammonium dihydrogen phosphate, NH4H2PO4	1.0 g
Yeast powder	1.0 g
Glucose	1.0 g
Carboxymethylcellulose powder	8.0 g
Agar-agar powder	17.0 g
Deionised water	1000 mL

* Chemicals used in the media were from Sigma-Aldrich, USA.

Xylan Agar Preparation

(i) Preparation of Dyed Xylan

Xylan for the xylan agar was first prepared by dying the xylan from beechwood powder with Remazol Brilliant Blue-R (RBBR) dye. The ingredients used in preparing the dyed xylan are as shown in Table 2.

Table 2: Ingredients required for preparing dyed xylan.

Ingredients	Amount per preparation	
Xylan, from beechwood	2.5 g	
Remazol Brilliant Blue –R	2.5 g	
Deionised water, dH_2O	60 mL	
Sodium hydroxide solution, NaOH	20 mI	
$(1.5g \text{ in } 20\text{mL sterile } \boldsymbol{dH}_2\boldsymbol{0})$	20 IIIL	
Sodium acetate solution, $C_2H_3NaO_2$	20 mI	
$(0.675g \text{ in sterile } \boldsymbol{dH_2O})$	20 IIIL	
96% ethanol	200 mL	
Wash solution	1 T	
(660mL ethanol, 330 mL dH_2O , 1.35g sodium acetate)	I L	
Acetone	50 mL	

* Chemicals used in the media were from Sigma-Aldrich, USA except for xylan from beechwood which was produced by Himedia, India.

Xylan and Remazol Brilliant Blue-R (RBBR) dye were first mixed together in a 500 mL beaker containing 60 mL of deionised water. The RBBR-xylan solution was stirred using a magnetic stirrer that was placed on a hot plate (Favorit, Malaysia) with its heating plate off. At room temperature (25°C), 20 mL of sodium acetate solution was added drop wise every 5-minute interval while leaving the mixed solution to stir continuously. After mixing in the sodium acetate solution, 20 mL of sodium hydroxide solution was poured into it and left to stir for another 90 minutes at room temperature (25°C). To precipitate the RBBR-xylan, 200 mL of 96% ethanol was added to the mixture. The precipitate of the RBBR-xylan was obtained by using a vacuum flask and Whatman filter paper No. 1. The precipitate was washed sequentially with 1 L of wash solution until a colourless filtrate was present. The final precipitate was washed with 100 mL 75% ethanol followed by 50 mL of acetone. The precipitate was left to dry in a fume hood (Lab Craft Company, USA) overnight then kept in a sterile container until required for analysis.

(ii) Preparation of Xylan Agar

Xylan agar was prepared using the modified recipe from CMC agar as shown in Table 1. The CMC powder was replaced with 2 g/L of dyed xylan. The addition of agar-agar powder was added to the medium only when a solid growth medium was required.

Preparation of Nutrient Agar (NA)

NA (Himedia, India) was prepared by suspending 28.0 g of the NA powder in 1L of distilled.

Preparation of Potato Dextrose Agar (PDA)

PDA (Himedia, India) was prepared by suspending 39.0 g of the PDA powder in 1 L of distilled water.

Preparation of Jensen's Agar Medium

Jensen's medium (Himedia, India) was prepared by suspending 39.1 g of the Jensen's medium powder in 1 L of distilled water.

Preparation of Pikovskaya Agar

Pikovskaya agar (Himedia, India) was prepared by suspending 31.3 g of the Pikovskaya mediums powder in 1 L of distilled water.

Preparation of Aleksandrov Agar

Aleksandrov agar (Himedia, India) was prepared by suspending 29.60 g of Aleksandrov medium powder in 1 L of distilled water.

Preparation of Nutrient Broth (NA)

The NB (Himedia, India) was prepared by suspending 13.0 g of the NB powder in 1 L of distilled water.

Preparation of Potato Dextrose Broth (PDB)

PDB (Himedia, India) was prepared by suspending 24.0 g of the PDB powder in 1 L of distilled water.

Sterilization and Storage of Growth Mediums

All growth mediums, chemicals, and glassware used in this study were sterilised by autoclaving at 121°C, 103.42 kPa for 20 min using an autoclave machine (Hirayama-HVE-50, Japan). Growth mediums containing agar that has been sterilised were placed into 60°C oven until required while growth mediums in broth were kept tightly capped on the clean bench working area.

Dispensation of Growth Mediums

Sterilised mediums that contained agar were aseptically dispensed into a sterile 90mm petri dish in a laminar flow cupboard (ESCO, Singapore) to avoid contamination. The sterilised medium was mixed well by inverting the Duran Schott bottle before dispensing its content into the petri dishes. The petri dishes containing the freshly poured agar was left open to dry for 45 minutes in the laminar flow cupboard before being closed and kept aside until required.

Section 2 Standard curve of glucose constructed using the 2-cyanoacetamide method.

